Extended Models of Coulomb Scattering for the Monte Carlo Simulation of Nanoscale Silicon MOSFETs

Ewan Alexander Towie



Submitted to the Faculty of Engineering, Department of Electronics and Electrical Engineering, University of Glasgow in fulfilment of the requirements for the degree of Doctor of Philosophy

November 2009

All work © Ewan Towie, 2009

Abstract

The International Technology Roadmap for Semiconductors (ITRS) specifies that MOSFET logic devices are to be scaled to sub-10nm dimensions by the year 2020, with 32nm bulk devices ready for production and double-gate FinFET devices demonstrated down to 5nm channel lengths. Future device generations are expected to have lower channel doping in order to reduce variability in devices due to the discrete nature of the channel dopants. Accompanying the reduced channel doping is a corresponding increase in the screening length, which is even now comparable with the channel length. Under such conditions, Coulomb scattering mechanisms become increasingly complex as the scattering potential interacts with a larger proportion of the device.

Ionized impurity scattering within the channel is known to be an important Coulombic scattering mechanism within MOSFETs. Those channel impurities located close to the heavily doped source and drain or both, will induce a polarisation charge within the source and drain. These polarisation charge effects are shown in this work to increase the net screening of the channel impurities, due to the inclusion of remote screening effects, and significantly decrease the scattering rate associated with ionized impurity scattering. Remote screening can potentially reduce the control by ionized channel impurities over channel transport properties, leading to an increased sub-threshold current.

A potential model has been obtained that is based on an exact solution of Poisson's equation for an ionized impurity located close to one or both of these highly doped contact regions. The model shows that remote screening effects are evident within a few channel screening lengths of the highly doped contact regions. The resultant scattering model developed from this potential, which is based on the Born approximation, is implemented within a Monte Carlo simulator and is applied to MOSFET device simulation.

The newly developed ionized impurity scattering model, which allows for remote screening, is applied in the simulation of two representative MOSFET devices: the first device being a bulk MOSFET device developed for the 32nm technology generation; the second device is an Ultra-Thin-Body Double Gate (UTB DG) MOSFET developed for the forthcoming 22nm technology generation. Thorough investigative simulations show that for both the bulk MOSFET and the UTB DG MOSFET, that remote screening of channel impurities in these devices is not a controlling effect. These results prove that the current model for ionized impurity scattering employed in Monte Carlo simulations is sufficient to model devices scaled to at least the 22nm technology node, predicted to be in production in the year 2012.

Acknowledgements

There are many people that I would like to thank for helping me to complete this thesis, which I am sure without their help would never have reached an end! Starting with Jeremy Watling who seemed reasonably happy to read through the many drafts of chapters I produced, carefully pointing out errors, typos and discussing with me the points in my work which needed reinforcing. Despite our many heated discussions during my PhD studies, I am now very sure that Jeremy's guidance did really get me to the end of this work and I truly appreciate this.

I would also like to thank John Barker who provided me with this project to study which I found interesting and often extremely challenging, the very reasons I signed up for the PhD in the first place! I wish John all the best in his retirement (which I am sure he looked forward to if only to avoid having to read my thesis!) and I hope that this work achieves all he wanted from the project in the first place.

Thanks also go to Craig Alexander who was very happy to let me use his Monte Carlo simulation program and then answering the many questions I had about it. I also appreciate the help and support of Karol Kalna who was prepared to answer to my often long and winding questions, and of course to regularly beating me at the badminton and squash matches we often played (I have yet to win a game). Really, thanks to everyone in the Device Modelling Group at the University of Glasgow for all your help over the years, including: Asen Asenov for his endless encouragement; Scott Roy for his supportive advice (and straight answers!); Andrew Brown, Gareth Roy and Campbell Millar for their help with everything drift-diffusion and MOSFET related; and finally also to Fiona Baxter who always seemed interested and enthusiastic about my progress!

All my friends in work and at home I should really also thank for the many times I was bummed and had a good old moan about my PhD. Although I swear I doubt that any of them have actually any idea what I was really doing, or for that matter any interest! Thanks also to my Mum who really, really tried to understand what I was doing if only so that she could tell people what I did when they asked, and to the very well received home-cooked dinners!

Publications and Conferences

E. A. Towie, J. R. Barker and J. R. Watling, "Inhomogeneous Electron-Impurity Scattering in High Degeneracy, Nanometre Scale Silicon MOSFETs," Condensed Matter and Materials Physics 2007: University of Leicester, UK, Apr. 12-13, 2007.

J. R. Barker, E. A. Towie, and J. R. Watling, "Inhomogeneous Electron-Impurity Scattering in High Degeneracy, nanometre Scale Silicon MOSFETs including image charge effects: new models," 12th International Workshop on Computational Electronics: Oct. 8-10, 2007.

J. R. Barker, E. A. Towie, and J. R. Watling, "The influence of polarisation and image charges on electron- impurity scattering in high degeneracy, nanometre scale silicon wrap-round gate MOSFETs", 2008, International Symposium on Advanced Nanodevices and Nanotechnology, Waikoloa, Hawaii, USA, Dec. 2-7, 2007.

J. R. Barker, E. A. Towie, and J. R. Watling, "The influence of polarisation and image charges on electron- impurity scattering in high degeneracy, nanometre scale silicon wrap-round gate MOSFETs," Journal of Physics: Conference Series, vol. 109, 012009, 2008

Contents

1	Introduction	1
	1.1 Coulomb Scattering	2
	1.2 Aims and Objectives	5
	1.3 Thesis Outline	6
2	Literature Review	8
-	2.1 Introduction	8
	2.2 Conwell-Weisskonf	10
	2.3 Brooks-Herring	12
	2.3.1 Momentum-Dependent Screening	
	2.4 Third-Body Exclusion	17
	2.5 Atomistic Impurity	19
	2.6 Conclusion	
•		
3	Monte Carlo Simulation	
	3.1 Introduction	
	3.2 Monte Carlo Fundamentals	
	3.2.1 Boltzmann Transport Equation	
	3.2.2 Band Structure	
	3.2.3 Herring-Vogt Transformation	
	3.2.4 Fermi's Golden Rule	
	3.2.5 Self-Scattering	
	3.3 Carrier Scattering Mechanisms	
	3.3.1 Acoustic Phonons	
	3.3.2 Optical Phonons	
	3.3.3 Ionized Impurities	
	3.3.4 Interface Roughness	
	3.4 Monte Carlo Process	
	3.4.1 Single Particle Monte Carlo	
	3.4.2 Ensemble Monte Carlo	
	3.4.3 Device Monte Carlo	
	3.5 Monte Carlo Calibration	
	3.5.1 Bulk Silicon	
	3.5.2 Inversion Layer	
	3.6 Conclusion	
4	Scattering Potential Calculation	50
	4.1 Introduction	
	4.2 Single Interface Potential Model	51
	4.2.1 Potential Solution	
	4.2.2 Contour Plots	54
	4.2.3 Limits of Potential	56
	4.2.4 Comparison with Non-Linear Poisson Solution	
	4.2.5 Strong-Screening Limit	63
	4.3 Double Interface Potential Model	
	4.3.1 Potential Solution	71
	4.3.2 Contour Plots	74
	4.3.3 Long Channel Limit	75
	4.3.4 Comparison with Non-Linear Poisson Solution	76
	4.3.5 Strong-Screening Limit	
	4.4 Conclusion	83
5	Scattering Rate Calculation	\$ 4
0	5.1 Introduction	

	5.2	Single Interface Scattering Model	85
	5.2.1	Scattering Matrix Element	86
	5.2.2	Scattering Rate	88
	5.2.3	Scattering Reference Frame	91
	5.2.4	Analysis of Scattering Rate	104
	5.3	Double Interface Scattering Model	109
	5.3.1	Scattering Matrix Element	110
	5.3.2	Scattering Rate	111
	5.3.3	Scattering Reference Frame	112
	5.3.4	Analysis of Scattering Rate	118
	5.4	Sub-Threshold Impurity Scattering	121
	5.4.1	Numerical Cap to II Scattering	124
	5.4.2	Numerically Capped Single/Double Interface Models	127
	5.4.3	Testing Numerical Capped II Scattering	127
	5.5	Simulator Implementation	129
	5.5.1	Scattering Rate	130
	5.5.2	Scattering Process	130
	5.5.3	Double Interface Coefficient Calculation	131
	5.6	Conclusion	132
6	Resi	Its and Discussion	135
Ū	61	Introduction	135
	6.2	35nm Bulk Device	
	6.2.1	Device Structure and Calibration	
	6.2.2	Analytical Estimate of Remote Screening Effect	
	6.2.3	$I_{\rm D}$ -V _G Behaviour with Remote Screening	
	6.2.4	Summary	
	6.3	22nm Double-Gate Device	
	6.3.1	Device Structure and Calibration	
	6.3.2	Analytical Estimate of Remote Screening Effect	
	6.3.3	$I_{\rm D}$ -V _G Behaviour with Remote Screening.	
	6.3.4	Summary	175
	6.4	Conclusion	176
_	C	, .	150
/		Clusions	1 01
	/.1	Future work	
A	ppendix	A Single Interface Potential Calculation	
	A.1	Linearized Poisson's Equation	
	A.2	General Solution	184
	A.3	Particular Solution	187
	A.4	Coefficients	192
	A.5	Single Interface Solution	193
Δ	nnendiv	B Double Interface Potential Calculation	195
	R 1	I inearized Poisson's Equation	,
	B.1 B.2	Solution of Coefficients	197
	B.2 B.1	Double Interface Solution	201
	1,1		
A	ppendix	C Full Single Interface Scattering Model	
	C.1	Matrix Element	
	C.2	Scattering Rate	
R	eference	PS	

List of Figures

Figure 1.1: Plot of the bare and screened Coulomb potentials for a point charge located at $r_1 = 03$
Figure 1.2: Various Coulomb scattering processes found in modern MOSFET devices
Figure 2.1: Definition of the impact parameter, b, from Rutherford scattering11
Figure 2.2: (a) Semi-log plot of the impact parameter for an electron of energy $E = 25meV$ and (b) the plot of the average inter-ion separation using $a = N_I^{-\frac{1}{3}}$
Figure 2.3: Plot of the degenerate Debye-Hückel screening length
Figure 2.4: (a) The momentum dependent screening factor, $F(\xi,\eta)$, over several values of the reduced Fermi energy, η , and (b) the ratio of momentum dependent screening to Debye-Hückel screening
Figure 2.5: Cross-over of TBE scattering model from the CW to the BH approaches18
Figure 3.1: (a) Energy dispersion relations and (b) group velocity for parabolic and non-parabolic band structures
Figure 3.2: Comparison of parabolic and non-parabolic density of states models with a full numerical description. Reprinted with permission from T. Kunikiyo <i>et al</i> , J. Appl. Phys. 75, 297 (1994). Copyright 1994, American Institute of Physics [80]
Figure 3.3: Flowchart of (a) the single particle and (b) ensemble Monte Carlo simulators40
Figure 3.4: Flowchart for a device-ensemble Monte Carlo technique
Figure 3.5: (a) Timestep stability plot (from Palestri [91]) and (b) mesh spacing stability (from Palestri [93]). Open symbols denote stable Monte Carlo simulations and crosses unstable simulations. (Both figures © 2006 IEEE)
Figure 3.6: (a) Velocity-field and (b) energy-field calibration of Monte Carlo46
Figure 3.7: (a) Bulk mobility and (b) universal mobility calibration of Monte Carlo47
Figure 4.1: Physical picture of the problem system, defining source and channel regions with impurity located Z_1 from interface
Figure 4.2: Contour plots of the potential for the single interface model. Plots (a)-(d) show four different impurity locations, Z_1 , where the channel screening length, $\lambda_c = 4.133nm$, and the source screening length is $\lambda_s = 0.69nm$
Figure 4.3: Figure demonstrating the doping profile of the Poisson test device for the single interface model. The doping transition from $N_D = 10^{20} cm^{-3}$ to $N_A = 10^{18} cm^{-3}$ is assumed to be abrupt
Figure 4.4: Plots of the potential iso-contours for an atomistic impurity in the ideal single interface structure. Plots (a)-(d) denote varying positions, Z_1 , of the atomistic impurity given in relation to (a) the channel depletion width, or (b)-(d) the channel screening length
Figure 4.5: Comparison of the potential of a single atomistic impurity using a fully self-consistent, non-linear Poisson solution and the single interface, analytical solution obtained in section 4.262
Figure 4.6: Comparison of the non-linear Poisson and analytical solutions of a single atomistic impurity located at $Z_1 = 2\lambda_c$. Discrepency between solutions of point charge due to numerical discretisation of Non-Linear Poisson solution at mesh spacing $\Delta x = 0.5$ mm 63
The second seco
HIGHED / // // / OMMORICON OF THE COMPLETE ONLY CORRECTED A SECONDAL SECONDAL TOP TO THE INCOMPLETE

Figure 4.8: Detailed comparison of complete and strongly screened potential models for $Z_1 = 5nm$.
Figure 4.9: Detailed comparison of complete and strongly screened potential models for $Z_1 = 1nm$.
Figure 4.10: Detailed comparison of complete and strongly screened potential models for $Z_1 = 0.5nm$
Figure 4.11: Mobility comparison of the complete and strongly screened models assuming a single ionized dopant at a range of positions Z_1 , given three background doping concentrations70
Figure 4.12: Pictorial representation of the double interface system with the impurity located at Z_1 and the channel length is given by L_c
Figure 4.13: Potential contour plots of a device with a channel length $L_c = \lambda_c = 4.133 nm$. Channel doping is $N_A = 10^{18} cm^{-3}$ and the source-drain doping is $N_D = 10^{20} cm^{-3}$. Plots (a)-(c) identical screening in the source, channel and drain. Plots (d)-(f) include the polarisation charge effects of the highly-doped source and drain regions
Figure 4.14: Lambda channel device for the comparison of the double interface potential with a non-linear Poisson solver
Figure 4.15: Plots of the non-linear Poisson comparison with the analytical model for remotely- screened impurities within the lambda channel device77
Figure 4.16: Comparison of the complete double interface potential with the strongly screened model for two different channel length devices. Atomistic impurity located exactly mid-channel of each device
Figure 4.17: Detailed comparison of the strongly screened and complete double interface potential models
Figure 4.18: Comparison between the strongly screened and complete double interface model using the lambda channel device. Atomistic impurity located at (a) $Z_1 = 1nm$, (b) $Z_1 = 0.5\lambda_c$ and (c) $Z_1 = \lambda_c - 1nm$
Figure 4.19: Plots of the ratio and potential difference between the strongly screened model and the complete model for the lambda channel device. Plot (a) shows the impurity at $Z_1 = 1nm$ and (b) the impurity positioned at $Z_1 = 0.5\lambda_c = 0.5L_c$
Figure 5.1: Fixed scattering reference frame with example case of remotely screened impurity potential contours and incoming/outgoing carrier wave-vectors
Figure 5.2: Definition of the fixed scattering reference frame using a spherical co-ordinate system.
Figure 5.3: Plot of the remotely screened scattering rate at $Z_1 k_c \gg 4$ over all possible incoming carrier angles
Figure 5.4: Plot of the remotely screened scattering rate at $Z_I k_c \approx 1$ over all possible incoming carrier angles
Figure 5.5: Plot of the remotely screened momentum relaxation rate at $Z_1 k_c \approx 1$ over all possible incoming carrier angles
Figure 5.6: Momentum relaxation rate showing dependence on φ_k for incoming carrier dependent model at $n = N_I = 10^{17} cm^{-3}$

Figure 5.7: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{15} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model
Figure 5.8: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position
Figure 5.9: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_I = 10^{17} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model
Figure 5.10: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position102
Figure 5.11: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{19} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model
Figure 5.12: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position103
Figure 5.13: Polar plots of the differential scattering cross-section for an electron at low (solid lines) and high energy (dashed lines) at various impurity locations
Figure 5.14: (a) Scattering and (b) momentum relaxation rates under the low screening conditions of $n = N_1 = 10^{17} cm^{-3}$ for various impurity locations
Figure 5.15: (a) Scattering and (b) momentum relaxation rates under the high screening conditions of $n = N_1 = 10^{19} cm^{-3}$ for various impurity locations107
Figure 5.16: Ratio between the mobilities of the remotely screened and Brooks-Herring models. The impurity position from the source interface, Z_i , is given in (a) units of the respective channel screening length, λ_c , and (b) nanometres from the source interface
Figure 5.17: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{14} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model
Figure 5.18: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length114
Figure 5.19: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{16} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model
Figure 5.20: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length116
Figure 5.21: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_I = 10^{18} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model
Figure 5.22: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length117
Figure 5.23: Polar plots of the double interface scattering cross-section for varying channel lengths with a single impurity located mid-channel, $Z_1 = 0.5L_c$

Figure 5.24: (a) Scattering and (b) momentum relaxation rates under the low screening conditions of $n = N_I = 10^{16} cm^{-3}$ over a range of channel lengths with a single impurity located mid-channel, $Z_I = 0.5L_C$
Figure 5.25: (a) Scattering and (b) momentum relaxation rates under the high screening conditions of $n = N_I = 10^{18} cm^{-3}$ for varying channel lengths with a single impurity located mid-channel, $Z_I = 0.5L_c$
Figure 5.26: Ratio between the mobilities of the remotely screened and Brooks-Herring models. The channel length, L_c , is given in units of (a) the respective channel screening length, λ_c , and (b) nanometres where the impurity is always positioned at $Z_I = 0.5L_c$
Figure 5.27: Plots of (a) the scattering rates and (b) the momentum relaxation rates for standard II scattering models at sub-threshold conditions with $N_1 = 10^{18} cm^{-3}$ and $n = 10^{12} cm^{-3}$
Figure 5.28: Impact parameter for single interface remotely screened impurity model against half the average inter-ion separation distance for sub-threshold conditions of $N_1 = 10^{18} cm^{-3}$ with a low carrier density of $n = 10^{12} cm^{-3}$
Figure 5.29: (a) Energy-field and (b) velocity-field curves for bulk silicon at a donor concentration of $N_D = 10^{18} cm^{-3}$
Figure 5.30: (a) Bulk mobility and (b) universal mobility for silicon at $N_A = 2 \times 10^{18} cm^{-3}$ against experimental data at 300K
Figure 6.1: Structure of the bulk MOSFET device
Figure 6.2: Density plot of bulk MOSFET net doping with the positive scale denoting donor doping and the negative acceptor doping
Figure 6.3: (a) Linear and (b) semi-log I_D -V _G curves for the bulk MOSFET device showing the comparison between Sentaurus, Drift-Diffusion and Monte Carlo simulation
Figure 6.4: (a) Net doping profile with signed log scale and (b) net doping difference plot for the bulk MOSFET
Figure 6.5: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 0V$ and a drain voltage of $V_D = 0.1V$
Figure 6.6: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 0.4V$ and a drain voltage of $V_D = 0.1V$
Figure 6.7: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 1V$ and a drain voltage of $V_D = 0.1V$
Figure 6.8: (a) Linear and (b) semi-log I _D -V _G curves for the bulk MOSFET with remotely screened impurity scattering
Figure 6.9: I_D -V _G characteristics for the bulk MOSFET with error bars for the low gate voltage points at $V_D = 0.1V$
Figure 6.10: Ionized impurity scattering tally along the channel for the bulk MOSFET at (a) $V_G = 0.5V$ and (b) $V_G = 1V$ with $V_D = 0.1V$
Figure 6.11: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 0.5V$ and $V_D = 0.1V$
Figure 6.12: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 1V$ and $V_D = 0.1V$

Figure 6.13: Energy profile along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering at (a) $V_G = 0.5V$ and $V_D = 0.1V$ and (b) $V_G = 1V$ and $V_D = 0.1V$.
Figure 6.14: I_D - V_G characteristics for the bulk MOSFET with error bars for the low gate voltage points at $V_D = 1V$
Figure 6.15: Ionized impurity scattering tally along the channel for bulk MOSFET at (a) $V_G = 0.4V$ and (b) $V_G = 1V$ with $V_D = 1V$
Figure 6.16: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 0.4V$ and $V_D = 1V$
Figure 6.17: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 1V$ and $V_D = 1V$
Figure 6.18: Energy profile along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering at (a) $V_G = 0.4V$ and (b) $V_G = 1V$, for $V_D = 1V$
Figure 6.19: Structure of the Ultra-Thin-Body Double-Gate device
Figure 6.20: Net doping concentration for the Ultra-Thin-Body Double-Gate device162
Figure 6.21: (a) Linear and (b) semi-log I_D - V_G calibration plots showing the comparison between Sentaurus, Drift-Diffusion and Monte Carlo simulation
Figure 6.22: (a) Net doping profile with signed log scale and (b) net doping difference plot for the Ultra-Thin-Body Double-Gate device
Figure 6.23: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 0V$ and a drain voltage of $V_D = 0.1V$ 164
Figure 6.24: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 0.4V$ and a drain voltage of $V_D = 0.1V$
Figure 6.25: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 1V$ and a drain voltage of $V_D = 0.1V$
Figure 6.26: (a) Linear and (b) semi-log I _D -V _G characteristics for the UTB DG MOSFET with remotely screened impurity scattering
Figure 6.27: Semi-log plot of the low gate voltage I_D - V_G characteristics with error bars for the UTB DG MOSFET at $V_D = 0.1V$
Figure 6.28: Ionized impurity scattering tally along the channel for UTB DG MOSFET at (a) $V_G = 0.6V$ and (b) $V_G = 1V$ with $V_D = 0.1V$
Figure 6.29: (a) Velocity profile and (b) electron density along channel of UTB DG device with $V_G = 0.6V$ and $V_D = 0.1V$
Figure 6.30: (a) Velocity profile and (b) electron density along channel of UTB DG device with $V_G = 1V$ and $V_D = 0.1V$
Figure 6.31: Energy profile along channel of UTB DG device with (a) $V_G = 0.6V$ and (b) $V_G = 1V$ for $V_D = 0.1V$
Figure 6.32: Semi-log plot of the low gate voltage I_D - V_G characteristics with error bars for the UTB DG MOSFET at $V_D = 1V$

Figure 6.33: Ionized impurity scattering tally along the channel for UTB DG MOSFET at (a) $V_G = 0.5V$ and (b) $V_G = 1V$ with $V_D = 1V$
Figure 6.34: (a) Velocity profile and (b) electron density profile along channel of UTB DG device with $V_G = 0.5V$ and $V_D = 1V$
Figure 6.35: (a) Velocity profile and (b) electron density profile along channel of UTB DG device with $V_G = 1V$ and $V_D = 1V$
Figure 6.36: Energy profile along channel of UTB DG device with (a) $V_G = 0.5V$ and (b) $V_G = 1V$ for $V_D = 1V$
Figure 7.1: Expected behaviour of the Z dependent component of the potentials
Figure 7.2: Expected behaviour of the <i>Z</i> component of the potential with the source region $Z < 0$, the drain region $Z > L_c$ and the impurity at $Z = Z_1$

List of Tables

Table 1: Values used in empirical correction to II scattering	36
Table 2: Silicon band parameters and constants.	49

Chapter 1 Introduction

The International Technology Roadmap for Semiconductors (ITRS) [1] is a set of manufacturing guidelines which are used by industry to ensure that modern devices are both profitable to manufacture and are efficient in use. The roadmap covers all aspects of production from process technology to final device performance and also highlights the areas where research and development (R&D) are required. In this work the interest within the ITRS is concerned with the digital logic devices roadmap which discusses the scaling and performance of CMOS devices, that is the MOSFET.

The current generation of MOSFETs are based on the bulk or planar devices with a physical gate length of 32nm and are quite advanced with high- κ /metal-gate stacks and strained channels [2-4]. High- κ insulators have been introduced to combat the scaling of the gate oxide, which when using the traditional SiO₂ would otherwise only be a few atomic layers thick [5, 6]. The reintroduction of metal gates (poly-Si gates had been used for almost the previous two decades of scaling due to the increased control over the device and minimal production cost) reduces the negative impact of the high- κ dielectric [7-10]. Channel performance can be improved through the use of strain which increases the carrier mobility [11-13] and there is also an interest in moving to a (110) silicon crystal orientation [9, 14].

The roadmap projects that the MOSFET device can be scaled to a physical gate length of 8.1nm by the year 2022. The bulk device is predicted to be scaled to a physical channel length of 14nm by 2016, therefore to achieve the end of roadmap device scaling requires a different device structure that reduces Short-Channel Effects (SCEs) (see the textbook by Taur [15] for more detail on SCEs) and completes the required performance levels specified by the ITRS.

A device which offers greater scaling than the bulk device has been led by IBM and is the Silicon-On-Insulator (SOI) MOSFET. It remains mostly unchanged from the bulk device in terms of the layout of the gate, channel and source/drain with the difference being that the silicon substrate is much shallower and is placed on an insulator. The SOI MOSFET device offers a performance gain of 20-35% over the bulk MOSFET and is particularly suitable for low-power applications [16]. It too utilises the technological improvements developed for use in the bulk device of high-κ/metal gate and channel strain as seen in the latest generations [17-19]. Although this device provides greater scaling, the roadmap predicts a physical channel length of 10.7nm by 2019, it is not considered to be the device structure to take MOSFET devices to sub-10nm dimensions.

1.1 Coulomb Scattering

The device which is described in the ITRS as the "ultimate MOSFET device" due to its scalability to the end of the roadmap is the Double-Gate (DG) or FinFET device [1]. The DG device is a particularly favourable device for the roadmap as it greatly reduces the deleterious short-channel effects (SCEs) whilst maximising drive current [20]. This device can also be easily integrated into the existing conventional planar manufacturing process [21]. DG devices have been demonstrated down to 5nm physical channel lengths [12, 22, 23] where fully functional SRAM cells have been recently produced using the DG FinFET structure [24].

It is well understood that each new scaling generation in the coming years provides new challenges and will require that new device structures be utilised. This of course means that the device simulators must also be extended and improved to model the complex processes that are involved. A particular effect evident in devices that has been under extensive research is that of Coulomb scattering which is an increasingly deleterious effect on device performance as the dimensions are reduced below typical screening lengths [25].

The focus of this work is to develop an advanced Coulomb scattering model that describes the interaction of channel ionized impurities with their corresponding polarisation charges induced in the source and drain regions. In the following section a brief review of the major Coulomb scattering processes under study in current generation MOSFETS is undertaken. Following this, the aims and objectives of this research will be stated and an outline of the structure of this work will be presented.

1.1 Coulomb Scattering

This form of scattering is based on the Coulomb potential, given by equation (1.1) [26], for a point charge with a charge Ze(e) is the electronic charge and Z is the number of free charges) located at \mathbf{r}_{I} in a medium of permittivity $\varepsilon_{r}\varepsilon_{0}$. This form of the potential is often called the bare Coulomb potential as it does not include the interaction of the other carriers in the definition of the potential.

$$V(\mathbf{r}) = \frac{Ze}{4\pi\varepsilon_{r}\varepsilon_{0}} \frac{1}{|\mathbf{r} - \mathbf{r}_{\mathbf{I}}|}$$
(1.1)

It is clear from the definition of this potential that there will be a singularity when $\mathbf{r} = \mathbf{r}_{\mathbf{I}}$ and that the potential will drop off slowly. Inclusion of the surrounding charge density allows the screening potential of the other carriers to be considered in this potential and is often termed the screened Coulomb potential. This form of the Coulomb potential can be written as

$$V(\mathbf{r}) = \frac{Ze}{4\pi\varepsilon_r\varepsilon_0} \frac{\exp(-|\mathbf{r} - \mathbf{r}_{\mathbf{I}}|/\lambda)}{|\mathbf{r} - \mathbf{r}_{\mathbf{I}}|}$$
(1.2)

where λ is the static screening length which is typically given by the Debye-Hückel form, written here for a non-degenerate system as

$$\lambda = \sqrt{\frac{\varepsilon_r \varepsilon_0 k_B T}{e^2 n}}$$
(1.3)

where k_B is Boltzmann's constant, *T* is the lattice temperature in Kelvin and *n* is the free electron density. The screened Coulomb potential retains the infinitely high peak of a point charge when $\mathbf{r} = \mathbf{r}_{\mathbf{I}}$ but the introduction of the static screening length, λ , reduces the range over which the potential is noticed. This is more clearly demonstrated in Figure 1.1 where the two potential models are plotted for arbitrary units.



Figure 1.1: Plot of the bare and screened Coulomb potentials for a point charge located at $\mathbf{r}_{\mathrm{I}}=0$.

Coulomb scattering in MOSFET simulation commonly describes the effect on device performance of carrier interactions with ionized impurities and the other carriers. Ionized impurity (II) scattering has the more dominant effect in controlling the carrier transport in doped silicon, in particular the mobility of carriers [27-29]. This mechanism is by far the most important Coulomb scattering mechanism due to the unavoidable nature of impurity scattering in MOSFET devices. It arises from the doping of the silicon semiconductor material with either donor or acceptor atoms. Doping is used to modify and improve the electrical properties of silicon such as to improve the threshold voltage by doping the channel or improving drive current by heavily doping the source/drain wells.

1.1 Coulomb Scattering

Carrier-carrier (e-e) scattering is dominant at high carrier concentrations, above $\sim 10^{17} - 10^{18} cm^{-3}$ in silicon and redistributes the carrier momentum among the ensemble [30]. The e-e scattering mechanism is broken into two separate regimes: the interaction of two carriers with each other, binary e-e scattering [31, 32]; the interaction of a carrier with an oscillating carrier collective, that is plasmon scattering [33, 34]. The distinction between the two regimes, the long-range or plasmon interaction and the short-range or e-e interaction is typically defined by the screening length. E-e interactions at a distance greater than the screening length apart are described in terms of plasmon scattering and those closer are described by the short-range interaction. This distinction follows the natural division of e-e interactions where plasma oscillations are manifest over distances greater than screening length, and at distances less than the screening length the interactions behave more like interacting individual particles [35].

These interactions are two of the many different forms of Coulombic scattering that are modelled in MOSFET devices. A depiction of the various Coulomb scattering processes that have been studied for the simulation of MOSFET devices is given in Figure 1.2.



Figure 1.2: Various Coulomb scattering processes found in modern MOSFET devices.

In this figure the majority of Coulombic scattering processes are remote, that is the charged scattering centre is located some distance from the carrier which it scatters. There has been a large increase in the number of researchers studying remote Coulomb scattering recently as the shrinking device dimensions enhance the remote effects.

Remote charge scattering from ionized impurities located in the gate region of poly-silicon gate (remote impurities) devices [36-38] is a strong scattering mechanism in oxide layers less than 3nm thick [39]. This scattering mechanism has been reduced by the introduction of the high- κ materials in the oxide layer which provide a thicker oxide layer for an equivalent silicon dioxide capacitance and a reduced Coulomb potential strength through an increased permittivity, $\varepsilon_r \varepsilon_0$.

1.2 Aims and Objectives

The introduction of high- κ materials may have reduced poly-Si charge scattering but it has introduced further remote Coulomb scattering due to trapped charges in the gate stack (fixed charges) [40]. The effect of these trapped charges is quite significant on effective device mobility and alters the drive current [41, 42]. Trapped charges are formed during the production of these devices and there is the possibility that future process techniques may reduce the number of trapped charges in the oxide layers.

The introduction of metal gates to MOSFET devices with high- κ dielectrics has been demonstrated to increase device performance and also improves the screening of the remote interactions [7]. Despite this, a recent and extensive study of such devices still found that remote coulomb scattering still causes significant mobility degradation [8].

Remote e-e scattering has also been presented to have a negative impact on device performance for devices with thin oxides and channel length less than 40nm [43]. The high density carrier gasses or plasmons that are present in the heavily doped source, drain and poly-silicon gate regions interact with carriers in the channel region and degrade channel mobility. Interactions of channel carriers with the plasmons in the source and drain regions will reshape the carrier distribution in the channel, moving carriers towards the high energy tail of the distribution [44].

The interaction of the channel carriers with gate plasmons, also referred to as Coulomb drag has been studied [45] and quantitative agreement has been shown experimentally [46]. This remote interaction is strong in poly-silicon gate devices but with the increased screening of a metal gate, the remote Coulomb interaction is minimised [7].

1.2 Aims and Objectives

The effect of the closeness of the source and drain highly-doped regions on channel ionized impurity scattering has not been studied previously and is the focus of this work. The induced polarisation charges in the source and drain will increase the screening of the channel impurities (hence remote screening) and will impact the transport of carriers through the channel. This newly studied effect is entitled remote screening of channel impurities and the effect will increase as the channel length is reduced in future generations.

The aim of this work is to produce the new remotely screened scattering model that can be utilised efficiently in the Monte Carlo simulation of MOSFET devices to describe the interaction of channel ionized impurity with the source and/or drain. This new scattering model must be able to cooperate with the existing ionized impurity model. Using the newly developed scattering model,

1.3 Thesis Outline

Monte Carlo simulations of two example MOSFET devices will be simulated to examine the effect on current and next generation devices.

1.3 Thesis Outline

This thesis will be laid out in the following structure:

Chapter 2 will discuss the most common methods of ionized impurity scattering within Monte Carlo simulations in a literature review. This will include some discussion of existing extensions (or advancements) to the basic model which have been previously studied. The distinctions of each model will be highlighted and the relative advantages of each will be discussed.

Chapter 3 presents a brief introduction to the Monte Carlo simulation methodology used within this work. Starting by covering the fundamentals of the Monte Carlo approach, this includes the Boltzmann Transport Equation (BTE) and the self-scattering procedure, the chapter then moves on to the scattering models and the numerical procedure. Calibration results of the Monte Carlo simulator are also presented to demonstrate the accuracy of the used approach with experimental data.

Chapter 4 calculates the scattering (or interaction) potential for the remotely screened ionized impurity model. The resultant potential is then verified against a fully self-consistent, numerical Poisson solution to ensure the calculation is correct. As will be discussed in this chapter, a simplified model is obtained which is shown to be almost identical to the full model.

Chapter 5 utilises the scattering potential to calculate and analyse a complete scattering model suitable for Monte Carlo simulation. A thorough study is also completed to demonstrate that a further simplification to the scattering model can be made which increases the computational efficiency with negligible loss of accuracy. A new numerical approach to the simulation of ionized impurity scattering is also introduced which greatly reduces simulation time without negative impact on the scattering model.

Chapter 6 completes Monte Carlo simulation of MOSFET devices with the newly developed remote screening model for channel ionized impurities. A typical current generation bulk MOSFET device is simulated along with a future generation double-gate structure to examine the effects of remote screening on the channel performance. In accompaniment to the numerical device simulation, an estimated analysis of the analytical channel mobilities is also calculated for the devices to confirm the simulation results.

1.3 Thesis Outline

Finally in Chapter 7 the conclusions of this work are summarised and suggestions for future work are discussed.

Chapter 2 Literature Review

2.1 Introduction

The focus of this PhD work is on ionized impurity (II) scattering in Monte Carlo (MC) simulation, where in a later chapter, an advanced correction to the Brooks-Herring (BH) model will be developed. There are many methods and techniques used to model the scattering of carriers by ionized impurities, of which the most common and important techniques shall be reviewed here.

The first approach to impurity scattering model developed, the Conwell-Weisskopf (CW) model, will be discussed in Section 2.2. This model utilises the Rutherford ion scattering approach, combined with an empirical cut-off to remove the divergent nature of the Coulomb potential approach. Although this approach is not used here in MC simulations of room-temperature MOSFET devices, it is a significant model and is essential to include in a review of impurity scattering.

The BH model is perhaps the most commonly referenced model with regard to II modelling and as such has had the most development in terms of extensions. This approach differs from the CW model in the definition of the scattering potential, where the BH model includes the screening effect of the surrounding carriers. In section 2.3 this model will be reviewed.

The CW and BH models propose different methods of dealing with the divergence of the Coulomb potential, of which both have strong disadvantages at differing points in MOSFET device simulation. Ridley's Third-Body Exclusion (TBE) model combines these two approaches to remove the disadvantages and obtain a model suitable for device simulation. This model is successfully applied to MC simulations within the simulator used in this work and is reviewed in section 2.4 along with a discussion on the need for such an approach.

With modern devices being scaled to sub-50nm dimensions, the number of actual dopants and their position within the channel region has a large effect on device performance. The discrete nature of dopants in modern nanoscaled devices can be modelled using an *ab initio* atomistic approach within both Drift-Diffusion (DD) and MC simulations. Section 2.5 will briefly review this approach of atomistic impurity scattering where dopant atoms are treated as discrete charges in the electrostatic solution of the device. Although this approach does not use a typical scattering rate representation, it is an important method for modelling the effect of ionized impurities in MC simulation.

For reference, the method used here to obtain scattering rates for MC is based on application of Fermi's Golden Rule to obtain a scattering probability rate, $P(\mathbf{k}, \mathbf{k'})$, for a carrier from a state \mathbf{k} to a state $\mathbf{k'}$. Fermi's Golden Rule for an elastic collision, such as the interactions involved in II scattering, is given by [30]

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} |\mathbf{H}_{\mathbf{k}\mathbf{k}}|^2 \,\delta\big(E(\mathbf{k}') - E(\mathbf{k})\big)$$
(2.1)

where $|\mathbf{H}_{\mathbf{k}\mathbf{k}}|^2$ is the square of scattering matrix element and the Dirac delta function, $\delta(..)$, ensures that energy is conserved in the interaction. The scattering matrix element is obtained from the Fourier Transform of the scattering potential, U_s , and can be written as

$$\mathbf{H}_{\mathbf{k}\mathbf{k}} = \frac{1}{\Omega} \int d^{3}\mathbf{r} \exp(-i\mathbf{q}\cdot\mathbf{r}) U_{s}(\mathbf{r})$$
(2.2)

where $\mathbf{q} = \mathbf{k}' \cdot \mathbf{k}$ denotes the transfer of momentum between the carriers and Ω is the unit volume. The scattering rate which can be used within the MC simulation can be found by summing the scattering probability rate over all possible final states, defined as equation (2.3), where the scattering rate is a function of the magnitude of the state \mathbf{k} .

$$\Gamma(\mathbf{k}) = \sum_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}')$$
(2.3)

This can easily be converted to an integral for a 3-dimensional system using the following definition

$$\sum_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') \equiv N_k \int_{\mathbf{k}'} d^3 \mathbf{k}' P(\mathbf{k}, \mathbf{k}') \text{ with } N_k = \frac{\Omega}{(2\pi)^3}$$
(2.4)

In the calculation of the scattering rate, both the incoming and outgoing carrier wave vector, **k** and **k'** respectively, are expanded in the spherical coordinate system, $\mathbf{k} = \{k, \theta_k, \varphi_k\}$ and $\mathbf{k'} = \{k', \theta_{k'}, \varphi_{k'}\}$. By allowing the reference frame of the outgoing wave vector to be aligned with the incoming wave vector (that is by allowing $\theta_k = 0$), the angles of the incoming wave vector are cancelled. As a result of this expansion, often the scattering rate equations are written as function of a vector variable **k** yet utilise a scalar magnitude *k* on the RHS.

2.2 Conwell-Weisskopf

The CW approach to ionized impurity scattering is based in the Rutherford formula for scattering by a charged ion [47]. This approach treats the ionized impurity as a single point charge which is entirely independent from all other ions. The scattering electron is assumed to have a well-defined trajectory. The model utilises the classic Coulomb point charge which models a single impurity atom in a perfect crystal lattice [47, 48]. The scattering potential for the CW model, based on the bare Coulomb potential, is given by equation (2.5).

$$U_{s}(\mathbf{r}) = eV(\mathbf{r}) = \frac{Ze^{2}}{4\pi\varepsilon_{si}\varepsilon_{0}}\frac{1}{\mathbf{r}}$$
(2.5)

Completing the Fourier transform and Born approximation of the scattering potential, the following scattering matrix element is obtained

$$\left|\mathbf{H}_{\mathbf{k}'\mathbf{k}}\right|^{2} = \left(\frac{Ze^{2}}{4\pi\varepsilon_{Si}\varepsilon_{0}}\right)^{2} \frac{1}{\Omega^{2}} \left(\frac{4\pi}{\mathbf{q}^{2}}\right)^{2}$$
(2.6)

Here the number of impurity charge units is given by Z, the static permittivity of silicon is $\varepsilon_{Si}\varepsilon_0$ and the momentum transfer wave-vector for an elastic collision (assuming that $E(\mathbf{k}) = E(\mathbf{k}')$) is specified as $\mathbf{q}^2 = 2\mathbf{k}'^2(1-\cos\theta)$. Evaluating this using spherical co-ordinates for parabolic, spherical bands with Fermi's Golden Rule, equation (2.1), gives the following scattering rate

$$\Gamma_{CW} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^3\pi m^*}{\hbar^3} k \int_0^{\pi} \frac{\sin\theta}{\mathbf{q}^2} d\theta$$
(2.7)

where N_I is the impurity density. The remaining integral in equation (2.7) diverges for $\theta \to 0$. This is a known problem with the bare Coulomb potential, in that the potential has an infinite range over which it is felt.

Using Rutherford scattering, the scattering electron is treated as a classical particle which interacts with the impurity according to the impact parameter, b. This impact parameter describes the interaction of the electron with the impurity via the scattered angle of the electron. It is defined as the perpendicular distance between the scattered electron and impurity ion and is depicted in Figure 2.1.



Figure 2.1: Definition of the impact parameter, b, from Rutherford scattering.

Conwell and Weisskopf solved the problem of the diverging scattering rate by introducing a limit on the potential that defines a cut-off distance for the impact parameter, given by equation (2.8). The limit that was introduced is based on the assumption that the scattering event is a two-body event, an electron scattering with a single impurity.

$$b = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right) \frac{m^*}{\hbar^2 k^2} \cot\left(\frac{\theta}{2}\right)$$
(2.8)

To ensure that the electron scatters from only the closest impurity, the impact parameter is cut-off at half the average inter-ion separation distance. Here the definition of the average inter-impurity ion distance for a uniform distribution is given as $a = N_I^{-\frac{1}{3}}$ [48]. The impact parameter and average inter-ion separation distance are plotted in Figure 2.2 (a) and (b) respectively.



Figure 2.2: (a) Semi-log plot of the impact parameter for an electron of energy E = 25meV and (b) the plot of the average inter-ion separation using $a = N_l^{-\frac{1}{3}}$.

2.3 Brooks-Herring

Defining half the average inter-ion distance as the maximum impact parameter in equation (2.8), e.g. $b_{\text{max}} = \frac{1}{2} N_I^{-\frac{1}{3}}$, and solving for the angle, the definition for the minimum scattering angle can be obtained as

$$\theta_{\min} = 2 \tan^{-1} \left(\left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0} \right) \frac{m^*}{\hbar^2 k^2} \frac{1}{b_{\max}} \right)$$
(2.9)

Using equation (2.9), θ_{\min} , as the lower limit of the integral in equation (2.7) allows the integral to be completing yielding the CW scattering rate.

$$\Gamma_{CW} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{\pi}{2^{\frac{3}{2}}m^{*\frac{1}{2}}E^{\frac{3}{2}}} \cot^2\left(\frac{\theta_{\min}}{2}\right)$$
(2.10)

The CW model finds a solution to the problem of the divergent scattering model due to the diverging scattering potential, although it has been often criticised for the artificial manner of using half the average inter-ion separation. It has also been noted that the choice of exactly half the separation distance is rather arbitrary and other methods have been suggested. These have included the spherical symmetry on the distribution of impurity ions [49] and even a probabilistic method [50, 51] such as equation (2.11).

$$a = \left(\frac{4}{3}\pi N_I\right)^{-\frac{1}{3}}\Gamma\left(\frac{4}{3}\right) \approx \left(2\pi N_I\right)^{-\frac{1}{3}}$$
(2.11)

The major limitation of the CW approach is that it fails to take into account the effect of the mobile charge surrounding the impurity on the scattering potential. This space charge would effectively screen the potential forcing a quicker drop off at large distances from the ionized impurity. This failure in the model will cause an over-estimation of the scattering rate when the mobile charge is much less than impurity density, as in the depletion-region of a p-n junction [52].

2.3 Brooks-Herring

The BH model for ionized impurity scattering uses the Coulomb potential to model a single ionized impurity but also includes the effect of screening from the space charge surrounding the impurity. The screening effect is generally calculated as a length scale which describes the distance the potential will take to respond to a change in the charge density and can be found from the linearized form of the Poisson equation. The Poisson equation for a point charge is given as

equation (2.12), here Ze is the charge on the impurity located at $\mathbf{r}_{\mathbf{I}}$ and the charge density in the system is $\rho(\mathbf{r}) = e(p(\mathbf{r}) - n(\mathbf{r}) - N_A^-(\mathbf{r}) + N_D^+(\mathbf{r}))$.

$$\nabla^2 V(\mathbf{r}) = -\frac{\rho(\mathbf{r})}{\varepsilon_{S_i} \varepsilon_0} - \frac{Z e^2}{2\pi \varepsilon_{S_i} \varepsilon_0} \delta(\mathbf{r} - \mathbf{r}_{\mathbf{I}})$$
(2.12)

Linearization of this equation through the Thomas-Fermi method [53] and assuming a small perturbation in the neutral space-charge, the Poisson equation can be written as

$$\nabla^2 V(\mathbf{r}) = \varepsilon_{s_i} \varepsilon_0 k_c^2 V(\mathbf{r}) - \frac{Ze}{2\pi \varepsilon_{s_i} \varepsilon_0} \delta(\mathbf{r} - \mathbf{r}_{\mathbf{I}})$$
(2.13)

where $\lambda_C = k_C^{-1}$ and

$$k_{c}^{2} = \frac{e^{2}n}{\varepsilon_{s_{t}}\varepsilon_{0}k_{B}T} \frac{\mathfrak{F}_{-\frac{1}{2}}(\eta)}{\mathfrak{F}_{s_{t}}(\eta)}$$
(2.14)

This solution uses Fermi-Dirac statistics for the carrier densities and the degenerate Debye-Hückel form of the inverse screening length, k_c , where $\eta = E_F/k_BT$ and \mathfrak{F}_j is the Fermi-Dirac integral of order j [54]. Typical values for the degenerate screening length given by equation (2.14) are shown in Figure 2.3.



Figure 2.3: Plot of the degenerate Debye-Hückel screening length.

The potential solution to equation (2.13) can then be found using appropriate boundary conditions as the screened Coulomb model, given as the energy in equation (2.15). After completion of the

2.3 Brooks-Herring

Fourier transform and taking the Born approximation, the scattering matrix element can be derived as equation (2.16).

$$U_{s}(\mathbf{r}) = eV(\mathbf{r}) = \frac{Ze^{2}}{4\pi\varepsilon_{st}\varepsilon_{0}} \frac{\exp(-k_{c}\mathbf{r})}{\mathbf{r}}$$
(2.15)

$$\left|\mathbf{H}_{\mathbf{k}'\mathbf{k}}\right|^{2} = \left(\frac{Ze^{2}}{4\pi\varepsilon_{Si}\varepsilon_{0}}\right)^{2} \frac{1}{\Omega^{2}} \left(\frac{4\pi}{\mathbf{q}^{2}+k_{C}^{2}}\right)^{2}$$
(2.16)

Comparing this equation with the scattering matrix element from the CW approach in the previous section, equation (2.6), we can see that the screening length removes the singularity as $\mathbf{q} \rightarrow 0$. Clearly, the CW scattering matrix element can be found by setting the inverse screening length to zero, that is $k_c = 0$.

The scattering rate can be calculated using equation (2.16) in Fermi's Golden Rule and summing over all final states. Here $E_{k_c} = \hbar^2 k_c^2 / 2m^*$ and the scattering rate has been multiplied by the number density of impurities per unit volume, $N_I \Omega$.

$$\Gamma_{BH} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^{\frac{5}{2}}\pi}{m^{\frac{*}{2}}E_{k_c}^2} \frac{E^{\frac{1}{2}}}{1+4E/E_{k_c}}$$
(2.17)

The BH approach is generally more widely accepted [48] as it has removed the divergence of the scattering rate with low-angle scattering. This is true as long as there is enough mobile charge available to screen the Coulomb potential within a reasonable distance [55]. In conditions where there are very few carriers to screen the potential, the BH model returns to the similar divergent nature found in the unscreened model. There has been many corrections and improvements made to the BH model, many of which are detailed in the review by Chattopadhyay & Queisser [48].

2.3.1 Momentum-Dependent Screening

The standard BH approach uses a static screening model which in this chapter is the degenerate Debye-Hückel model of equation (2.14) model which is momentum independent. It has been suggested by some authors [56, 57] that this model for screening is not sufficiently accurate at high doping densities and is being over-estimated. To correct this a momentum-dependent screening correction has been developed which can be applied to the BH case [58, 59].

2.3 Brooks-Herring

Momentum dependent screening can be calculated from the dielectric function assuming that the impurity potential is time-independent and thus the frequency is zero [56]. The wave-vector dependent dielectric function for degenerate statistics can then be written as

$$\varepsilon(\mathbf{q},0) \equiv \varepsilon(\mathbf{q},0) = 1 + \frac{k_c^2}{\mathbf{q}} F(\xi,\eta)$$
(2.18)

$$F\left(\xi,\eta\right) = \frac{1}{\mathfrak{F}_{-\frac{1}{2}}\left(\eta\right)} \frac{1}{\xi\sqrt{\pi}} \int_{0}^{\infty} \frac{x}{1 + \exp\left(x^{2} - \eta\right)} \ln\left|\frac{x + \xi}{x - \xi}\right| dx$$
(2.19)

Here k_c^2 is the inverse screening length given by equation (2.14), $\mathbf{q}^2 = 2\mathbf{k}^2(1-\cos\theta)$ is the momentum transfer of the scattering and $\mathfrak{F}_j(\eta)$ is Fermi integral of order j. The values ξ , x and η are related to the momentum transfer wave vector, carrier momentum and Fermi level respectively.

$$\xi^2 = \frac{\hbar^2 \mathbf{q}^2}{8m^* k_B T} \tag{2.20}$$

$$x^{2} = \frac{\hbar^{2}k^{2}}{2m^{*}k_{B}T}$$
(2.21)

$$\eta = \frac{E_F}{k_B T} \tag{2.22}$$

The momentum dependent screening correction factor has been plotted in part (a) of Figure 2.4. The plot demonstrates the correction to the screening at large momentum transfer and the return to the static screening model at $F(0,\eta)=1$. Within degenerate conditions ($\eta > 0$) the screening correction is reduced as is expected from a higher screening density.



Figure 2.4: (a) The momentum dependent screening factor, $F(\xi,\eta)$, over several values of the reduced Fermi energy, η , and (b) the ratio of momentum dependent screening to Debye-Hückel screening.

The corrected impurity potential is calculated by applying the dielectric function of equation (2.18) to the Fourier Transform of equation (2.5), the CW scattering interaction potential.

$$V_{mom}(\mathbf{q}) = \frac{V(\mathbf{q})}{\varepsilon(\mathbf{q},0)} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right) \frac{1}{\Omega} \frac{4\pi}{\mathbf{q}^2 + k_c^2 F(\xi,\eta)}$$
(2.23)

The scattering rate can then be calculated using Fermi's Golden Rule with the square of the above matrix element, equation (2.23). Noting that the integral in equation (2.19) must be completed numerically which restricts the θ integral from being completed, the scattering rate can be written as

$$\Gamma_{mom}\left(\mathbf{k}\right) = \left(\frac{Ze^{2}}{4\pi\varepsilon_{si}\varepsilon_{0}}\right)^{2} N_{I} \frac{8\pi m^{*}k}{\hbar^{3}} \int_{0}^{\pi} \frac{\sin\theta}{\left(\mathbf{q}^{2} + k_{c}^{2}F\left(\boldsymbol{\xi},\boldsymbol{\eta}\right)\right)^{2}} d\theta \qquad (2.24)$$

Momentum dependent screening is expected to decrease the inverse screening length at large \mathbf{q} values. This will occur at high carrier densities and cause an increase in the scattering rate corresponding to the reduction in magnitude of the screening.

The momentum dependent correction is demonstrated with respect to the static screening model in plot (b) of Figure 2.4. Here the carriers are assumed to have the room temperature average thermal energy, $E = \frac{3}{2}k_BT$ at 300K. The scattering angle defining the momentum transfer wave-vector, **q**, is selected through the corresponding peak angular scattering probability. The peak angular scattering probability is found by obtaining the maximum of the integrand in equation (2.24).

The effect of the correction is negligible below $\sim 10^{17} cm^{-3}$ where most ionized impurity scattering is small-angle forward scattering and hence $\mathbf{q}(\theta \rightarrow 0) \approx 0$. At higher densities, where high-angle scattering is dominant, the momentum dependent correction can reduce the screening by up to 30%.

In the limit of $\lim_{q\to 0} \xi \to 0$, the momentum dependent scattering model given by equation (2.24) will return to the BH model. In other words, when the carrier densities are very low the scattering model will return to the screened BH approach. At the opposite limit, $\lim_{q\to\infty} \xi \to \infty$, when carrier densities are very high, the scattering model will behave like the unscreened CW approach.

2.4 Third-Body Exclusion

The problem of divergent scattering rates with ionized impurity scattering is a difficult problem to solve. The two well-known methods by CW and BH use different solutions to reduce the range of the Coulomb potential. Of these two methods the BH approach has been more widely accepted as it includes the screening of the impurity potential by mobile charge in the system. It allows the Coulomb potential to be curtailed at a distance defined by the screening length. This method is generally very successful when there is sufficient mobile charge in the system to screen the impurity potential. If there is a lack of mobile charge carriers available to screen the potential, the model will return to the divergent nature.

In the CW model the Coulomb potential is strictly cut-off at half the average inter-ion separation distance, assuming a uniform impurity distribution. This method guarantees that the divergent nature of the Coulomb potential is contained and ensures that the scattering event is solely a two-body process. Ridley has developed a model which reconciles the two approaches used by CW and BH, entitled the TBE model. By introduction of a function that gives the probability of another ionized impurity being closer to the scattering carrier, the BH model can be modified to incorporate the CW cut-off [55, 60].

This model provides the CW and BH models as limiting cases such that if the screening length is less than half the average inter-ion separation the screened potential is used otherwise it returns the unscreened, cut-off model used by CW. Figure 2.5 plots the three scattering models against the mobile electron concentration and the cross-over from the BH to CW approaches is clear in the TBE model. In this figure, the electron energy is assumed to be E = 25meV and the background impurity concentration is fixed at $N_I = 10^{18} cm^{-3}$.



Figure 2.5: Cross-over of TBE scattering model from the CW to the BH approaches.

The probability that no other ion is closer to the scattering electron is a function of the impact parameter, b, and is given as [55]

$$P(b) = \exp(-\pi a b^2 N_I) \tag{2.25}$$

where *a* is half the average inter-ion separation given by equation (2.11) and N_I is the density of ionized impurities. The impact parameter is then defined from the differential scattering cross-section for the BH model as

$$\pi b^{2} = 2\pi \int_{\theta}^{\pi} \sigma(\theta_{b}) \sin \theta_{b} d\theta_{b}$$

$$= \left(\frac{Ze^{2}}{4\pi\varepsilon_{si}\varepsilon_{0}}\right)^{2} \frac{2^{3}\pi m^{*2}}{\hbar^{4}} \frac{1}{4k^{2} + k_{c}^{2}} \frac{1 + \cos\theta}{2k^{2}(1 - \cos\theta) + k_{c}^{2}}$$
(2.26)

To obtain the Ridley TBE model, the probability function is applied to the differential scattering cross-section to obtain [60]

$$\sigma_{R}(\theta) = \sigma(\theta) P(b) \tag{2.27}$$

The scattering rate is calculated by integrating the Ridley differential cross-section, σ_R , over θ and multiplying by the impurity density and the group velocity, $\mathbf{v}(\mathbf{k}) = \hbar \mathbf{k}/m^*$.

$$\Gamma_{R} = \mathbf{v}(\mathbf{k}) N_{I} \int_{0}^{\pi} \sigma_{R}(\theta) d\theta \qquad (2.28)$$

Completing the integral and after some manipulation, the final scattering rate can be expressed as

$$\Gamma_{R} = \frac{\mathbf{v}(\mathbf{k})}{a} \left[1 - \exp\left(-\frac{a}{\mathbf{v}(\mathbf{k})}\Gamma_{BH}\right) \right]$$
(2.29)

In this expression Γ_{BH} is given by equation (2.17), the BH scattering rate and $a = (2\pi N_I)^{-\frac{1}{3}}$ is the average inter-ion separation distance.

The Ridley TBE model combines the approach of the more widely accepted screened approach used by BH with the CW cut-off method of removing the divergence. This model for ionized impurity scattering can be applied consistently to all conditions in a MC simulation without worry of it producing excessively large scattering rates when there is few mobile carriers to screen the electrons. It is an advantageous II scattering model as it can be applied to all simulation conditions but it is not as accurate a II model as the BH approach and will underestimate the effect of II scattering in MOSFET devices [61].

The TBE model has traditionally been the approach for II scattering in the MC code used in this work but it is not an ideal approach. As a result, in Chapter 5 a more advanced II model is developed and utilised which retains the accuracy of the BH approach whilst reducing the divergent scattering rate nature inherent to BH.

2.5 Atomistic Impurity

Although the atomistic impurity approach to scattering from ionized impurities is distinct to the research within this thesis, it is important to discuss the method used. A brief overview of the technique will be covered here and further detail of the method and effect on device performance is referred to the original authors [62-64].

As device dimensions are scaled to well below $0.1\mu m$, the effect of impurity centres within the device become much more important [63]. There will be relatively few dopant ions needed to obtain the required doping level. Hence, it is necessary to look at the effect of the discrete nature of these dopants on device performance and carrier transport [65].

Typical MC impurity scattering methods utilise an approach based on Fermi's Golden rule where ionized impurities are included via a continuous background doping level. Atomistic impurity scattering takes a different approach to the normal method used within such MC simulations. In atomistic scattering the discrete impurity charge is included directly into the mesh-based solution

of the non-linear Poisson solution for the electrostatic potential. This electrostatic solution is then used in an Ensemble Monte Carlo (EMC) simulator where conventional II scattering is removed from the scattering tables [66]. Hence, by incorporating the dopant potential into the electrostatic solution of the system, the effect of the dopant on carrier transport can be examined in detail within the MC simulation [67].

The atomistic impurity scattering approach is a classical approach to the problem in that the scattering of the carriers from the II is modelled through the classical transport of MC particles. In other words, rather than using a quantum mechanical description of II scattering utilising a scattering rate, the atomistic approach scatters particles through the electrostatic field of the ionized impurity directly. Therefore as particles are scattered from the electric fields present using Newton's laws of motion, the classical component of MC simulation, the atomistic approach is a classical solution to the problem.

This approach highlights the effect that discrete impurities have on the electrostatics and current drive of the device. Threshold voltage and drain currents vary with exact dopant number and position within the channel and is unavoidable as MOSFET sizes are scaled to decanano dimensions [62, 64, 68].

2.6 Conclusion

This chapter has completed a review of the three major II scattering models used within MC simulations. Starting with the CW model which is considered to be the first approach developed to model II scattering in semiconductors. The CW model is developed from the bare Coulomb potential that neglects the screening effect of the mobile charge. To control the divergence of the bare Coulomb potential it uses an empirical cut-off on the impact parameter such that any scattering event with an impact parameter larger than half the inter-ion separation is neglected. This approach is successful in controlling the divergence but can be quite inaccurate in device conditions when the mobile carrier density can be several orders of magnitude smaller than the impurity density. This reason limits the use of the CW approach to simulation of bulk semiconductors where the mobile charge cannot deviate far from the background impurity density.

The BH model includes the screening effect of the mobile charge on II scattering and is more suitable to a wide variety of device simulations. The use of the mobile charge density to cut-off the potential is a more physical approach to the problem than that utilised in the CW model and provides a better measure of carrier mobility in conditions where carrier density is far from the impurity density. Unfortunately, the BH scattering model suffers a problem of divergent scattering

2.6 Conclusion

rates when the carrier density is several orders of magnitude smaller than the impurity density. As the magnitude of the scattering rate has a direct effect on MC simulation by increasing the physical number of scattering events per unit time, a divergent scattering rate renders simulation unfeasible.

The CW and BH models each use a different approach to controlling the Coulomb potential which have advantages in different circumstances. The TBE model combines these approaches to provide a model which doesn't diverge but also includes the important mobile charge screening effect. This model is particularly advantageous in MC simulations as it can be used consistently in all device conditions but will underestimate the effect of II scattering when the mobile charge density is much smaller than the II density. The importance of including the screening effect of mobile carriers whilst controlling the divergent nature of II scattering forces the use of this non-ideal approach. As mentioned earlier in this chapter, a better model based on the BH approach which reduces the divergent nature is developed in Chapter 5 and applied in this work.

In reviewing the major techniques for II scattering, it is also important to include the approach of atomistic impurity scattering. This is not a typical scattering model utilising Fermi's Golden Rule, instead it resolves discrete ionized impurities in the electrostatic solution to allow simulation of the effect of position and number on device performance. It can be considered a classical approach to the problem by modelling the II scattering through the electrostatic fields which transport the particles using Newton's laws of motion. This approach allows device variability of IIs at the atomic scale to be modelled which is increasingly important as device dimensions shrink.

Chapter 3 Monte Carlo Simulation

3.1 Introduction

The Monte Carlo (MC) simulation methodology is ideal to study the effects of complex scattering mechanisms in MOSFET devices and is the simulation methodology used in this PhD thesis. The MC method is a stochastical technique to solve large and complex mathematical problems, and is applied here for semiconductor device simulation by means of a particle method to solve the Boltzmann Transport Equation (BTE).

The MC simulation program used in this research has been developed to simulate electron transport in nMOSFET devices. It is a fully self-consistent 3D approach coupled to a non-linear Poisson solver and is capable of solving various MOSFET device structures such as the bulk, silicon-overinsulator and double-gate structures. Calibration of the simulator is completed against experimental data for the properties of silicon such as the energy- and velocity-field characteristics and the bulk, concentration-dependent mobility. There is also device calibration in terms of the universal or inversion mobility which has to be matched with experimental data. Once this calibration has been completed, a wide variety of silicon MOSFET structures can be simulated.

This chapter will begin with a brief review of MC fundamentals in section 3.2, including a discussion on the BTE, the band structure and the process of carrier scattering. Following this, the scattering mechanisms employed in the MC simulation used here will be presented and discussed in section 3.3. The method with which devices are solved numerically with the MC procedure is covered in section 3.4. Finally, the calibration of the simulation with experimental data will be demonstrated in section 3.5.

3.2 Monte Carlo Fundamentals

3.2.1 Boltzmann Transport Equation

The BTE is a complex integro-differential equation that describes the semi-classical transport of carriers in a volume of phase space. The BTE is semi-classical as it describes the carrier transport using the classical equations of motion with Newton's laws, but describes the scattering of the carriers through quantum mechanical terms.

The solution to the BTE is the distribution function, $f(\mathbf{r}, \mathbf{k}, t)$, which gives the probability of finding a carrier at a time *t* in the 6-dimensional phase space. Essentially the BTE is a bookkeeping equation for the distribution function recording the flow of carriers in and out of position and momentum space. The BTE must also ensure that particle continuity is conserved, for instance if there is a greater in-flow than out-flow in a volume in phase space or carrier scattering directs many carriers to a volume, the distribution function must change to conserve the particle number.

Once the distribution has been found, many important properties of devices can be obtained. These include the charge density, the carrier density, the mean carrier velocity and the mean carrier energy. These quantities can be found from the distribution function by integrating over all \mathbf{k} states, such as the carrier density in equation (3.1) or the average kinetic energy density in equation (3.2).

$$n(\mathbf{r},t) = \frac{N_k}{\Omega} \int_{\mathbf{k}} f(\mathbf{r},\mathbf{k},t) d^3 \mathbf{k}$$
(3.1)

$$W(\mathbf{r},t) = \frac{N_k}{\Omega} \int_{\mathbf{k}} E(\mathbf{k}) f(\mathbf{r},\mathbf{k},t) d^3 \mathbf{k}$$
(3.2)

For equilibrium systems the distribution function is often expressed by the Fermi distribution for degenerate systems or the Maxwell-Boltzmann distribution for non-degenerate systems. For such equilibrium conditions the distribution function can be calculated from the Fermi energy and the lattice temperature. It is also possible to define the displaced or drifted distribution function which represents a non-equilibrium distribution function and is considered only a good approximation for low-field conditions [30].

The BTE can be derived from the quantum mechanical Liouville-Von Neumann transport equation under a number of simplifying assumptions [69, 70]. It is also possible to define the BTE using an elemental procedure in terms of carrier in- and out-flows of a small volume in phase space over a short time period [30]. It is worth noting that the Drift-Diffusion (DD) and hydrodynamic approaches can be derived from moments of the BTE.

The general form for the semi-classical BTE is given by equation (3.3) below [30, 71, 72]

$$\frac{\partial f}{\partial t} + \mathbf{v} \cdot \nabla_{\mathbf{r}} f + \dot{\mathbf{k}} \cdot \nabla_{\mathbf{k}} f = \frac{\partial f}{\partial t} \Big|_{coll}$$
(3.3)
where f is the distribution function that is to be solved, **v** denotes the change in position of the carriers (the carrier group velocity) and $\hbar \mathbf{k}$ the change in crystal momentum of the carriers subject to the externally applied fields. The 2nd and 3rd terms on the LHS of the BTE refer to the time rate of change of the distribution function from the movement of the carriers in position space and from the movement of carriers in **k** space respectively, ensuring particle continuity. These terms essentially represent the carrier dynamics which obey the classical equations of motion given by equations (3.4) and (3.5) where the Hamiltonian can be written as $H(\mathbf{r}, \mathbf{k}) = E(\mathbf{k}) + eV(\mathbf{r})$ [32].

$$\mathbf{v} = \frac{d\mathbf{r}}{dt} = \frac{1}{\hbar} \nabla_{\mathbf{k}} H = \frac{1}{\hbar} \frac{\partial E(\mathbf{k})}{\partial \mathbf{k}}$$
(3.4)

$$\dot{\mathbf{k}} = \frac{d\mathbf{k}}{dt} = -\frac{1}{\hbar} \nabla_{\mathbf{r}} H = -\frac{1}{\hbar} e \frac{\partial V(\mathbf{r})}{\partial \mathbf{r}} = -\frac{1}{\hbar} e E(\mathbf{r})$$
(3.5)

Incidentally, the electric field is determined by Poisson's equation

$$\nabla^2 V = -\nabla E = -\frac{e}{\varepsilon_{Si}\varepsilon_0} \left(p - n + N_D^+ - N_A^- \right)$$
(3.6)

where $\varepsilon_{si}\varepsilon_0$ is the dielectric constant for silicon, V is the electrostatic potential, e is the electronic charge, p the mobile hole density, n the mobile electron density and N_D^+, N_A^- the ionized donor and acceptor density.

The term on the RHS of the BTE, equation (3.3), refers to the time rate of change of the distribution function from collisions of the carriers in phase space. This scattering term on the RHS of the BTE equation can be expressed for a degenerate system as

$$\frac{\partial f}{\partial t}\Big|_{coll} = \int_{\mathbf{k}'} f(\mathbf{r}, \mathbf{k}', t) (1 - f(\mathbf{r}, \mathbf{k}, t)) P(\mathbf{k}', \mathbf{k}) d^{3} \mathbf{k}' - \int_{\mathbf{k}'} f(\mathbf{r}, \mathbf{k}, t) (1 - f(\mathbf{r}, \mathbf{k}', t)) P(\mathbf{k}, \mathbf{k}') d^{3} \mathbf{k}'$$
(3.7)

where $P(\mathbf{k}, \mathbf{k}')$ is the probability of scattering from a state \mathbf{k} to a state \mathbf{k}' , $f(\mathbf{r}, \mathbf{k}, t)$ gives the probability of finding a carrier at state \mathbf{k} and $(1 - f(\mathbf{r}, \mathbf{k}', t))$ gives the probability of state \mathbf{k}' being empty. For the MC approach used in this work this equation can be simplified assuming a non-degenerate system such that $f(\mathbf{r}, \mathbf{k}, t) \ll 1$ allowing the terms (1 - f(..)) to be removed.

3.2 Monte Carlo Fundamentals

The BTE is a linear equation (for a non-degenerate system) which is not easily solved and although analytical solutions have been found, these are generally limited to small regions of interest by the approximations involved to obtain the solution. Numerical approaches to solving this equation are more commonly applied and have been far more successfully, offering solutions which can be applied to many device conditions. Numerical solutions to the BTE include an iterative approach which solves the whole distribution function at each iteration [30, 71]. This approach was adopted early in the development of such numerical approaches and is suitable for low-field conditions. Another more advanced approach is the expansion of the BTE using spherical harmonics and has been readily applied to MOSFET simulation [30, 73].

The approach used in this work is also an early approach but has been very successful in device simulation and is the most popular solution. The numerical method used is of course the particle based MC methodology and is quite different from the other numerical approaches in that it models the transport of particles directly. The MC method allows the physics of carrier transport to be modelled and yields results for devices that are close to experimental data [71, 74-76]. The numerical MC simulation methodology will be discussed in more detail later in the chapter.

It is important to briefly discuss the three major approximations employed in the BTE equation. A more detailed examination of the approximations involved than that given here can be found in references [30, 69].

The most important approximation made in the BTE is that of the single particle description of the ensemble of particles in a real system. The distribution function in the full problem describes the probability of state of a large number of carriers. Assuming that carrier-carrier interactions are weak, true for dilute concentrations, the ensemble carrier distribution function can be reduced to a single carrier distribution. Although in silicon MOSFETs the carrier concentrations are not always weak the carrier-carrier interactions can be included separately via the scattering term on the RHS. The single particle description remains valid in this case as carrier-carrier interactions affect the distribution function by redistribution of carrier momentum which can be equally well described in the single particle description. It should be noted that the influence of long-range carrier-carrier interactions (electron-plasmon interactions) on the system is included indirectly through the electric field term, governed by Poisson's equation (3.6), provided that the mesh spacing and timestep intervals are carefully chosen [77, 78].

The second approximation, mentioned earlier in this section, is the treatment of the carriers as classical particles which obey Newton's laws. This is clear from the definition of the distribution function which defines both the position and momentum of a given particle. This assumption will hold as long as the electric field is slowly varying over a length comparable to the average carrier

wavelength. Using the thermal de Broglie wavelength equation, the wavelength of an average thermal carrier is $\lambda_B = h / \sqrt{2m^* k_B T} \approx 13nm$ at room temperature (300K).

The third approximation relates to scattering and states that scattering events are assumed to be instantaneous and localised in space. In other words, the carrier does not change position or gain energy from the electric field during the collision. This approximation can be considered valid as long as the mean time between collisions, τ , is greater than $\tau \gg \hbar/k_BT$. This can be related to the mean distance between collisions using the relation $l = v\tau$ and multiplying both sides by the average carrier velocity. This then states than the mean distance between collisions must be greater than the average thermal wavelength, λ_B .

3.2.2 Band Structure

The electrons in a perfect crystal can be modelled by Bloch states, including the periodic crystal potential, with the wave function [71]

$$\Psi_{nk}(\mathbf{r}) = u_{nk}(\mathbf{r})\exp(i\mathbf{k}\cdot\mathbf{r})$$
(3.8)

where *n* is the band index, $\hbar \mathbf{k}$ is the carrier crystal momentum and $u_{nk}(\mathbf{r})$ denotes the periodicity of the crystal. The Bloch states relate a carrier energy *E* to a state \mathbf{k} , with the function $E_n(\mathbf{k})$ often referred to as the energy dispersion relation. The $E_n(\mathbf{k})$ functions describe the band structure of the material for the given band *n*. In this work the material under study is silicon and the band index will be neglected in future references. Also, only the lowest minima conduction band for silicon, the X-valleys, will be considered as this work is concentrated on relatively low energy electron transport (E < ~1.5eV).

For the conduction band often a simple quadratic expression is used to represent the energy dispersion relation as

$$E(\mathbf{k}) = \frac{\hbar^2 \mathbf{k}^2}{2m^*} \tag{3.9}$$

In this expression all the detail of the band structure is contained within the effective mass, m^* . The effective mass plays a pivotal role in defining the bands and many other aspects of the physical theory. As such there are several definitions of this term depending on the application. For a detailed review of the definitions see [71].

The basic model for the energy bands, given by equation (3.9), is sufficient to gain rough estimates on the transport of electrons in a material. In this work a slightly more advanced energy dispersion relation is employed by incorporating the non-parabolic band model. The non-parabolic band model improves the agreement with experiment for the energy bands at higher carrier energies [30]. Nonparabolicity is introduced by using an advanced energy dispersion relation given as

$$E(1+\alpha E) = \gamma(\mathbf{k}) \tag{3.10}$$

$$\gamma(\mathbf{k}) = \frac{\hbar^2 \mathbf{k}^2}{2m^*} \tag{3.11}$$

where α is the nonparabolicity parameter and $\gamma(\mathbf{k})$ is essentially the quadratic expression, equation (3.11). The nonparabolicity parameter is often used as a fitting parameter for the transport data but it can be defined from band parameters with an expression for conduction bands given by equation (3.12) [74].

$$\alpha = \frac{1}{E_G} \left(1 - \frac{m^*}{m_0} \right)^2$$
(3.12)

In this work an experimentally obtained value for the nonparabolicity parameter is used where $\alpha = 0.5 eV^{-1}$ [30, 71, 79]. Obtaining the positive root of equation (3.10) (only the positive root is considered as negative energies are not physical), the energy dispersion relation can be written

$$E(\mathbf{k}) = \frac{-1 \pm \sqrt{1 + 4\alpha \gamma(\mathbf{k})}}{2\alpha}$$
(3.13)

Plotting the parabolic (equation (3.9)) and non-parabolic (equation (3.13)) energy dispersion functions in Figure 3.1, plot (a) demonstrates the differences between the models.



Figure 3.1: (a) Energy dispersion relations and (b) group velocity for parabolic and non-parabolic band structures.

As shown in plot (a) of Figure 3.1 the non-parabolic model increases the spread of energies at higher \mathbf{k} states. This corresponds with an increase in the Density of States (DOS) which for non-parabolic bands is given by equation (3.14).

$$\rho(E) = \frac{3(2m^*)^{\frac{3}{2}}}{\pi^2 \hbar^3} E^{\frac{1}{2}} (1 + \frac{5}{2} \alpha E)$$
(3.14)

The DOS for the parabolic model (obtained from equation (3.14) by allowing $\alpha = 0$) and the nonparabolic model is shown plotted against a full numerical description of the band structure in Figure 3.2.



Figure 3.2: Comparison of parabolic and non-parabolic density of states models with a full numerical description. Reprinted with permission from T. Kunikiyo *et al*, J. Appl. Phys. 75, 297 (1994). Copyright 1994, American Institute of Physics [80].

It is clear from this figure that the DOS using the parabolic model is valid only for very low carrier energies. Whereas the non-parabolic model is much closer to the complete description of the band structure to roughly 1.5eV. Nonparabolicity also has the effect of reducing the carrier velocity for a given state \mathbf{k} . The group velocity for a state \mathbf{k} for non-parabolic bands can be found from the energy dispersion relation, equation (3.13), as

$$v(\mathbf{k}) = \frac{1}{\hbar} \frac{\partial E}{\partial \mathbf{k}} = \frac{\hbar k}{m^* (1 + 2\alpha E)}$$
(3.15)

Plotting the non-parabolic group velocity in part (b) Figure 3.1 highlights the drop in velocity at higher \mathbf{k} states.

3.2.3 Herring-Vogt Transformation

In the equations discussed so far within this chapter the energy bands are assumed to have an isotropic effective mass such that the equi-energy surfaces are spherical. The energy bands for silicon that are under consideration in this work, the X-valleys, actually have ellipsoidal equienergy surfaces, that is an anisotropic effective mass. For ellipsoidal bands the energy dispersion relation, neglecting nonparabolicity, can be written [32]

$$E(\mathbf{k}) = \frac{\hbar^2}{2} \left(\frac{k_x^2}{m_t^*} + \frac{k_y^2}{m_t^*} + \frac{k_z^2}{m_l^*} \right)$$
(3.16)

This representation of the energy dispersion relation makes analytical calculations such as those for the scattering mechanisms extremely challenging. To reduce the complexity of analytical calculations the Herring-Vogt transformation can be applied which reduces the ellipsoidal equienergy surfaces to spherical surfaces. The Herring-Vogt transformation makes use of a starredspace which represents the ellipsoidal wave-vector transformed into the representative spherical system. The transformation is defined by

$$k_i^* = T_{ii}k_i \tag{3.17}$$

where \mathbf{k}^* is the transformed wave-vector and the transformation matrix \mathbf{T} , in the valley frame of reference, is of the form

$$\mathbf{T} = \begin{bmatrix} \left(\frac{m_0}{m_l}\right)^{\frac{1}{2}} & 0 & 0\\ 0 & \left(\frac{m_0}{m_l}\right)^{\frac{1}{2}} & 0\\ 0 & 0 & \left(\frac{m_0}{m_l}\right)^{\frac{1}{2}} \end{bmatrix}$$
(3.18)

Finally, for non-parabolic bands the energy dispersion relation can then be written as

$$\gamma(\mathbf{k}) = \frac{\hbar^2 \mathbf{k}^{*2}}{2m_0} \tag{3.19}$$

with the corresponding non-parabolic group velocity as

$$v_i = \frac{\hbar}{m_0 \left(1 + 2\alpha E\right)} T_{ij} k_j^* \tag{3.20}$$

3.2.4 Fermi's Golden Rule

Scattering plays an important role in carrier transport as it defines the carrier interaction with the lattice, impurity ions and defects in the material. As discussed previously, carrier scattering in the definition of the BTE is a quantum mechanical concept. The collision operator of the BTE, expanded in equation (3.7), includes the quantum mechanical scattering probability transition rate, $P(\mathbf{k}, \mathbf{k}')$. The transition rate describes the probability per unit time of a carrier scattering from a state \mathbf{k} to a state \mathbf{k}' . The probability transition rate is calculated using Fermi's Golden Rule, given by equation (3.21). For a complete discussion on the derivation of the Golden Rule see the textbooks [30, 32, 71].

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} |\mathbf{H}_{\mathbf{k}\mathbf{k}}|^2 \,\delta\big(E(\mathbf{k}') - E(\mathbf{k}) \pm \hbar\omega\big) \tag{3.21}$$

Fermi's Golden Rule is the basic result of scattering theory which is used to describe carrier scattering in semiconductors. In the notation in this chapter, the upper and lower signs are for absorption and emission respectively. The Dirac-delta function $\delta(..)$ ensures that energy is conserved during the scattering interaction by only allowing non-zero interaction probabilities for arguments that are zero.

The scattering matrix element, $\mathbf{H}_{\mathbf{k}\mathbf{k}}$, must be defined from the scattering potential and defines the particular scattering event. Once a scattering potential has been identified the scattering matrix element can be calculated as

$$\mathbf{H}_{\mathbf{k}\mathbf{k}} = \int_{\mathbf{r}} \exp(-i\mathbf{k}' \cdot \mathbf{r}) U_{s}(\mathbf{r}) \exp(i\mathbf{k} \cdot \mathbf{r}) d^{3}\mathbf{r}$$
(3.22)

where $U_s(\mathbf{r}) = eV(\mathbf{r})$ is the scattering potential energy. Here the overlap integral has been assumed to ≈ 1 which is an accurate approximation for non-parabolic conduction bands in silicon [79, 81].

In MC solutions of the BTE the carrier scattering is typically defined as a scattering rate, $\Gamma(\mathbf{k})$, that describes the number of scattering events per unit time of a carrier at an energy $E(\mathbf{k})$. The scattering rate can be calculated by integrating the probability transition rate over all final states \mathbf{k}' described by equation (3.23).

$$\Gamma(\mathbf{k}) = N_k \int_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') d^3 \mathbf{k}'$$
(3.23)

Here $N_k = \Omega / (2\pi)^3$ is related to the number of electron states within the volume Ω .

It is important to note that Fermi's Golden Rule is valid when the duration of a collision is much smaller than the free time between collisions. This condition allows the effect of uncertainty in the carrier energy due to collisions to become significantly small that the carrier energy can be well defined.

3.2.5 Self-Scattering

An important development in the numerical solutions of the BTE and in particular MC simulations is the introduction of self-scattering. Self-scattering is a simplification of the free-flight time choice by introduction of a fictitious scattering event which greatly reduces the computational complexity of the choice of free-flight times for carriers.

The probability per unit time, $P(\tau)$, of a carrier travelling for a time τ and then being scattered is given by [32]

$$P(\tau) = \Gamma_T(\mathbf{k}(t)) \exp\left[-\int_0^{\tau} \Gamma_T(\mathbf{k}(t')) dt'\right]$$
(3.24)

where $\Gamma_T(\mathbf{k}) = \sum_i \Gamma_i(\mathbf{k})$ is the total scattering rate given by the sum of *i* scattering mechanisms, $\mathbf{k}(t) = \mathbf{k}_0 + e\mathbf{F}t/\hbar$ is the carrier wave vector at a given time *t*, \mathbf{k}_0 is the carrier wave-vector at the beginning of the free flight, t = 0.

To obtain the free-flight time from equation (3.24) requires that τ be evaluated for a given $P(\tau)/\Gamma_{\tau}(\mathbf{k})$ using random numbers distributed between 0 and 1. This is a complicated expression which requires numerical integration as the integral over all the scattering rates within the exponential cannot be solved analytically. A simple solution was proposed by Rees [82, 83] through introduction of a scattering rate that does not alter the carrier \mathbf{k} state, aptly titled self-scattering. The simplification is achieved by introducing $\Gamma_0(\mathbf{k}(t))$, the self scattering mechanism into the total scattering rate such that

$$\Gamma_{T}^{sc}\left(\mathbf{k}\left(t\right)\right) = \sum_{i} \Gamma_{i}\left(\mathbf{k}\left(t\right)\right) + \Gamma_{0}\left(\mathbf{k}\left(t\right)\right) = \Gamma_{T}\left(\mathbf{k}\left(t\right)\right) + \Gamma_{0}\left(\mathbf{k}\left(t\right)\right)$$
(3.25)

Substituting this into the probability per unit time of free flight, equation (3.24) yields

$$P(\tau) = \left\{ \Gamma_{T}\left(\mathbf{k}\left(t\right)\right) + \Gamma_{0}\left(\mathbf{k}\left(t\right)\right) \right\} \exp\left[-\int_{0}^{\tau} \left\{ \Gamma_{T}\left(\mathbf{k}\left(t'\right)\right) + \Gamma_{0}\left(\mathbf{k}\left(t'\right)\right) \right\} dt'\right]$$
(3.26)

This expression can be greatly simplified by allowing the value of $\Gamma_0(\mathbf{k}(t))$ to be carefully chosen to remove the energy dependent scattering rate term and introduce a constant value. This is represented as

$$\Gamma_{0}\left(\mathbf{k}\left(t\right)\right) = \Gamma - \Gamma_{T}\left(\mathbf{k}\left(t\right)\right) \tag{3.27}$$

where Γ is a constant value representing the self-scattering rate such that $\Gamma_0 \ge 0$ for all **k** states of interest (this value is selected at the start of simulation). Then the free flight probability becomes

$$P(\tau) = \Gamma \exp(-\Gamma \tau) \tag{3.28}$$

which can be solved for the free-flight time τ through use of a random number r by rearranging the equation to give

$$\tau = -\frac{1}{\Gamma} \ln\left(1 - r\right) \tag{3.29}$$

Although this method introduces more scattering events increasing the computational time, it is more than compensated by reducing the complexity of calculating the free-flight duration. The number of self-scattering events that occur in a simulation can be minimised by selecting the selfscattering rate, Γ , to be the largest value of the total scattering rate for the **k** states considered.

3.3 Carrier Scattering Mechanisms

3.3.1 Acoustic Phonons

Acoustic phonon scattering is the mechanism which describes the interaction of the carriers with the crystal lattice producing a relatively low frequency oscillation of the neighbouring atoms in the lattice. Modelling the exact change in the periodic crystal potential of the oscillating atoms is very challenging and a simplified approach making use of a deformation potential, D_{ac} , is typically employed [32].

The acoustic phonon scattering model used within this MC simulator is based on the inelastic approach given in Jacoboni's MC textbook [71] with a modification to the acoustic phonon dispersion relation taken from a journal paper by Pop [84]. The final scattering rate suitable for non-parabolic, ellipsoidal bands can be written as

$$\Gamma_{ac} = \frac{D_{ac}^2 m_d}{4\pi\rho \hbar^2 k^*} \int_{q_{\min}^*}^{q_{\max}^*} d\mathbf{q}^* \frac{\mathbf{q}^{*3}}{\omega_q(\mathbf{q}^*)} \Big(N_q \Big(\omega_q(\mathbf{q}^*) \Big) + \frac{1}{2} \mp \frac{1}{2} \Big) \frac{d\gamma}{dE}$$
(3.30)

where $m_d = (m_l m_l^2)^{\frac{1}{3}}$ is the density of states mass, ρ is the silicon density, k^* is magnitude of the carrier wave-vector in starred space, $\hbar \omega_q$ is the acoustic phonon energy and N_q is the phonon occupation number given by equation (3.31). Noting that the upper sign is for absorption of an acoustic phonon, the lower for phonon emission which is the convention used throughout this section. All silicon material and band parameters for this scattering rate are specified in Table 2 at the end of the chapter.

$$N_q(\omega_q) = \frac{1}{\exp\left(\frac{\hbar\omega_q}{k_B T}\right) - 1}$$
(3.31)

To evaluate the scattering rate given by equation (3.30) the integral must be completed hence requiring that \mathbf{q}_{\min}^* and \mathbf{q}_{\max}^* be found. These bounds on the integral can be found from energy conservation using equation (3.32) [84].

$$\cos\theta = \mp \frac{\mathbf{q}^*}{2\mathbf{k}^*} + \frac{m_d \omega_q \left(\mathbf{q}^*\right)}{\hbar \mathbf{q}^* \mathbf{k}^*} \left(1 + \alpha \left(2E \pm \hbar \omega_q \left(\mathbf{q}^*\right)\right)\right)$$
(3.32)

Here the RHS can be equated to a function $f(\mathbf{q}^*)$ and using the definition that $-1 \le \cos\theta \le 1$ allows the values of \mathbf{q}^*_{\min} and \mathbf{q}^*_{\max} to be defined as $f(\mathbf{q}^*_{\min})=1$ and $f(\mathbf{q}^*_{\max})=-1$. Here and in the scattering rate the phonon frequency, ω_q , as a function of the momentum transfer, \mathbf{q}^* , is determined from equation (3.33)

$$\omega_{q}\left(\mathbf{q}^{*}\right) = \langle u \rangle \left|\mathbf{q}^{*}\right| + c \left|\mathbf{q}^{*}\right|^{2}$$
(3.33)

where $\langle u \rangle = \frac{1}{3} (2u_t + u_l)$ is the average velocity and $c = -2 \times 10^{-3} cm^2/s$ is a coefficient for the dispersion relation given by [84].

3.3.2 Optical Phonons

Optical phonon scattering also uses a deformation potential scattering methodology to simplify the description of the oscillating atoms in the crystal lattice. Optical phonons describe crystal oscillations at a much higher frequency than those of acoustic phonons and correspondingly have a much larger phonon energy.

In the MC simulator there are two distinct mechanisms for optical phonon scattering. The first to be considered here is intervalley scattering between equivalent valleys given by equation (3.34) [71].

$$\Gamma_{op} = \theta \left(E_{f} \right) \frac{m_{d}^{\frac{3}{2}} \left(D_{t} K \right)^{2}}{2^{\frac{1}{2}} \pi \rho \hbar^{3} \omega_{op}} \left(N_{q} \left(\omega_{op} \right) + \frac{1}{2} \mp \frac{1}{2} \right) \gamma^{\frac{1}{2}} \left(1 + 2\alpha E_{f} \right)$$
(3.34)

Here the upper signs are for absorption and the lower signs for emission of an optical phonon, $E_f = E \pm \hbar \omega_{op}$ is the final carrier energy with $\hbar \omega_{op}$ the optical phonon energy, $D_t K$ is the optical coupling constant, $\gamma = E_f (1 + \alpha E_f)$ is the non-parabolic carrier energy term and N_q is the phonon occupation number given by equation (3.31). Again all material and scattering constants used within the MC simulator are presented in Table 2.

The second optical phonon scattering mechanism is that of the f- and g-type phonons, which is also an intervalley scattering mechanism and is given by equation (3.35) [71]. The g-type phonons describe scattering between equivalent valleys and the f-type phonons describe scattering between non-equivalent valleys.

$$\Gamma_{op,i} = \theta \left(E_f \right) \frac{m_d^{\frac{3}{2}} \left(D_t K_i \right)^2}{2^{\frac{1}{2}} \pi \rho \hbar^3 \omega_{op,i}} Z_f \left(N_q \left(\omega_{op,i} \right) + \frac{1}{2} \mp \frac{1}{2} \right) \gamma^{\frac{1}{2}} \left(1 + 2\alpha E_f \right)$$
(3.35)

In this scattering rate the terms with a subscript index i relate to the particular f- or g-type phonon index, details of which are given in Table 2. The number of final valleys available for a particular phonon mode is given by Z_f and all other symbols are the same as defined previously.

3.3.3 Ionized Impurities

Ionized impurity (II) scattering is an elastic process and uses the screened Coulomb potential as the scattering potential. The screened Coulomb potential can be found from a solution of Poisson's equation for a point charge in a charge neutral region of a semiconductor (this method is presented in detail in [32] and discussed in section 2.3). The electrostatic screening introduced by the background charge is represented in the model employed here by the degenerate Debye-Hückel screening model for non-parabolic bands. This screening model is expressed as

$$k_{C} = \sqrt{\frac{e^{2}n}{\varepsilon_{Si}\varepsilon_{0}k_{B}T} \frac{\left(\mathfrak{F}_{-\frac{1}{2}}(\eta) + \frac{15}{4}\alpha k_{B}T\mathfrak{F}_{\frac{1}{2}}(\eta)\right)}{\left(\mathfrak{F}_{\frac{1}{2}}(\eta) + \frac{5}{2}\alpha k_{B}T\mathfrak{F}_{\frac{1}{2}}(\eta)\right)} \text{ for } \eta = \frac{\mathbf{E}_{F}}{k_{B}T}}$$
(3.36)

where *e* is the electronic charge, *n* the electron density, \mathfrak{F}_j is the Fermi integral of order j [54] and E_F the Fermi energy used in the reduced Fermi level η .

The II scattering model utilised in this MC simulator is Ridley's Third-Body Exclusion (TBE) model [51, 55, 60] in conjunction with an empirical fitting parameter which is calibrated such that

the bulk mobility is matched to experimental data. The TBE model, along with other important II scattering models are discussed in detail in Chapter 2 of this thesis and further discussion on the model is not repeated here. It should be highlighted that in this work a new model for II scattering is developed in Chapter 5 which is a more accurate approach than the TBE model discussed here which has been traditionally been used in this MC simulator.

To compute the TBE model it is important to be able to express the Brooks-Herring (BH) II scattering rate which is given by equation (3.37). This is the standard BH expression for non-parabolic ellipsoidal bands.

$$\Gamma_{BH} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^{\frac{3}{2}}\pi}{m_d^{\frac{1}{2}}E_{k_c}^2} \frac{\gamma^{\frac{1}{2}}}{1+4\gamma/E_{k_c}} \frac{d\gamma}{dE}$$
(3.37)

Here N_I is the impurity density and $E_{k_c} = \hbar^2 k_c^2 / 2m^*$ is the screening length represented as an energy. From Van de Roer [60] the TBE scattering model can be expressed as

$$\Gamma_{R} = \frac{\nu(\mathbf{k})}{a} \left[1 - \exp\left(-\frac{a}{\nu(\mathbf{k})}\Gamma_{BH}\right) \right]$$
(3.38)

where $v(\mathbf{k})$ is the group velocity, given by equation (3.15), for the magnitude of the carrier wavevector $k = \sqrt{2m_d^* \gamma} / \hbar$, $a = (2\pi N_I)^{-\frac{1}{2}}$ is half the average inter-ion separation distance and Γ_{BH} is given by equation (3.37).

Finally the II scattering rate used in the MC simulation can be obtained by multiplying the TBE rate by the empirical correction factor, K_n , to obtain the complete scattering rate as $\Gamma_{II} = \Gamma_R \times K_n(N_I)$ [85]. Values used for the fitting parameter are given in Table 1 which are tabulated and then interpolated for a given impurity density N_I . The empirical correction factor, K_n , is a fitting parameter which is calibrated such that the II scattering model is fitted to experimental bulk mobility data. This parameter highlights the difference between the physical model and experimental data over a wide range of impurity densities.

Table 1: Values used in empirical correction to II scattering

$\log_{10} N_I$	14	15	16	17	18	19	20	21	22
K_n	1	1	1.4	2.3	3.1	3.7	2.5	1.4	1.4

3.3.4 Interface Roughness

An important scattering mechanism for simulation of MOS devices is Interface Roughness (IR) scattering. This mechanism describes the perturbation in the potential due to roughness at the interface of the silicon substrate with the oxide insulator layer. This mechanism is only evident when (for a nMOS device) there is an inversion layer present in the device confining carriers close to the interface.

IR scattering is dependent on the electric field perpendicular to the interface, the vertical electric field, which controls the inversion charge and the corresponding confinement of carriers at the interface. As the electric field within the inversion layer varies with position, an average is introduced to simplify the scattering model. The average electric field in the inversion layer is expressed as the effective vertical field [30, 58], E_{eff} , written as [27]

$$E_{eff} = \frac{e}{\varepsilon_{si}\varepsilon_0} \left(N_{dep} + \frac{n_s}{2} \right)$$
(3.39)

where N_{depl} is the depletion charge density and n_s is the inversion carrier density.

In order to describe the interface fluctuations a statistical function is introduced. The IR scattering model uses an exponential autocorrelation function to describe the rough interface as discussed in a paper by Goodnick [86]. The form of this autocorrelation function is

$$\langle \Delta(\mathbf{r})\Delta(\mathbf{r}-\mathbf{r'})\rangle = \Delta_{rms}^2 \exp\left(-\sqrt{2}\,\mathbf{r}/L_e\right)$$
 (3.40)

where **r** represents a position in the 2D plane parallel to the interface, Δ_{rms} is the RMS amplitude of the fluctuations in the interface and L_e is their correlation length.

Making use of the Ando model [86, 87] to define the scattering rate, the completed model can be found as

$$\Gamma_{IR}(\mathbf{k}) = \frac{e^2 m_d^*}{\hbar^3} \left(E_{eff}^2 \Delta_{rms}^2 L_e^2 \right) \int_0^{2\pi} d\varphi \frac{1}{\left(1 + \frac{1}{2} L_e^2 \mathbf{q}^2 \right)^{\frac{3}{2}}}$$
(3.41)

where $\mathbf{q}^2 = 2\mathbf{k}^2(1-\cos\varphi)$ is the momentum transfer of the scattering event. For all future device simulations the RMS amplitude of fluctuation is taken as $\Delta_{rms} = 0.35nm$ with a correlation length of $L_e = 1.3nm$.

It should be noted that the at each scattering event the local effective vertical electric field is used in place of the definition given by equation (3.39). The scattering event with the local effective field is compared to the scattering rate with the definition using equation (3.39). Using a rejection technique, which is discussed in the appendix of [75], the scattering event will be allowed if the ratio is greater than a randomly selected number. This approach is similar to one discussed in a journal paper by Formicone [88].

3.4 Monte Carlo Process

3.4.1 Single Particle Monte Carlo

The single particle MC approach is the original approach to the modelling of carrier transport first developed in the 1960's. This approach is suitable for modelling transport of a semiconductor material under fixed electric fields and can be used to obtain such quantities as the drift velocity, mean carrier energy and the bulk mobility of the material.

The single particle method is an approximation of the complete system which consists of an ensemble of carriers which mutually interact. Allowing this system to be described as a system of independent carriers acting as an ensemble leads to the single particle simulation description. By simulating a single particle accelerating in an electric field and undergoing many scattering events, an approximate description of an ensemble of carriers can be found.

A typical breakdown of the single particle simulation process is given in part (a) of Figure 3.3. The program follows a simple procedure of stochastically selecting a free-flight time according to the self-scattering procedure discussed in section 3.2.5 where the particle is subject to acceleration by the electric field. Following this the particle scattering event which ends the free-flight is stochastically selected and the carrier is scattered. This procedure repeats until the simulation time, t_{sim} , has been completed.

An important stage in this process is the gathering of carrier data which of course provides the drift velocities, mean energy and other quantities of interest. This is completed at the end of each

iteration of the procedure and is generally completed as an average of the quantity of interest over the entire simulation period which can be written as [32, 71]

$$\left\langle A\right\rangle_{T} = \frac{1}{T} \sum_{\tau} \left\langle A\right\rangle_{\tau} \tau \tag{3.42}$$

where $\langle A \rangle_{\tau}$ is the quantity average over the free-flight period τ , T is total simulation time and $\langle A \rangle_{\tau}$ is the average quantity over the entire simulation period. Use of this method requires that T be long enough to ensure that the average quantity obtained can be considered an unbiased estimator, that is the results can be considered in a steady-state and independent of the initial conditions.

This approach has been used successfully to simulate many different materials and is discussed in more detail in [32, 74, 75]. It has an advantage in the simplicity of the procedure but is limited by the fact that it often requires very large simulation times to obtain stable results. This approach is not widely used in current times due to the increase in computational power which allows more advanced procedures to be implemented. Although the single particle MC is not the approach used in the simulator employed in this work, it is an important evolutionary stage of the MC procedure and explains the original background to the approach.

3.4 Monte Carlo Process



Figure 3.3: Flowchart of (a) the single particle and (b) ensemble Monte Carlo simulators.

3.4.2 Ensemble Monte Carlo

The Ensemble Monte Carlo (EMC) process is essentially an extension to the single particle approach by introduction of an ensemble of particles to the simulation. This allows the transient characteristics of carrier transport in semiconductors such as velocity overshoot to be simulated. The ensemble approach is necessary when simulating a inhomogeneous or non-stationary process although it can be used to solve stationary problems by allowing the simulation to continue until steady state.

The EMC approach is that used to simulate the properties of bulk semiconductors within this work. In particular the EMC approach is used to calibrate the simulator which is discussed in more detail in section 3.5. The program flow is described by part (b) of Figure 3.3 where the differences between this approach and the single-particle method become clear. Obviously the free-flight and scattering process is repeated for the entire ensemble of particles at each timestep.

A timestep, Δt , is no longer defined as a single free flight, τ , terminated by a scattering event and is instead defined externally as a fixed time period. Within a timestep each particle will have as many free flights terminated by scattering events as is required to fill the timestep (specified as n_{τ} free flights in the figure). In the final n_{τ} free-flight, for the case that $n_{\tau}\tau > \Delta t$, the carrier is only propagated for the time remaining in the current timestep and not for the full free-flight. This ensures that all carriers are propagated for the full timestep, Δt , only.

Similar to the single particle approach, the ensemble particle data is collected at the end of each timestep but a different procedure for obtaining results is required. An average for the quantity required is obtained at the end of each timestep from the average of the ensemble of particles [32, 71], which can be written as

$$\left\langle A(t)\right\rangle_{N} = \frac{1}{N} \sum_{n} A_{n}(t)$$
(3.43)

Here A_n is the quantity of particle number *n* at a time *t*, *N* is the total number of particles and $\langle A(t) \rangle_N$ is the quantity average over all particles at the given time. The time average of the quantity can be used to reduce the statistical error in ensemble simulations.

The number of particles in the ensemble is typically specified at the start of the simulation and is related to physical number of carriers in the device through the superparticle approach. Using the number of charges in the device, N, the superparticle charge can be found using [30]

$$Q = -e\frac{N}{N_{sim}}$$
(3.44)

where N_{sim} is the number of particles in the simulation. The superparticles are only considered in terms of the charge density and for all other calculations such as scattering the superparticles are treated as single carriers.

3.4.3 Device Monte Carlo

Simulation of semiconductor devices such as MOSFETs requires a more advanced technique for EMC simulation which allows for non-stationary transport through inhomogeneous materials. The Device Ensemble Monte Carlo (DEMC) approach is possible through extension of the EMC procedure by inclusion of a self-consistent solution of the electrostatic potential. A flowchart of the

3.4 Monte Carlo Process

general procedure of the DEMC approach is given in Figure 3.4 showing the extension to the EMC approach. In this figure the EMC procedure is essentially that of Figure 3.3 (b) without the feedback loop over all the timesteps which is re-specified in Figure 3.4.



Figure 3.4: Flowchart for a device-ensemble Monte Carlo technique.

An important stage in the DEMC procedure is the specification of the boundaries as particles in a device are bounded. These boundaries must also be consistent with the solution for the electrostatic potential, which will be discussed later. Bulk material simulation such as those of the single-particle and EMC approaches use a boundless simulation (one that assumes an infinite material) but for a device a particle must be appropriately treated when reaching a surface. Typically there are two boundary conditions, one for a particle reaching an outer boundary for which the particle velocity normal to the surface is reflected. This boundary condition relates to the Neumann boundary condition for the electrostatic potential, that is a zero electric field component normal to the boundary surface.

The second boundary is for a particle reaching a contact where it is allowed to be absorbed. Of course as particles leave the device through a contact, particles must be injected to ensure that charge neutrality is kept within the contacts. The corresponding boundary condition for the electrostatic potential is the Dirichlet boundary which states that the boundary potential be set at the applied bias potential for the contact.

3.4 Monte Carlo Process

During the simulation, particles follow the EMC procedure of free-flight followed by a scattering event until the ensemble has completed the timestep. This movement of the particles causes the charge density to evolve with the simulation and will in turn cause the electrostatic potential to change. Before solution of the electrostatic potential can be completed the updated position of the particles and hence the charge density must be updated.

The DEMC simulator uses a Particle-Mesh (PM) method to resolve the particle charges to a mesh and define the forces for each particle. A full discussion on the PM technique can be found in the textbook by Hockney and Eastwood [89], here only a very brief discussion on the charge assignment mechanism will be undertaken. The particle charges are assigned to the mesh using a Cloud-In-Cell (CIC) technique originally developed for plasma simulations [90]. The CIC approach allows each particle charge to be assigned to the two nearest neighbour nodes which aids the smoothing of the forces and reduces the amplitude of fluctuations [71].

The electrostatic potential is found by solving Poisson's equation which relates the spatially varying charge density to the potential and is given by equation (3.6). There is extensive documentation in textbooks such as [32, 71, 89] regarding the solution of this equation in a PM system which is not repeated here. With the PM method the charge density is known at each mesh point thereby allowing the Poisson equation to be discretized over the mesh and solved using a finite difference approximation. Once the potential is found for each mesh point it is possible to define the electric fields and the corresponding forces for each particle.

When utilising a self-consistent PM approach the timestep, Δt , and the mesh spacing, Δx , must be considered to ensure that the DEMC simulation is stable. The timestep stability criterion is related to the plasma frequency for the highest carrier density, n, specified within the device model. The timestep criterion is given as [91, 92]

$$\Delta t \le \frac{2v_C}{\omega_p^2} \tag{3.45}$$

where v_c is the momentum relaxation rate and the plasma frequency, ω_p , can be written

$$\omega_p = \sqrt{\frac{e^2 n}{\varepsilon_{si} \varepsilon_0 m^*}}$$
(3.46)

Taking a typical MOSFET device the peak carrier density can be estimated at $n \approx 5 \times 10^{19} cm^{-3}$ giving the plasma frequency as $\omega_p \approx 2 \times 10^{14} s^{-1}$. Estimating a typical momentum relaxation rate of $v_c \approx 2 \times 10^{15} s^{-1}$ [33], this yields a stable time step of $\Delta t \le 1 \times 10^{-13} s$.

The mesh spacing criterion for resolution of the electrostatic potential is related to the expected charge variations within the simulation. The wavelength of the charge variations is typically approximated by the degenerate Debye length given by equation (3.36) which for $n = 5 \times 10^{19} cm^{-3}$ gives a length of $\lambda_D = 0.8nm$. Hence for stable simulation the mesh spacing is generally taken such that $\Delta x < 2\lambda_D$ [93], hence a mesh spacing of $\Delta x = 0.5nm..1nm$ is suitable.

Although these stability criteria seem well specified there is a certain amount of freedom in the choice for the timestep and mesh spacing. A thorough study has recently been conducted by Palestri [91, 93] with the results given by Figure 3.5.





This figure demonstrates that a large scattering rate, v_c , helps to stabilise the simulation by damping energy oscillations [93]. For the simulation data considered here, $v_c/\omega_p \approx 10$ which certainly allows for a greater range of timesteps and mesh spacing for stable MC simulation.

As a final stability criterion it is important to ensure that the timestep and mesh spacing are a correctly chosen pair. That is, within the chosen timestep the particles will not travel through several mesh spaces leading to greater charge oscillations and an increased instability. This can be checked by calculating the distance a particle will travel during the chosen timestep, which can be estimated by the maximum group velocity of a carrier in the semiconductor ($v_{sat} \approx 1 \times 10^7 \text{ cm/s}$).

Therefore, for this example a reasonable timestep of $\Delta t = 1 \times 10^{-14} s$ can be chosen, then the maximum expected distance a particle can travel is $l_{max} = v_{sat} \Delta t \approx 1 nm$ which corresponds with the mesh spacing criterion. It should be noted that this timestep is suitable for bulk silicon simulations but for device simulations a timestep several orders of magnitude smaller is generally selected.

3.5 Monte Carlo Calibration

3.5.1 Bulk Silicon

Before using any MC simulator it is necessary to ensure that it has been calibrated against experimental data. For simulations of carrier transport in bulk silicon this requires calibration of the phonon scattering models and in the case of the II scattering model used here, the empirical correction factor discussed in section 3.3.3. The phonon scattering models are calibrated through adjustment of the corresponding deformation potential and phonon energy where there are several published sets of such data [30, 75, 94] which can be used. These parameters are chosen from the published data to match experimental energy- and velocity-field data for undoped silicon. Undoped silicon is used to remove the dependence of II scattering on the results.

To further simplify the phonon calibration process, many of the optical phonon mechanisms can be essentially frozen out by simulating the bulk silicon at a lattice temperature of 77K. This temperature is too low for most of the optical phonons which due to their high phonon energy, become statistically unimportant. Simulation at a lattice temperature of 300K then allows the full optical phonon model parameters to be calibrated. All calibrated parameters for the phonon models are given in Table 2 at the end of the chapter. In Figure 3.6, plots (a) and (b) are the results of the velocity-field and energy-field calibration respectively. Experimental data is taken from Canali [94] for an undoped sample of silicon at the temperatures of 77K and 300K.



Figure 3.6: (a) Velocity-field and (b) energy-field calibration of Monte Carlo.

The velocity-field plot shows a good calibration with experimental data for both the low and high lattice temperatures over the range of applied fields. The energy-field figure also shows a close trend with experimental data. At 77K the low-field energy is higher than experimental data which, as reported in [31], is a cause of using an analytical band structure representation. As all future simulations in this work are completed for a lattice temperature of 300K, the calibration of the phonon models is considered a close match to experimental data at the simulation temperature.

The II scattering model used in this work utilises an empirical correction factor which improves the calibration of the II model with experimental data. To calibrate the correction factor the bulk silicon mobility is matched with experimental data over a wide range of dopant densities. Here the experimental data is taken from Thurber [29] and the values used for the II scattering empirical correction factor are given in Table 1. The result of the calibration is shown in plot (a) of Figure 3.7 and is clearly a close match with experimental data as is expected with the empirical II correction. It is also worthwhile to note that at impurity concentrations lower than $N_I = 10^{15} cm^{-3}$ where II scattering is ineffective and phonon scattering dominates, that there is good agreement with experimental data.



Figure 3.7: (a) Bulk mobility and (b) universal mobility calibration of Monte Carlo.

3.5.2 Inversion Layer

To accurately simulate MOSFET devices it is important to ensure that the MC simulator is calibrated to experimental data for carrier transport in the inversion layer. This process is required to calibrate the IR scattering model with experimental data as phonon and II scattering models should remain unchanged from the bulk silicon calibration.

Inversion layer calibration is completed by matching the universal mobility of carriers in an inversion layer with experimental universal mobility data. The experimental data is taken from Takagi [27, 28] for the substrate impurity concentration of $N_I = 2 \times 10^{18} cm^{-3}$ and a applied lateral field of E = 0.5 kV / cm.

The universal mobility trend is well replicated by the MC simulation as shown in plot (b) of Figure 3.7, with the correct drop in mobility at low effective field due to II scattering. At high effective fields the IR scattering model reduces the mobility and reproduces the universal curve [27] expected from the mobility.

The calibration of the simulator with the experimental data shown in plot (b) of Figure 3.7 is considered to be suitable despite the discrepancy between simulated and experimental data at low effective fields. This is based on the simulators ability to reproduce the overall trend of the universal mobility curve, specifically at high fields where interface roughness scattering dominates. At low effective fields the universal mobility is dominated by II scattering which in the current state-of-the-art simulators is modelled by a bulk 3D approach and is not an accurate solution to the II scattering problem in the inversion layer.

3.6 Conclusion

In this chapter the MC simulation procedure has been discussed in terms of the fundamentals of the simulation approach, the scattering mechanism involved, the procedure utilised in the numerical procedure through to the calibration of the MC code used in this work.

The discussion on the fundamentals of the MC procedure included the all-important BTE which is the core problem which the numerical procedure solves. The BTE can be described as a bookkeeping function for the distribution function which describes the state of the carrier ensemble in a device and through which all important quantities of interest can be calculated. Also discussed in the fundamentals section was the description of the silicon band structure used in the simulator. An analytical description of the band structure is utilised with the non-parabolic approximation. Ellipsoidal bands are also employed with the Herring-Vogt transformation which simplifies the description by transferring to and from a *starred* space.

Scattering in the BTE and the MC simulation procedure is typically completed using Fermi's Golden Rule approach which is discussed in section 3.2.4. The scattering in MC simulations is a quantum mechanical process which uses the interaction potential of a scattering event to obtain a probabilistic scattering rate.

Finally in the fundamentals section, the self-scattering procedure was discussed which greatly simplifies the way in which particle free-flights are selected by introducing a fictitious scattering event. This advancement in numerical simulation greatly advanced numerical MC simulations by vastly reducing the complexity of free-flight calculations.

The scattering mechanisms that are required for accurate simulation of silicon MOSFET devices are presented in section 3.3. These include the intravalley acoustic phonon, the intervalley optical phonons, II scattering and IR scattering. These scattering mechanisms are the minimum required to obtain accurate simulation data for bulk silicon and silicon MOSFET devices.

The evolution of the numerical procedure used within MC simulations was discussed in section 3.4 from the original 1960's single particle approach through to the state-of-the-art self-consistent DEMC approach. In the single particle model the simulation can only be used to obtain stationary homogeneous processes. The EMC approach allows transient, inhomogeneous systems to be modelled by simulating large numbers of particles and is the modern approach to simulating bulk semiconductor characteristics. For MOSFET simulation the required inclusion of a self-consistent Poisson solution leads to the DEMC approach.

3.6 Conclusion

The MC simulator calibration with experimental data is demonstrated in the final section. There the energy and velocity characteristics of carrier transport in silicon from the numerical MC simulation are shown to be in close match with experimental data. The carrier mobility for both the bulk and universal cases is also shown to be in good agreement with the experimental data.

Table 2: Silicon band parameters and constants.

Silicon Constants	$u_l = 5.34 \times 10^3 m/s$	$\rho = 2.329 g/cm^3$			
	$u_t = 9.04 \times 10^3 m/s$	$\varepsilon_r = 11.7$			
	$D_{ac} = 9eV$	$a_0 = 5.43$ Å			
Band 1 (X-Valleys)	$m_l = 0.916m_0$	$m_t = 0.190m_0$			
	$\alpha = 0.5 eV^{-1}$	$E_G = 1.12 eV$			
Intervalley Optical					
equivalent X-valleys:	$D_t K = 1.75 \times 10^{10} eV/m$	$\hbar\omega_{op} = 43meV$			
g-type:	$Z_f = 1$				
	$\left(D_t K\right)_{g1} = 0.5 \times 10^{10} eV/m$	$\left(\hbar\omega_{op}\right)_{g1} = 12.06meV$			
	$(D_t K)_{g^2} = 0.8 \times 10^{10} eV/m$	$\left(\hbar\omega_{_{op}}\right)_{_{g2}}=18.53meV$			
	$\left(D_{t}K\right)_{g3}=3\times10^{10}eV/m$	$\left(\hbar\omega_{op}\right)_{g3} = 63meV$			
f-type:	$Z_f = 4$				
	$(D_t K)_{f1} = 0.15 \times 10^{10} eV/m$	$\left(\hbar\omega_{op}\right)_{f1}=18.96meV$			
	$\left(D_t K\right)_{f2} = 3.4 \times 10^{10} eV/m$	$\left(\hbar\omega_{op}\right)_{f2} = 47.40 meV$			
	$\left(D_{t}K\right)_{f3}=4\times10^{10}eV/m$	$\left(\hbar\omega_{op}\right)_{f3} = 59.03 meV$			

Chapter 4 Scattering Potential Calculation

4.1 Introduction

Ionized impurity (II) scattering in modern MOSFET devices has a considerable effect on device performance. A great deal of research has been put into studying the electrostatic and transport effects of atomistic scattering [63, 67] yet little work has been done on the effect of impurity scattering close to interfaces. Here the intention is to formulate a scattering potential which describes the effect of a single atomistic impurity located next to highly-doped regions. The work will focus on looking at the effect of polarisation charges on channel IIs located close to highly doped source and/or drain regions of MOSFETs. The polarisation charges which are induced in the source and/or drain regions are shown here to remotely screen channel II ions.

Note that for the purpose of this work the typically named source-channel and channel-drain junctions shall be referred to as the source and drain interfaces respectively. This definition should not be confused with the typical device context definition of the interface between the silicon substrate and the silicon dioxide layer, which is not considered in this work.

This chapter presents and discusses scattering potentials which describe an atomistic impurity interacting with a single interface and also a double interface. The single interface case represents the II ion acting with the source, the double interface case represents the II interacting with the source and the drain. The structure of this chapter is split between the two potential models developed here, with section 4.2 presenting the single interface model and section 4.3 the double interface model. As the process of calculating and then verifying the models is identical in each case, the repetition in the content presented is carefully minimised.

The calculation of the single interface model in section 4.2 begins with the system definition before defining and solving the Poisson's equation for the potential in section 4.2.1. Much of the detail in calculation is avoided in this section with the full procedure given in Appendix A. Plots of the potential isolines for the single interface solution are shown in section 4.2.2 with a brief discussion. Validation of this solution is then completed in section 4.2.3 by checking the limits and ensuring that the behaviour is as expected of the model.

As the potential derived for the single interface model is an exact analytical solution of Poisson's equation it is important to validate the result with that of a fully self-consistent, non-linear

Poisson's equation solution. This comparison is given in section 4.2.4 with a brief discussion of the method used to complete the comparison.

The solution found for the potential in the single interface model is very complex as it includes the detailed behaviour of a doped semiconductor source region. A simplified model has been calculated by taking a limit on the screening of the source, entitled the strong-screening limit and is presented in section 4.2.5. Although the use of the strong screening limit in this work simplifies the expressions involved, primarily it has been introduced to provide a worst-case scenario for this interaction. Utilising an upper limit on the interaction will allow an initial indication on the importance of remote screening effects of channel II ions on device performance.

This structure is repeated for the calculation and verification of the double interface model in section 4.3. The complete potential model is calculated in section 4.3.1 with the detailed procedure given in Appendix B. Validation of the model by means of the limits and comparison with the non-linear Poisson solver is given in sections 4.3.3 and 4.3.4 respectively. The strongly screened potential is obtained in section 4.3.5 for the double interface model.

4.2 Single Interface Potential Model

A self-consistent potential for a single impurity located close to the source –channel interface shall now be calculated by solving the Poisson equation over the source/channel region. The Linear Thomas-Fermi (LTF) approximation [95] is used to simplify this solution of the Poisson equation in conjunction with the Debye-Hückel screening model [53]. Due to the cylindrical symmetry of the system, cylindrical co-ordinates are used throughout the potential calculation [96]. Here the Zaxis is normal the source-channel interface with the **R** plane parallel to the interface.



Figure 4.1: Physical picture of the problem system, defining source and channel regions with impurity located Z_i from interface.

Figure 4.1 provides a pictorial representation of the physical system to be solved. Here the source and channel carrier densities are represented by n_s and n_c respectively. The source is assumed to have a larger carrier density in this figure, that is $n_s > n_c$ but this is not an assumption made in the calculation of the potential in section 4.2.1.

The impurity for which the potential is to be found is given by the red circle in the channel region for which Z > 0. In this case, the source interface is located at Z = 0 and the impurity is located at Z_I which must be in the channel region. To simplify the problem and incorporate radial symmetry around the Z axis, the impurity is assumed to be located at $\mathbf{R}_I = 0$.

4.2.1 Potential Solution

The solution for the electrostatic potential of this system can be found by solving the following Poisson equations for the free charge density.

$$\nabla^2 \varphi_s = -\frac{\rho_s}{\varepsilon_{si} \varepsilon_0} \quad \text{for } Z < 0 \tag{4.1}$$

$$\nabla^2 \varphi_C = -\frac{\rho_C}{\varepsilon_{S_I} \varepsilon_0} - \frac{Q}{2\pi \varepsilon_{S_I} \varepsilon_0} \frac{\delta(\mathbf{R})}{R} \delta(Z - Z_I) \quad \text{for } Z > 0$$
(4.2)

Where ρ_s is the free charge density in the source region and ρ_c the free charge density in the channel region. On the right-hand side of equation (4.2) the atomistic impurity can be seen as a charge Q at position Z_I . For this model, the impurity is positioned at $\mathbf{R}_I = 0$ in the \mathbf{R} plane. This simplifies the solution but retains the important Z dependence that controls the interaction with the source region.

The inverse bulk screening length, $k_x = \lambda_x^{-1}$, can be related to the free charge density with the LTF approximation by using the following relation

$$\rho_X = -\varepsilon_{si}\varepsilon_0 k_X^2 \varphi_X \tag{4.3}$$

Here, the subscript X denotes either the source (S) or channel (C) region. This then gives the linearized Poisson equation, also known as a Helmholtz equation as

$$\nabla^2 \varphi_s - k_s^2 \varphi_s = 0 \quad \text{for } \mathbb{Z} < 0 \tag{4.4}$$

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{2\pi\varepsilon_{SI}\varepsilon_0} \frac{\delta(\mathbf{R})}{R} \delta(Z - Z_I) \text{ for } Z > 0$$
(4.5)

Solutions to equations (4.4)-(4.5) can be found by specifying boundary conditions on the potential, $\lim_{\mathbf{R}, Z \to \pm \infty} \varphi = 0$. Using a standardized solution of the Helmholtz equation using cylindrical coordinates in conjunction with the Bessel function [96], the electrostatic potential for the source and channel region can be found exactly as

$$\varphi_{S} = \left(1 - \theta(Z)\right) \int_{0}^{\infty} dk J_{0}\left(kR\right) A_{S}\left(k\right) \exp\left(Z\sqrt{k^{2} + k_{S}^{2}}\right)$$
(4.6)

$$\varphi_{C} = \theta(Z) \int_{0}^{\infty} dk J_{0} (kR) \left\{ A_{C}(k) \exp\left(-Z\sqrt{k^{2} + k_{C}^{2}}\right) + \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{C}^{2}}} \exp\left(-|Z - Z_{I}|\sqrt{k^{2} + k_{C}^{2}}\right) \right\}$$
(4.7)

Here $\theta(Z)$ is the unit step function and J_0 is the zero order Bessel function of the first kind [97]. As we assume that $\mathbf{R}_{\mathbf{I}} = 0$ giving symmetry around the Z axis, only the magnitude of \mathbf{R} has any importance in these equations. This correctly allows the use of the magnitude of \mathbf{R} in the Bessel function. It is important to note that the use of the variable k here shouldn't be confused with the typical notation for the carrier wave vector which has the identical symbol. In this case the variable is used to denote a separation constant used within the solution of the equation.

This leaves 2 coefficients, A_s and A_c , which need to be found to complete the solution for the potentials. This can be completed by using the static boundary continuity conditions on **E** and **D** for electric fields [26] such that

$$\mathbf{E}_{\parallel S} = \mathbf{E}_{\parallel C} \text{ or } \frac{\partial \varphi_S}{\partial \mathbf{R}} = \frac{\partial \varphi_C}{\partial \mathbf{R}}$$
(4.8)

$$\mathbf{D}_{S} = \mathbf{D}_{C} \text{ or } \boldsymbol{\varepsilon}_{Si} \boldsymbol{\varepsilon}_{0} \frac{\partial \boldsymbol{\varphi}_{S}}{\partial Z} = \boldsymbol{\varepsilon}_{Si} \boldsymbol{\varepsilon}_{0} \frac{\partial \boldsymbol{\varphi}_{C}}{\partial Z}$$
(4.9)

These define that the tangential electric field, \mathbf{E}_{\parallel} , must be the same along the interface and that the electric flux, \mathbf{D} , must be continuous across the interface. After solving the resulting simultaneous equations we find the solutions to the 2 unknown coefficients as

$$A_{s}(k) = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{c}^{2}} + \sqrt{k^{2} + k_{s}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$
(4.10)

$$A_{C}(k) = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{C}^{2}}} \frac{\sqrt{k^{2} + k_{C}^{2}} - \sqrt{k^{2} + k_{S}^{2}}}{\sqrt{k^{2} + k_{C}^{2}} + \sqrt{k^{2} + k_{S}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{C}^{2}}\right)$$
(4.11)

After some manipulation, the complete solutions for φ_s and φ_c can be written in the form

$$\varphi_{s} = \left(1 - \theta(Z)\right) \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0}(kR) \frac{2k}{\sqrt{k^{2} + k_{c}^{2}} + \sqrt{k^{2} + k_{s}^{2}}}$$

$$\times \exp\left(Z\sqrt{k^{2} + k_{s}^{2}}\right) \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$

$$(4.12)$$

$$\varphi_{C} = \theta(Z) \frac{Q}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0}(kR) \frac{k}{\sqrt{k^{2} + k_{C}^{2}}} \left\{ \exp\left(-|Z - Z_{I}|\sqrt{k^{2} + k_{C}^{2}}\right) - \frac{\sqrt{k^{2} + k_{S}^{2}} - \sqrt{k^{2} + k_{C}^{2}}}{\sqrt{k^{2} + k_{S}^{2}} + \sqrt{k^{2} + k_{C}^{2}}} \exp\left(-(Z + Z_{I})\sqrt{k^{2} + k_{C}^{2}}\right) \right\}$$

$$(4.13)$$

These potential equations define the self-consistent screened solution to an II located within the channel region which is coupled to the source, Figure 4.1. Here this calculation has been summarised to the major steps, for more detail of the procedure used to find these solutions see Appendix A.

Examining the solution to the channel potential of equation (4.13), we can see the additional term present from the polarisation charge effect. Looking at the two terms within the curly brackets of equation (4.13), the second term represents the potential from the polarisation charge. The sign of the impurity location, Z_I , is negative in this second term, representing the location of the fictitious polarisation (image) charge.

4.2.2 Contour Plots

Plotting the total potential given by equation (4.14) in some contour plots allows for examination of the effect of the interface on an atomistic impurity. Figure 4.2 shows contour plots of the total potential with an atomistic impurity located in four different locations in the channel region, Z > 0.

$$\varphi = \varphi_{\rm S} + \varphi_{\rm C} \tag{4.14}$$



Figure 4.2: Contour plots of the potential for the single interface model. Plots (a)-(d) show four different impurity locations, Z_I , where the channel screening length, $\lambda_c = 4.133nm$, and the source screening length is $\lambda_s = 0.69nm$.

In Figure 4.2, the source region doping concentration is $N_I = 10^{20} cm^{-3}$ and the channel concentration is $N_I = 10^{18} cm^{-3}$ which yields screening lengths of $\lambda_s = 0.69nm$ and $\lambda_c = 4.133nm$ respectively. The screening lengths are calculated using the degenerate Debye-Hückel screening model, equation (3.36).

In plot (d) the impurity is located at $Z_I = 16nm$ from the source-channel interface and we see very little effect of the interface on the resulting potential isolines. This can be explained by the exponential roll-off of the polarisation charge effect with respect to the channel screening length and impurity position. At a positon of $Z_I \approx 4\lambda_C$, the polarisation charge term is very small, e.g. $1 - \exp(-4) \approx 0.98$.

The case where the impurity is located far enough away from the interface so as to introduce no polarisation charge effects, $Z_I > 4\lambda_c$, will be termed the screened Coulomb limit. This is so called as the potential represents only the screened Coulomb component where the polarisation charge term tends to zero. This will be further demonstrated in section 4.2.3 where this limit and others will be analysed.

At an impurity position of roughly two screening lengths from the interface, the potential contours in plot (c) show the increased screening effect induced from the interaction with the polarisation charge. This effect increases in plot (b) when the impurity is positioned roughly one screening length away. Notice the penetration of the impurity potential into the source region at this distance is minimal despite there being a strong interaction with the polarisation charge term. At $Z_I \approx 0.1 \lambda_C$ from the source-channel interface, plotted in (a), the potential contours are highly distorted and the potential is being heavily screened by the polarisation charge. There is strong penetration of the impurity potential into the source region where, due to the much smaller screening length, the potential drops off much more rapidly.

4.2.3 Limits of Potential

It is important to confirm that the calculated potential is valid and yields the correct behaviour. This can be checked by testing the various limits of the equations (4.12)-(4.13) to ensure that the boundary conditions are properly held and that the expected result is obtained.

4.2.3.1 Screened Coulomb Limit

The first limit to check is to ensure that the model returns to the classic screened Coulomb potential when the impurity is located a large distance away from the source-channel interface, $Z_I k_c \gg 4$. Performing this limit on equations (4.12)-(4.13) gives

$$\lim_{Z_I k_C \gg 4} \varphi_S \simeq \left(1 - \theta(Z)\right) \exp\left(-Z_I \sqrt{k^2 + k_C^2}\right) = 0$$
(4.15)

$$\lim_{Z_I k_C \gg 4} \varphi_C = \theta(Z) \frac{Q}{4\pi\varepsilon_{SI}\varepsilon_0} \int_0^\infty dk J_0(kR) \frac{k}{\sqrt{k^2 + k_C^2}} \exp\left(-\left|Z - Z_I\right| \sqrt{k^2 + k_C^2}\right)$$
(4.16)

The source potential term is dominated by the negative exponential which tends to zero in the limit $Z_I k_C \gg 4$. This is consistent with theory that if the impurity is a great distance from the source, there will be no interaction with polarisation charges in this region. Similarly the polarisation charge term within the channel potential, the second exponential on the RHS of equation (4.13), will tend to zero.

After use of the transformation theorem given by equation (4.17) [98] the channel potential can be given by equation (4.18). This form is identical to the screened Coulomb model with no boundary interactions such as polarisation effects.

$$\int_{0}^{\infty} \frac{x}{\sqrt{x^{2} + a^{2}}} J_{0}(xy) \exp\left(-b\sqrt{x^{2} + a^{2}}\right) dx \equiv \frac{\exp\left(-a\sqrt{b^{2} + y^{2}}\right)}{\sqrt{b^{2} + y^{2}}}$$
(4.17)

$$\lim_{Z_I \gg 4k_C^{-1}} \varphi_C = \theta(Z) \frac{Q}{4\pi\varepsilon_{S_I}\varepsilon_0} \frac{\exp\left(-k_C \sqrt{\mathbf{R}^2 + (Z - Z_I)^2}\right)}{\sqrt{\mathbf{R}^2 + (Z - Z_I)^2}}$$
(4.18)

4.2.3.2 Matched Screening Limit

Ensuring that in the limit of matched screening in the source and channel regions, $k_s = k_c$, the polarisation charge terms disappear is the purpose of this check. This is the case if there is no boundary introduced via the screening in the source and channel region, then formation of polarisation charges is impossible.

$$\lim_{k_s=k_c} \varphi_s = \left(1 - \theta(Z)\right) \frac{Q}{4\pi\varepsilon_{si}\varepsilon_0} \int_0^\infty dk J_0(kR) \frac{k}{\sqrt{k^2 + k_c^2}} \exp\left(\left(Z - Z_I\right)\sqrt{k^2 + k_c^2}\right)$$
(4.19)

$$\lim_{k_{S}=k_{C}}\varphi_{C} = \theta(Z) \frac{Q}{4\pi\varepsilon_{S_{I}}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0}(kR) \frac{k}{\sqrt{k^{2}+k_{C}^{2}}} \exp\left(-\left|Z-Z_{I}\right|\sqrt{k^{2}+k_{C}^{2}}\right)$$
(4.20)

Completing the matched screening limit on the source and channel potential terms gives the equations (4.19) and (4.20). As expected the model returns to a screened Coulomb potential split over the source and drain regions. Again, if the impurity is located a large distance from the source region, the source potential will tend to zero as in the screened Coulomb limit discussed in the previous sub-section.

As the matched screening limit provides a solution of the screened Coulomb potential over the two regions, it is convenient to check the continuity of the solution at the interface. Taking the limit of Z = 0 on equations (4.19)-(4.20) and equating yields after some manipulation

$$\int_{0}^{\infty} dk \frac{J_{0}(kR)k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right) = \int_{0}^{\infty} dk \frac{J_{0}(kR)k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$

$$\varphi_{S}\left(Z = 0\right) = \varphi_{C}\left(Z = 0\right)$$
(4.21)

Equation (4.21) clearly shows that the potentials match at the interface.

4.2.4 Comparison with Non-Linear Poisson Solution

As the newly-developed model for the potential of an impurity located close to a region of highdoping solves a linearized form of the Poisson equation, it is important to compare this with a numerical Poisson solver. This will allow for an accurate test of the quality of the solution obtained here. First the discrete impurity method will be discussed and demonstrated before the detailed comparison between the solution is completed.

4.2.4.1 Discrete Dopant Simulation

A fully self-consistent, Non-Linear Poisson (NLP) solution will be used to simulate an ideal device of a single atomistic impurity located close to an interface. The method used to solve for atomistic impurities with the Poisson equation is discussed in detail in the paper by Asenov [63]. The method involves including a single dopant via the mesh-resolved charge distribution by assigning the chosen mesh cell to contain the charge density e/h^3 where h is the mesh spacing. This meshresolved charge distribution is used within the Poisson solver to calculate the electrostatic potential for the system.

For this experiment a device has been constructed which closely represents the ideal system used to develop the single interface potential, Figure 4.1. This device takes the form of a highly doped n-region located adjacent to a p-doped region and is depicted in Figure 4.3.



Figure 4.3: Figure demonstrating the doping profile of the Poisson test device for the single interface model. The doping transition from $N_D = 10^{20} cm^{-3}$ to $N_A = 10^{18} cm^{-3}$ is assumed to be abrupt.

The device has as 20nm square body and is 70nm long which is split into 10nm for the highlydoped source region and 60nm for the channel. Doping in the source region is given as $N_D = 10^{20} cm^{-3}$ and the channel is doped at $N_A = 10^{18} cm^{-3}$ with an abrupt doping transition between the regions. To improve the quality of this experiment a slight modification is made to the normal atomistic doping process. As the remote screening model developed here assumes a linear background charge density with a single atomistic impurity, a similar assumption is made when resolving the charge distribution for the Poisson solution. Hence, the inclusion of a single atomistic dopant is assumed not to alter the surrounding dopant concentration per unit volume of the simulated system.

Examining this assumption in more detail using equation (4.22) it is found that this assumption has little effect on the background doping. In the system described above the channel region has a volume of $V = 24 \times 10^{-18} cm^3$ with a dopant density $N_I = 10^{18} cm^{-3}$. Using the equation below, this leads to 24 dopant atoms in the channel volume. Adding one further dopant to the channel region roughly corresponds to a background doping concentration of $N_I = 1.042 \times 10^{18} cm^{-3}$ which is less than a 5% shift in doping.

$$n_{ion} = V \times N_I \tag{4.22}$$

A mesh is applied to the structure with a node resolution of 0.25nm. This fine mesh is important to ensure that the atomistic impurity is accurately resolved within a discretized Poisson solution [99]. The channel length is chosen to be long enough to accommodate the source interface depletion width of $W_{dm} \approx 36.71nm$ calculated using the long-channel formula given in equation (4.23) [15].

$$W_{dm} = \sqrt{\frac{2\varepsilon_{si}\varepsilon_0 \left(E_g/2 + k_B T \ln\left(N_A/n_i\right)\right)}{eN_A}}$$
(4.23)

The reference channel screening length is calculated using the Debye-Hückel screening model, equation (3.36), as $\lambda_c = 4.133 nm$ where the carrier concentration is assumed to be fixed at the background dopant concentration, $n = N_A = 10^{18} cm^{-3}$. This reference screening length is used only to provide a fixed length scale to measure the position of the impurity in the channel.

The potential solution for this device with a single atomistic impurity located at four different locations is shown in Figure 4.4. In part (a) of this figure, the atomistic impurity is placed at $Z_I = 52nm$ which is beyond the end of the depletion region. This is to ensure that the atomistic impurity will be minimally affected by the source interface. The potential isolines for this impurity are spherical close to the impurity and slowly disperse farther from the impurity centre.




In parts (b) and (c), the atomistic impurity is located around $4\lambda_c$ and $1\lambda_c$ from the source-channel interface. The potential contour lines change from being spherical to being teardrop shaped as the impurity is located closer to the interface. Clearly the vicinity of the interface is altering the screening of the atomistic impurity potential as is expected.

In part (d) of Figure 4.4, the impurity is located directly next to the interface at roughly $Z_I = 0.1\lambda_c$. At this distance the polarisation charge effect is very strong as the potential isolines demonstrate. The impurity is heavily screened and the impact of the impurity is felt over a cross-sectional area much smaller than the case where the interface plays little role such as in part (a) of this figure. These plots show a consistent result with the effect modelled by the single interface potential calculated earlier in this chapter, Figure 4.2.

4.2.4.2 Comparison with Analytical Model

To ensure that the calculated analytical potential is accurately modelling the impurity it is important to compare the results in more detail. This is possible by a comparison of the analytical model with the NLP solution from the previous section. In Figure 4.5, 1D slice plots of the potential obtained from the NLP solver are compared to the analytic solution calculated in section 4.2. As the analytical solution uses a linearized Poisson solution, the comparison in the models will differ as the full Poisson solution will include the depletion region of the p-n junction. This makes the comparison difficult but a simplifying solution has been utilised to compare the impurity potential solutions between the models.

Solving the model system described by Figure 4.3 with the NLP solver and with no atomistic impurities, a uniform device, provides a solution of the potential for the p-n junction. The potential and carrier density profiles for this uniform device solution can then be extracted giving a solution for the depletion region. Through the use of superposition the uniform device potential can be added to the analytical model to provide a solution comparable to the NLP solver. Likewise the uniform solution could have been subtracted from the NLP solver to compare the impurity potential alone, but the current method is perhaps closer to real simulation conditions.



Figure 4.5: Comparison of the potential of a single atomistic impurity using a fully selfconsistent, non-linear Poisson solution and the single interface, analytical solution obtained in section 4.2.

For all impurity positions in plots (a)-(d) of Figure 4.5, the analytical solution is found to be very close to the discretized NLP solution. The largest difference is in the resolution of the singular peak of a Coulomb point charge, the atomistic impurity centre, which is a known drawback of the discretized Poisson solution.

To be thorough in this comparison, a single impurity position from Figure 4.5 is chosen to complete a comparison in the axis parallel to the interface. This comparison is shown in Figure 4.6 for $Z_1 = 2\lambda_c$ and again shows the close agreement between the models. Again there is a discrepancy in the singular peak of the point charge due to the discretization error induced from the numerical solution of the Poisson equation.



Figure 4.6: Comparison of the non-linear Poisson and analytical solutions of a single atomistic impurity located at $Z_I = 2\lambda_c$. Discrepency between solutions of point charge due to numerical discretisation of Non-Linear Poisson solution at mesh spacing $\Delta x = 0.5nm$.

This comparison demonstrates that the calculated analytical model accurately provides a method to model the polarisation charge effect of an atomistic impurity located close to an region of higher doping.

4.2.5 Strong-Screening Limit

A set of equations has now been developed which model the potential of a single atomistic impurity located close to a reflecting interface. From the potential equation it is possible to develop a scattering mechanism which can be used in Monte Carlo simulation. The development of such a scattering mechanism and the application to Monte Carlo simulation will be the subject of the following chapters.

In their current form the potential solution for the single interface model, given by equation (4.14), is quite large and unwieldy. The calculation of a Monte Carlo appropriate scattering mechanism is a daunting procedure and is challenging to complete with the potential described above. A simplified form of the potential has been found which allows a straightforward calculation of the scattering mechanism, yet retains as much of the complete physical model as possible. This simplified model has been obtained here by using a limit which is appropriate to the model system that the potential solutions have been developed for.

This limit assumes that the source region is highly degenerately doped, becoming metallic-like, and the corresponding screening length becomes very much less than the channel screening length, $\lambda_s \ll \lambda_c$. This limit has been termed the strong-screening limit and is shown here to simplify the potential solution. Using the strongly screened limit constitutes a model which assumes the worst

case scenario for this interaction in that the source region is a metal. This, as discussed below in section 4.2.5.2, leads to over estimation of the remote screening effect of the induced polarisation charges. For the purpose of this work, which is to study what effect polarisation charges have on modern device performance, this limit is considered viable. It is worthwhile mentioning that the strong-screening limit essentially reduces the problem to the classical image charge problem [100].

4.2.5.1 Strongly Screened Single Interface Model

The strong-screening limit for the single interface model greatly reduces the complexity whilst still retaining the important characteristics. Here it can be redefined as $k_s \gg k_c$ using the inverse screening length. Using the potential solutions from section 4.2, equations (4.12) and (4.13), and taking the limit yields

$$\lim_{k_s \gg k_c} \varphi_s = (1 - \theta(Z)) \frac{Q}{4\pi\varepsilon_{s_i}\varepsilon_0} \int_0^\infty dk J_0(kR) \frac{2k}{k_s} \exp(Z k_s) \exp(-Z_I \sqrt{k^2 + k_c^2})$$

$$\lim_{k_s \gg k_c} \varphi_s \simeq \frac{\exp(Zk_s)}{k_s} = 0 \quad \text{for } Z < 0$$
(4.24)

The source potential tends to zero as all of the impurity potential is screened out in a very small region on the source side of the interface. This is expected as the screening in the source becomes very strong.

The limit of the channel potential seems less effective but simplifies the coefficient of the polarisation charge term, the second term within the square brackets. Essentially, the strength of the polarisation charge term no longer depends on the ratio between the channel and source screening as the source is assumed to be an almost perfect reflecting surface.

$$\lim_{k_{S}\ggk_{C}}\varphi_{C} = \theta(Z)\frac{Q}{4\pi\varepsilon_{SI}\varepsilon_{0}}\int_{0}^{\infty} dkJ_{0}(kR)\frac{k}{\sqrt{k^{2}+k_{C}^{2}}}\left[\exp\left(-|Z-Z_{I}|\sqrt{k^{2}+k_{C}^{2}}\right)\right]$$
$$-\frac{k_{S}}{k_{S}}\exp\left(-(Z+Z_{I})\sqrt{k^{2}+k_{C}^{2}}\right)\right]$$
$$\lim_{k_{S}\ggk_{C}}\varphi_{C} = \theta(Z)\frac{Q}{4\pi\varepsilon_{SI}\varepsilon_{0}}\int_{0}^{\infty} dkJ_{0}(kR)\frac{k}{\sqrt{k^{2}+k_{C}^{2}}}\left[\exp\left(-|Z-Z_{I}|\sqrt{k^{2}+k_{C}^{2}}\right)\right]$$
$$-\exp\left(-(Z+Z_{I})\sqrt{k^{2}+k_{C}^{2}}\right)\right]$$
$$(4.25)$$

The strong screening limit allows the form of equation (4.25) to be further simplified by using the theorem, given by equation (4.17), which removes the integral.

$$\lim_{k_{S} \gg k_{C}} \varphi_{C} = \theta(Z) \frac{Q}{4\pi\varepsilon_{Si}\varepsilon_{0}} \left[\frac{\exp\left(-k_{C}\sqrt{\mathbf{R}^{2} + (Z - Z_{I})^{2}}\right)}{\sqrt{\mathbf{R}^{2} + (Z - Z_{I})^{2}}} - \frac{\exp\left(-k_{C}\sqrt{\mathbf{R}^{2} + (Z + Z_{I})^{2}}\right)}{\sqrt{\mathbf{R}^{2} + (Z + Z_{I})^{2}}} \right]$$
(4.26)

This simplified form of the single interface potential consists of the recognisable screened Coulomb potential, the first term in the square bracket, minus the potential of the polarisation charge, the second term. The polarisation charge term can be distinguished by the change in the sign of the impurity position, Z_I . As the source potential has been screened to zero with the limit of strong-screening, the total potential becomes simply the channel potential. It is important to note that this potential is only valid for the channel region, that is for Z > 0.

4.2.5.2 Verification of Limit

Although the strong-screening limit provides a simpler model, it is necessary to check that the model still provides an accurate representation of the system. This can be easily shown by comparison of the potential for the complete model against that of the simplified strongly screened model. Such a comparison of the remotely screened impurity potentials is made in Figure 4.7.

For this comparison an ideal example device has been used which has a source region located at Z < 0 doped at $N_I = 10^{20} cm^{-3}$ and a channel region, Z > 0, with doping at three concentrations, $N_I = \{10^{14}, 10^{16}, 10^{18}\} cm^{-3}$. The screening length for these three channel densities has been calculated using the screening model given by equation (3.36) and corresponds to $\lambda_c = \{409, 41, 4.1\} nm$ respectively. The source region is only considered in the complete model as the strongly screened model screens out the potential in this region. Hence for the figures in this section the source region is not plotted.



Figure 4.7: Comparison of the complete and strongly screened potentials for four impurity positions.

In Figure 4.7 are plots of the impurity potential at four different locations in the channel. In these plots the potential is given for both the complete and the strongly screened models over the three different channel impurity densities. In each plot of Figure 4.7 the strongly screened model shows an increased drop of the potential at the source interface. In plots (c) and (d) this drop in potential becomes appreciable at a distance roughly less than one nanometre from the source interface. When the impurity is located at one nanometre or less from the interface, as in plots (a) and (b), this drop in potential becomes quite large.

The increased screening of the potential at the source interface is expected from the strongly screened model as this limit forces the potential in the source region to zero. This acts as a strong boundary condition for the channel potential which ensures that the channel impurity charge is neutralised at the interface. The validity of this approach can be roughly analysed from the behaviour shown in Figure 4.7. As discussed above the limit induces a large shift for impurities very close to the interface. Examining the potential comparisons in more detail in the following figures will allow a better quantitative analysis.



Figure 4.8: Detailed comparison of complete and strongly screened potential models for $Z_1 = 5nm$.

At an impurity distance of $Z_1 = 5nm$, as shown in Figure 4.8, the strong screening limit is adequate with a shift in impurity potential of around 5mV at a tenth of a nanometre from the interface. Although this corresponds to a 100% error at the interface, the potential can be considered to be well screened at this distance and the strongly screened model is quite accurate. Beyond the impurity into the channel region, the limit yields an error of ~5-10% or less than a millivolt shift in potential.

Interestingly at this impurity position the channel impurity density has a noticeable effect on the quality of the strong screening limit. The higher the impurity density in the channel, and therefore the higher the screening density the closer the models are. This can be explained by the increased screening of the impurity potential in the channel minimising the interaction with the source as seen in Figure 4.7, plot (d).



Figure 4.9: Detailed comparison of complete and strongly screened potential models for $Z_1 = 1nm$.

4.2 Single Interface Potential Model

At the impurity position $Z_1 = 1nm$, the detailed comparison plots of Figure 4.9 show a much larger shift in the potential, roughly 70mV at the interface. The percentage error plot of (b) highlights the problem of the strong screening limit at this distance from the source. Around this impurity position the error increases rapidly, but worryingly the error beyond the impurity into the channel bulk increases to 30-40%. At this distance the validity of this limit under these conditions is hard to justify given this increase in error.



Figure 4.10: Detailed comparison of complete and strongly screened potential models for $Z_1 = 0.5 nm$.

Examining the effect of the limit at a closer impurity position in Figure 4.10 highlights the increasing error. The drop in the potential becomes very significant, rising to well over 150mV. The error induced in the bulk of the channel, beyond the impurity also show a drastic increase to almost 50%. At this impurity position, within 1nm of the interface, the strong screening limit for the source does not provide an accurate solution of the potential.

Looking at Figure 4.7 it is clear that the error in the approximation becomes appreciable within roughly a 1nm region from the source interface. For impurities located outwith this region, the error constitutes a small potential drop. From this analysis the limit has been shown to be close to the complete model over a range of channel impurity concentrations and impurity positions greater than 1nm of the interface. Modelling impurities within 1nm of the interface leads to an overestimation in the screening of the potential.

To further the verification of the strongly screened limit it is appropriate to compare the two models in a context of effect on device behaviour. It is difficult to estimate the effect that the strongly screened limit will have on carrier scattering from the comparisons on potential alone and therefore an analysis on ionized impurity limited mobility is ideal. A numerical calculation of the impurity scattering limited mobility using the momentum relaxation rates of the complete and strongly screened models is shown in Figure 4.11. Although the discussion and use of momentum

relaxation rates in this chapter is a little out of sequence in the structure of this PhD thesis (see Chapter 5), it is necessary for the analysis of the mobility discussed here.

The electron mobility is calculated using the Kubo-Greenwood formula [33], equation (4.27), assuming a spherical, parabolic band structure. The momentum relaxation rate for the strongly-screened model is presented in Chapter 5 (assuming the non-parabolicity parameter $\alpha = 0$) and Appendix C presents the momentum relaxation rate for the complete single interface model.

$$\mu = \frac{2e}{3m^*n} \int_{0}^{\infty} dE\rho(E) E\tau_m(E) \left(\frac{\partial f_0}{\partial E_F}\right)$$
(4.27)

Here *e* is the electronic charge, *n* is the electron density, $\rho(E)$ is the density of states, f_0 is the equilibrium Fermi function and E_F is the Fermi energy. The electron density and the density of states can be written for parabolic bands as

$$n = 2 \left(\frac{m^* k_B T}{2\pi\hbar^2}\right)^{\frac{3}{2}} \mathfrak{F}_{\frac{1}{2}}(\eta) \quad \text{for } \eta = \frac{\mathbf{E}_F}{k_B T}$$
(4.28)

$$\rho(E) = \frac{3(2m^*)^{\frac{3}{2}}}{\pi^2 \hbar^3} E^{\frac{1}{2}}$$
(4.29)

In Figure 4.11 the ratio of the mobility between the two models is plotted against the position, Z_I , of a single II. The mobilities are calculated at three different background channel impurity concentrations where the carrier density is assumed to equal to impurity density in each case. Similar to the potential comparisons given above, the source impurity/carrier density is given as $n = N_I = 10^{20} cm^{-3}$ and is only referenced in the complete model.



Figure 4.11: Mobility comparison of the complete and strongly screened models assuming a single ionized dopant at a range of positions Z_i , given three background doping concentrations.

It is clear from this plot that the strongly screened model does in fact become increasingly invalid when the impurity is located within 1nm of the source interface regardless of channel screening density. This is very much consistent with the conclusion of the potential comparison that the strongly screened limit will lead to an over-estimation of the effect of remote screening. At 1nm the strongly screened model gives a mobility approximately just over one and a half times larger than that of the complete model. This increases rapidly at decreasing impurity distance from the interface with a ratio of roughly three at $Z_1 = 0.5nm$ and a peak of over twenty in the plot above.

Considering the context of this work which is to model the polarisation charge effect on ionized impurity scattering, the use of the strongly screened limit is deemed acceptable to obtain a worst-case value. Use of the strongly screened model will correctly yield an upper limit on the effect of this interaction in a device simulation as has been completed in Chapter 6.

4.3 Double Interface Potential Model

The double interface potential can be calculated following the identical procedure as the single interface model. Cylindrical co-ordinates are used throughout and the LTF approximation, in conjunction with the Debye-Hückel screening model, is used to simplify the Poisson solution. The physical system is defined in Figure 4.12 with the source interface located at Z = 0 and the drain interface located at $Z = L_c$. In this figure the source and drain regions have carrier densities larger than the channel, $n_s, n_D > n_c$. Again the impurity is assumed to be always located at $\mathbf{R}_I = 0$ to incorporate radial symmetry around the Z axis, simplifying the solution.



Figure 4.12: Pictorial representation of the double interface system with the impurity located at Z_1 and the channel length is given by L_c .

The II is given by the red circle in the channel region and it's position is defined relative the the source and drain interfaces, in this case Z_I and $(L_C - Z_I)$ respectively where L_C denotes the channel length.

4.3.1 Potential Solution

The Poisson equations defining the double interface system of Figure 4.12 using the LTF approximation can be given as

$$\nabla^2 \varphi_s - k_s^2 \varphi_s = 0 \quad \text{for } \mathbf{Z} < 0 \tag{4.30}$$

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0} \frac{\delta(\mathbf{R})}{R} \delta(Z - Z_I) \quad \text{for } 0 < Z < L_C$$
(4.31)

$$\nabla^2 \varphi_D - k_D^2 \varphi_D = 0 \quad \text{for } Z > L_C \tag{4.32}$$

Using similar boundary conditions on the potential as the single interface model, $\lim_{\mathbf{R}, Z \to \pm \infty} \varphi \to 0$, and using the Bessel functions within cylindrical co-ordinates, the following solutions can be found

$$\varphi_{s} = \left(1 - \theta(Z)\right) \int_{0}^{\infty} dk J_{0}\left(kR\right) A_{s}\left(k\right) \exp(ZK_{s})$$
(4.33)

$$\varphi_{c} = \left(\theta(Z) - \theta(Z - L_{c})\right) \int_{0}^{\infty} dk J_{0} \left(kR\right) \left[\frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{c}} \exp\left(-|Z - Z_{I}|K_{c}\right) - A_{c} \left(k\right) \exp\left(-ZK_{c}\right) - B_{c} \left(k\right) \exp\left(-ZK_{c}\right)\right]$$

$$(4.34)$$

$$\varphi_D = \theta \left(Z - L_C \right) \int_0^{\infty} dk J_0 \left(kR \right) A_D \left(k \right) \exp \left(-ZK_D \right)$$
(4.35)

Here the simplified terms , K_x , are given by equations (4.36)-(4.38) below.

$$K_{C} = \sqrt{k^{2} + k_{C}^{2}}$$
(4.36)

$$K_{s} = \sqrt{k^{2} + k_{s}^{2}}$$
(4.37)

$$K_{D} = \sqrt{k^{2} + k_{D}^{2}}$$
(4.38)

The four coefficients, $\{A_s, A_c, B_c, A_b\}$, can be found by matching conditions at the interfaces using a simplified set of the electric field continuity equations specified in equation (4.39).

$$\frac{\partial \varphi_s}{\partial Z} = \frac{\partial \varphi_c}{\partial Z} \text{ and } \frac{\partial \varphi_c}{\partial Z} = \frac{\partial \varphi_D}{\partial Z}$$

$$\frac{\partial \varphi_s}{\partial \mathbf{R}} = \frac{\partial \varphi_c}{\partial \mathbf{R}} \text{ and } \frac{\partial \varphi_c}{\partial \mathbf{R}} = \frac{\partial \varphi_D}{\partial \mathbf{R}}$$
(4.39)

These boundary condition have been simplified as the dielectric permittivity is constant throughout the system.

The solution for the four coefficients has been found using a matrix method to solve the simultaneous equations, which is not reproduced here (see Appendix B). The resultant form for the coefficients is large and after some manipulation can be reduced to the set of equations given by (4.40)-(4.45).

$$A_{s}(k) = \frac{Q}{4\pi\varepsilon_{s_{i}}\varepsilon_{0}} 2k A_{m}(k) \exp(-Z_{I}K_{C})$$
(4.40)

$$A_{C}(k) = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{C}} (K_{s} - K_{C}) A_{m}(k) \exp(-Z_{I}K_{C})$$
(4.41)

$$B_{C}(k) = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{C}} (K_{D} - K_{C}) A_{n}(k) \exp(-Z_{I}K_{C})$$
(4.42)

4.3 Double Interface Potential Model

$$A_{D}(k) = \frac{Q}{4\pi\varepsilon_{SI}\varepsilon_{0}} 2k A_{n}(k) \exp(-Z_{I}K_{C}) \exp(-L_{C}(K_{C}+K_{D}))$$
(4.43)

The following two coefficients become common components which control the screened fields for the polarisation charges in the source, A_m , and in the drain, A_n .

$$A_{m} = \frac{K_{C} \left(\exp(2L_{C}K_{C}) + \exp(2Z_{I}K_{C}) \right) + K_{D} \left(\exp(2L_{C}K_{C}) - \exp(2Z_{I}K_{C}) \right)}{\left(K_{C}^{2} + K_{D}K_{S} \right) \left(\exp(2L_{C}K_{C}) - 1 \right) + K_{C} \left(K_{D} + K_{S} \right) \left(\exp(2L_{C}K_{C}) + 1 \right)}$$
(4.44)

$$A_{n} = \frac{K_{C} \left(\exp(2Z_{I}K_{C}) + 1 \right) + K_{S} \left(\exp(2Z_{I}K_{C}) - 1 \right)}{\left(K_{C}^{2} + K_{D}K_{S} \right) \left(\exp(2L_{C}K_{C}) - 1 \right) + K_{C} \left(K_{D} + K_{S} \right) \left(\exp(2L_{C}K_{C}) + 1 \right)}$$
(4.45)

After some re-arrangement the potential for the source, channel and drain regions can be simplified to use only the coefficients $A_m(k)$ and $A_n(k)$. These simplified forms are:

$$\varphi_{S} = \frac{Q}{4\pi\varepsilon_{Si}\varepsilon_{0}} (1 - \theta(Z)) \int_{0}^{\infty} dk J_{0}(kR) 2k A_{m}(k) \exp(-Z_{I}K_{C}) \exp(ZK_{S})$$
(4.46)

$$\varphi_{c} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \left(\theta(Z) - \theta(Z - L_{c})\right) \int_{0}^{\infty} dk J_{0}(kR) \frac{k}{K_{c}} \left[\exp\left(-|Z - Z_{I}|K_{c}\right) - (K_{s} - K_{c})A_{m}(k)\exp\left(-(Z + Z_{I})K_{c}\right) - (K_{D} - K_{c})A_{n}(k)\exp\left((Z - Z_{I})K_{c}\right)\right]$$

$$(4.47)$$

$$\varphi_{D} = \frac{Q}{4\pi\varepsilon_{s_{i}}\varepsilon_{0}} \theta(Z - L_{c}) \int_{0}^{\infty} dk J_{0}(kR) 2k A_{n}(k) \exp(-(Z_{I} - L_{c})K_{c}) \times \exp(-(Z - L_{c})K_{D})$$
(4.48)

$$\varphi = \varphi_{\rm s} + \varphi_{\rm c} + \varphi_{\rm D} \tag{4.49}$$

Examining the forms of the calculated potentials, there are some distinct similarities with the single interface model defined in section 4.2. The source potential term, φ_s , is almost identical apart from a change in the coefficient. The channel potential is also very similar but has an additional component included from the drain polarisation charges, the third term within the square brackets of equation (4.47). More detail on the calculation of these potentials is given in Appendix B. This includes an outline of the matrix method used to solve for the coefficients.

In the single interface model, the single induced polarisation charge could be represented by a simple coefficient which can be said to represent a ratio of the screening densities between the source and channel regions. In the double interface model above, the coefficients become considerably more complex although represent a similar ratio of screening between the source, channel and drain regions.

As mentioned earlier, the double interface model in certain conditions will induce multiple polarisaiton charges. This is a side-effect of having two reflecting interfaces a short distance apart, an analogy being that of having two mirrors facing each other reflecting the same image. Here this behaviour can be seen in the coefficients A_m and A_n which control the polarisation charges induced from the source and drain regions. Looking at the coefficients of equations (4.44) and (4.45), the extra complexity of the multiple polarisation charges can be seen by the introduction of the positive exponential components. The positive exponentials represent an infinite sum of polarisation charges which interact to increase the overall screening effect.

4.3.2 Contour Plots

Plotting the total potential, equation (4.49), for an impurity in three different locations in Figure 4.13 shows the effect of the double interface model. Polarisation charge effects are present at both ends of the channel and importantly, combine to further increase the screening effect.



Figure 4.13: Potential contour plots of a device with a channel length $L_c = \lambda_c = 4.133 nm$. Channel doping is $N_A = 10^{18} cm^{-3}$ and the source-drain doping is $N_D = 10^{20} cm^{-3}$. Plots (a)-(c) identical screening in the source, channel and drain. Plots (d)-(f) include the polarisation charge effects of the highly-doped source and drain regions.

The largest change between the single and double interface potential models comes in the form of the coefficients $A_m(k)$ and $A_n(k)$ given by equations (4.44)-(4.45). In this model the coefficients are large and complex as the polarisation charge effect develops beyond a reflection of charge from one surface to reflections between two surfaces. Polarisation charges or reflections between the heavily-doped source and drain regions is more complex than the single interface. Under circumstances that the channel length is sufficiently short and an atomistic impurity is not fully screened in the distance to the interfaces, the double interface model induces multiple polarisation charges. This situation will occur for channel lengths which are smaller than the channel screening length.

4.3.3 Long Channel Limit

As with the single interface model, it is important to check that the calculated potential behaves in the expected manner at appropriate limits. Due to the similarities between this model and the single interface model it is not felt necessary to repeat here the screened Coulomb and the matched screening limit. Instead the potentials for this model will be checked to ensure that in the appropriate limit they return to the single interface case. This limit has been entitled the long channel limit and can be found by allowing the channel length to become very large, that is the limit of $L_c \to \infty$. For this limit it is best to first take the limit on the coefficients $A_m(k)$ and $A_n(k)$ given by equations (4.44)-(4.45):

$$\lim_{L_{c}\to\infty} A_{m} = \frac{\exp(2L_{c}K_{c})(K_{c}+K_{D}) + \exp(2Z_{I}K_{c})(K_{c}+K_{D})}{\exp(2L_{c}K_{c})(K_{c}^{2}+K_{D}K_{S}+K_{c}(K_{D}+K_{S}))}$$

$$= \frac{1}{K_{c}+K_{s}}$$
(4.50)

$$\lim_{L_{C} \to \infty} A_{n} = \frac{\exp(2Z_{I}K_{C})(K_{C} + K_{S}) + K_{C} - K_{S}}{\exp(2L_{C}K_{C})((K_{C} + K_{S})(K_{C} + K_{D}))}$$
(4.51)
= 0

The coefficient A_n which controls the drain polarisation charges correctly tends to zero as the channel length tends to infinity. The source controlled polarisation charge terms, given by A_m , simplifies vastly as the coefficient is reduced to modelling a single polarisation charge term within the source region. Substituting these coefficients into equations (4.46)-(4.48) gives the single interface model, which is not repeated here, as expected if the drain region is a large distance from the source and the atomistic impurity.

4.3.4 Comparison with Non-Linear Poisson Solution

Comparison of the single interface solution with the NLP solver in section 4.2.4 has shown excellent agreement. To ensure that the double interface model also agrees with a full solution of Poisson's equation, a similar test will be completed here. Again using a solution of the uniform device with the NLP solver to find the solution including the depletion region.

A simple example device will be used to test the double interface remotely screened impurity solution and is depicted in Figure 4.14. The device will have a channel length which is approximately equal to a single screening length at $N_A = 10^{18} cm^{-3}$ of $L_C \simeq \lambda_C \simeq 4.25 nm$. This device will be referred to as the lambda channel device, for want of a better name.



Figure 4.14: Lambda channel device for the comparison of the double interface potential with a non-linear Poisson solver.

This device is important to test because under conditions when the channel length is around the screening length, multiple polarisation charges are likely to be present in the system as discussed at the end of section 4.3.2. Figure 4.15 gives the plots of the lambda channel device with a single atomistic impurity in three different locations. Plots (a) and (b) give the dopant in positions which correspond roughly to $Z_I = 0.1\lambda_c$, $0.9\lambda_c$ and plot (c) in position $Z_I = 0.5\lambda_c$.



Figure 4.15: Plots of the non-linear Poisson comparison with the analytical model for remotely-screened impurities within the lambda channel device.

These plots show the excellent agreement between the analytical and numerical solutions. Regardless of the impurity position the match is almost indistinguishable neglecting the discretization error of the potential peak in the NLP solution.

4.3.5 Strong-Screening Limit

The strong screening limit of the double interface model, that is $k_s, k_D \gg k_c$, will also simplify the potential terms much like in the single interface case. Here the potential equations and coefficients from section 4.3.1 (equations (4.44)-(4.48)) are used for the double interface model. For simplicity when applying the limit, it will taken that for the limit of $k_s, k_D \gg k_c$, the K_x terms will become:

$$\lim_{\substack{k_s, k_D \gg k_c}} \begin{cases} \mathbf{K}_s = k_s \\ \mathbf{K}_D = k_D \end{cases}$$
(4.52)

Taking the limit of the coefficients first:

$$\lim_{k_{s},k_{D}\ggk_{c}}A_{m} = \frac{k_{D}\left(\exp(2L_{C}K_{C}) - \exp(2Z_{I}K_{C})\right)}{k_{D}k_{s}\left(\exp(2L_{C}K_{C}) - 1\right) + k_{D}K_{C}\left(1 + \frac{k_{s}}{k_{D}}\right)\left(\exp(2L_{C}K_{C}) + 1\right)}$$

$$A_{m}^{\lim} = \frac{\exp(2L_{C}K_{C}) - \exp(2Z_{I}K_{C})}{k_{s}\left(\exp(2L_{C}K_{C}) - 1\right)}$$
(4.53)

$$\lim_{k_{S},k_{D}\ggk_{C}}A_{n} = \frac{k_{S}\left(\exp(2Z_{I}K_{C})-1\right)}{k_{S}k_{D}\left(\exp(2L_{C}K_{C})-1\right)+k_{S}K_{C}\left(1+\frac{k_{D}}{k_{S}}\right)\left(\exp(2L_{C}K_{C})+1\right)}$$

$$A_{n}^{\lim} = \frac{\exp(2Z_{I}K_{C})-1}{k_{D}\left(\exp(2L_{C}K_{C})-1\right)}$$
(4.54)

Substituting these coefficients into the limit of equations (4.46) and (4.48) gives the limiting form of the double interface source and drain potentials as

$$\lim_{k_s,k_b \gg k_c} \varphi_s \simeq \frac{\exp(Zk_s)}{k_s} = 0 \quad \text{for } Z < 0$$
(4.55)

$$\lim_{k_{S},k_{D}\gg k_{C}}\varphi_{D} \simeq \frac{\exp\left(-\left(Z-L_{C}\right)k_{D}\right)}{k_{D}} = 0 \quad \text{for } Z > L_{C}$$

$$(4.56)$$

Finally, substitution of the strongly screened coefficients into the channel potential term yields the following.

$$\lim_{k_{s},k_{D}\ggk_{C}}\varphi_{C} = \frac{Q}{4\pi\varepsilon_{s_{I}}\varepsilon_{0}} \left(\theta(Z) - \theta(Z - L_{C})\right) \int_{0}^{\infty} dk J_{0}\left(kR\right) \frac{k}{K_{C}} \left[\exp\left(-|Z - Z_{I}|K_{C}\right) - k_{s}A_{m}^{\lim}\left(k\right)\exp\left(-(Z + Z_{I})K_{C}\right) - k_{D}A_{n}^{\lim}\left(k\right)\exp\left((Z - Z_{I})K_{C}\right)\right]$$

$$(4.57)$$

This is a substantial reduction in complexity of the model, yet still retains the important polarisation charge terms from the reflecting source and drain regions, the 2^{nd} and 3^{rd} terms in the square brackets of equation (4.57) respectively.

Importantly the coefficients A_m^{\lim} , A_n^{\lim} retain the positive exponential components which can be considered to represent the *multiple-image* effect discussed earlier in this chapter. The *multiple-image* effect is the repeated reflection of the impurity point charge between the source and drain regions when the channel length is sufficiently small.

4.3.5.1 Verification of Limit

Repeating a procedure similar to that of the single interface case for the strongly-screened model, the strongly screened double interface model will now be compared to the complete potential. Again, the purpose of this limit is to provide an upper-bound on the remote screening effect of channel ionized impurities.

The comparison will initially be based on two test devices which have different channel lengths, a 25nm channel length device and a 15nm channel device. Both of these devices have a channel doping concentration of $N_I = 10^{15} cm^{-3}$ which corresponds to screening length of $\lambda_c = 129.29nm$ using equation (3.36), the Debye-Hückel screening model. The source and drain regions of these devices which is referenced only within the complete model is doped to $N_I = 10^{20} cm^{-3}$.





Figure 4.16: Comparison of the complete double interface potential with the strongly screened model for two different channel length devices. Atomistic impurity located exactly mid-channel of each device.

Figure 4.16 shows the comparison between the potentials for (a) the 25nm device and (b) the 15nm channel device. The potential of the strongly screened model is a close match to the complete model surrounding the impurity as expected. Again there is an increased screening of the potential close to the interfaces which is consistent with the strongly screened single interface model.

With the strongly screened interface model discussed in section 4.2.5.2, the region over which the limit caused a notable error in the potential was within roughly 1nm of the interface. It was shown that modelling an impurity within this region of the interface caused a significant error in the potential. Essentially the limit over estimated the polarisation charge effect and over screened the impurity potential. In the double interface model this error in the potential seems evident over a larger region from the interface. Looking at Figure 4.16 the difference in the potential becomes appreciable at around ~2nm from each interface. Plotting the comparison in more detail in Figure 4.17 allows a better analysis of the error.



Figure 4.17: Detailed comparison of the strongly screened and complete double interface potential models.

Figure 4.17 shows that at ~2nm from the interface the strongly screened model is approximately 80% of the complete model. This corresponds to around a 1-1.5mV potential difference for the 25nm and 15nm channel length devices.

To further test the validity of the strongly screened model another comparison has been completed with a device whose channel length is equal to the channel screening length, for want of a better name, the lambda channel device. This device has a channel doped to $N_I = 10^{18} cm^{-3}$ which corresponds to a screening length of $\lambda_c = 4.133 nm$. The source and drain regions are doped to $N_I = 10^{20} cm^{-3}$, again this is only referenced with the complete model.

In Figure 4.18 and Figure 4.19 are the potential comparisons between the models in the lambda channel device. Initially this comparison looks at an atomistic impurity at three different locations in the channel but the detailed comparison will look at only two positions due to the symmetry between the left and right positions.



Figure 4.18: Comparison between the strongly screened and complete double interface model using the lambda channel device. Atomistic impurity located at (a) $Z_1 = 1nm$, (b) $Z_1 = 0.5\lambda_c$ and (c) $Z_1 = \lambda_c - 1nm$.

Plot (c) of Figure 4.18 shows a single atomistic impurity located exactly mid channel, just over 2nm from each interface. There is an appreciable drop in the potential within 1nm of the interface which is given in more detail in plot (b) of Figure 4.19. At 1nm, the strongly screened model is

around 80% of the complete model with a corresponding drop of roughly 10mV. For an impurity here within 2nm of the interface, 50% of the potential between the interface is above 80%.



Figure 4.19: Plots of the ratio and potential difference between the strongly screened model and the complete model for the lambda channel device. Plot (a) shows the impurity at $Z_1 = 1nm$ and (b) the impurity positioned at $Z_1 = 0.5\lambda_c = 0.5L_c$.

Looking at the case of the impurity located close to the source interface given by plot (a) of Figure 4.18 and Figure 4.19. An impurity located at 1nm from the interface the maximum potential difference is increased by a factor of 2-3 times the mid channel impurity. At this distance the validity of the strong screening limit is questionable. This behaviour is not completely dissimilar to that of the single interface model with an impurity at this distance from the interface given by Figure 4.9.

It is clear that as the impurity is moved closer to the interface the error in the strongly screened model will increase and is very much appreciable within 2nm of an interface. At impurity positions greater than this distance from either the source or drain interface, the limited model is shown above to be close to the complete model with small shifts in the potential of several mV's near the interfaces.

Unfortunately due to time constraints in the PhD project, the complete double interface momentum relaxation rate and therefore the mobility has not been calculated for this model. As discussed in the above examination of the strong screening effect on the potential, it can be said that the limit will induce an increasing error as impurities are located close to either the source or drain interface (or both). This said, the objective of using this limit is to obtain a worst-case condition for the remote screening of channel impurities induced from polarisation charge effects in the source and drain regions. This is certainly achieved by imposing a limiting condition on the screening density in these regions.

4.4 Conclusion

In sections 4.2 and 4.3 of this chapter a potential solution for a single atomistic impurity located close to one or two highly doped regions has been obtained. This has been completed by solving Poisson's equation using the LTF approach to obtain an exact analytical solution. This solution naturally includes polarisation charge effects induced by the boundaries which are the focus of this work. Polarisation charge effects are shown to increase the screening of an atomistic impurity located close to abrupt interface with a highly doped region. Hence these potentials represent the remotely screened impurity potential for the cases of an impurity located close to the source and/or drain regions.

The mathematical limits of the potential solutions have been checked to ensure the correct behaviour. It has been shown that polarisation charge effects disappear when the impurity is a large distance from the interfaces. Under this condition the model returns to the screened Coulomb potential which is the classic potential for an impurity and is an important limit for this model.

To further prove the validity of the approach and solution presented, in section 4.2.4 and 4.3.4 a comparison between a fully self-consistent, non-linear Poisson solver and the calculated analytical approach has been completed. This comparison shows a close agreement between the approaches and highlights the accuracy of the remotely-screened impurity potentials.

Finally, a simplified model has been obtained by introduction of the strong screening limit in sections 4.2.5 and 4.3.5, which assumes that the source and drain regions are degenerately doped and become metallic like. The limit has the resultant effect of screening all the induced impurity potential in the source/drain regions at the interface, thereby reducing the potentials in these regions to zero.

In the single interface model, the strongly screened potential is shown to be almost exact for impurities which are located greater than 1nm from the source interface. For impurities located closer than 1nm from the interface the strongly screened potential greatly over-estimates the screening effect. Similar behaviour is seen in the double interface strongly screened potential for impurities located close to either interface.

The purpose of using the strongly screened models is to represent a worst-case scenario of this model in order to obtain an estimate of the effect on device performance. The strongly screened potentials can easily be used in scattering rate calculations whilst providing an upper bound on the remote screening features of polarisation charge effects.

Chapter 5 Scattering Rate Calculation

5.1 Introduction

In typical Monte Carlo (MC) simulations Ionized Impurity (II) scattering is based on a simple model of a single II located in a semiconductor material. Complex boundary effects such as polarisation charge effects are not included in this simple physical picture. Existing II scattering models based on this simple physical picture have been given a thorough review in Chapter 2. It should also be noted here that the *ab initio* atomistic approach to II scattering developed in this research group [62, 63] does include the complex boundary effects through the Poisson equation solution. Although, the *ab initio* approach is a classical approach to the problem in that the II scattering is achieved through the classical particle transport in MC, here the intention is to obtain a quantum description of the polarisation charge effect.

In this chapter the aim is to develop a scattering model for MC simulation which extends the existing II scattering model. This new scattering rate will allow the II scattering model to include the complex effects of IIs that are located close to the source and drain regions. The impurity potential equations developed in the previous chapter will be used to develop this new model. In this new model the effect of highly-doped regions located close to an ionized dopant is to alter the screening of the impurity potential. Hence, the new model has been entitled remotely screened impurity scattering to reflect the nature of the induced screening from polarisation charge induced from the source and/or drain regions.

Initially, the scattering matrix element and then the scattering rate will be calculated for both the single interface and double interface models in sections 5.2 and 5.3 respectively. The method used to obtain the scattering rate will be based on Fermi's Golden Rule approach. Calculation of the momentum relaxation rates and differential cross-section will also be included, which are very useful for analysis of the scattering model.

In section 5.4 the application of this new scattering model in sub-threshold device conditions is discussed where screening is very low. Low-screening densities cause large problems with II scattering in MC simulations as will be highlighted and resolved for this scattering model. This involves the development of a new II scattering approach which is presented in detail.

5.2 Single Interface Scattering Model

Following this, a brief discussion of the numerical implementation into the MC simulator will be covered in section 5.5. This section will cover the techniques used within the MC simulator to calculate the scattering rate and complete the scattering process.

5.2 Single Interface Scattering Model

In the previous chapter the scattering potential for an atomistic impurity located close to a reflecting interface was calculated. This scattering potential can be used to develop a scattering rate for MC simulation. For such a scattering rate, the matrix element must first be evaluated from the scattering potential.

In the case for the remotely screened impurity model, the strong screening limit potentials shall be used as they simplify the model whilst retaining the important polarisation charge effect of the source region. As discussed in the previous chapter, use of this limit is a worst-case condition which leads to an overestimation of the remote screening effect induced from the source region for impurities located close to the source interface. The aim of this work is to look at what effect remote screening has on device performance, therefore the use of strongly screened model is considered suitable as it will provide an upper limit on remote screening.

The scattering potential for the remotely-screened II model is not spherically symmetric like the standard II scattering models. With the remotely-screened model the scattering potential is anisotropic and therefore varies depending on the angle with which the carrier sees the impurity. The typical textbook approach for the scattering in Monte Carlo makes use of isotropic scattering potentials. Anisotropy in scattering is typically found through the band structure via anistropic effective masses which can also be modelled using an isotropic effective mass with an anisotropic scattering potential [101-104]. There are several approaches for modelling anisotropic scattering potentials such as a spherical harmonics expansion of the Schrödinger equation as discussed by Boardman [105], or through solutions to the linear Boltzmann equation [106-109]. Due to the complexity of these approaches and the context of this work which is to analyse what effect remote screening may have, a simpler approach is employed here. This simpler approach will allow for an initial examination of the strength of remote screening of ionized impurity scattering in a Monte Carlo simulation.

For the purposes of this work a simplifying approximation is made such that the anisotropy of the scattering potential is removed by allowing the incoming carrier angle to be aligned with the principle scattering axis. In other words, we make an assumption on the alignment of the scattering potential with the scattering carrier which removes the anisotropy. This alignment is discussed and

analysed in detail later in this chapter and is shown to lead to a negligible error in all cases. Further discussion of this simplification is left to section 5.2.3.

Initially the scattering matrix element suitable for use with Fermi's Golden Rule will be calculated from the scattering potential defined in the previous chapter, section 4.2. Following this, in section 5.2.2 the scattering rate for use in the MC simulation technique will be developed. Here other important scattering model equations are calculated such as the differential cross-section and momentum relaxation rate. These have a specific importance in the analysis of the scattering model, allowing the magnitude of the scattering probability and the effect on carrier transport to be examined. In section 5.2.3 the incoming angle simplification will be discussed in detail and the effect the simplification has on the model will be presented. Finally in section 5.2.4 the scattering model developed in this section will be analysed.

5.2.1 Scattering Matrix Element

The matrix element for a scattering rate is defined as [30]

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{-\infty}^{\infty} d^{3}\mathbf{r} \exp(-i\mathbf{k}' \cdot \mathbf{r}) U_{s}(\mathbf{r}) \exp(i\mathbf{k} \cdot \mathbf{r})$$
(5.1)

where electron plane wave functions have been assumed. The matrix element has been normalised over the 3D unit volume, Ω , and $U_s(\mathbf{r})$ is the scattering energy. Simplifying this using $\mathbf{q}^2 = |\mathbf{k}' - \mathbf{k}|^2$ and separating into cylindrical co-ordinates, the matrix element definition may be written as a Fourier transform over the variables \mathbf{q}_{\perp} and q_z .

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{-\infty}^{\infty} dZ \int_{0}^{2\pi} d\phi \int_{0}^{\infty} d\mathbf{R} \ U_{S}(\mathbf{R}, Z) \exp(-i\mathbf{q}_{\perp} \cdot \mathbf{R}) \exp(-iq_{Z}Z)$$
(5.2)

Throughout this section the scattering momentum transfer variables will be expressed in cylindrical co-ordinates using the perpendicular and Z-directed momentum transfer wave vectors, \mathbf{q}_{\perp} and q_{Z} . A more detailed discussion on the exact definition of the scattering momentum transfer wavevectors for this scattering model is left to section 5.2.3. For now it is sufficient to know that the definition of the scattering momentum transfer in cylindrical co-ordinates is given by equations (5.3)-(5.4).

$$\mathbf{q}_{\perp}^{2} = \left| \mathbf{k}_{\perp}^{\prime} - \mathbf{k}_{\perp} \right|^{2} \tag{5.3}$$

$$q_{Z}^{2} = \left|k_{Z}' - k_{Z}\right|^{2} \tag{5.4}$$

For the single interface model, the scattering potential to be used is the strong screening limit of the model. This potential which was calculated in section 4.2 of the previous chapter can be written as

$$V(\mathbf{R}, Z) = \theta(Z) \frac{Q}{4\pi\varepsilon_{Si}\varepsilon_0} \left[\frac{\exp\left(-k_C \sqrt{\mathbf{R}^2 + (Z - Z_I)^2}\right)}{\sqrt{\mathbf{R}^2 + (Z - Z_I)^2}} - \frac{\exp\left(-k_C \sqrt{\mathbf{R}^2 + (Z + Z_I)^2}\right)}{\sqrt{\mathbf{R}^2 + (Z + Z_I)^2}} \right]$$
(5.5)

Substituting this scattering potential into the scattering matrix definition equation (5.2) using the transform $U_s(\mathbf{R},Z) = eV(\mathbf{R},Z)$ gives equation (5.6). As the strongly screened potential is valid only for the Z > 0 region, the integral over the Z space in the Fourier transform has been corrected accordingly.

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{0}^{\infty} dZ \int_{0}^{2\pi} d\phi \int_{0}^{\infty} d\mathbf{R} \ eV(\mathbf{R}, Z) \exp(-i\mathbf{q}_{\perp} \cdot \mathbf{R}) \exp(-iq_{Z}Z)$$
(5.6)

After some lengthy integration, the Fourier transform above can be completed to give

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{eQ}{4\pi\varepsilon_{SI}\varepsilon_0} \frac{1}{\Omega} \frac{4\pi}{\mathbf{q}_{\perp}^2 + q_Z^2 + k_C^2} \bigg[\exp(-iq_Z Z_I) - \exp(-Z_I \sqrt{\mathbf{q}_{\perp}^2 + k_C^2}) \bigg]$$
(5.7)

Taking the magnitude-squared of this matrix element, also known as the Born approximation, provides us with a form suitable for use in Fermi's Golden Rule. After some algebraic manipulation the scattering matrix element becomes

$$\left|\mathbf{H}_{\mathbf{k}'\mathbf{k}}\right|^{2} = \left(\frac{eQ}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}}\right)^{2} \frac{1}{\Omega^{2}} \left(\frac{4\pi}{\mathbf{q}_{\perp}^{2} + q_{Z}^{2} + k_{C}^{2}}\right)^{2} \times \left[1 + \exp\left(-2Z_{I}\sqrt{\mathbf{q}_{\perp}^{2} + k_{C}^{2}}\right) - 2\cos\left(q_{Z}Z_{I}\right)\exp\left(-Z_{I}\sqrt{\mathbf{q}_{\perp}^{2} + k_{C}^{2}}\right)\right]$$
(5.8)

Equation (5.8) is the scattering matrix element for the single interface, remotely screened impurity scattering model. As is expected the matrix element has some similarities with that of the Brooks-Herring (BH) approach (see section 2.3) albeit given here in cylindrical co-ordinates. The terms

within the square brackets represents the remote screening coefficient and defines the extent of the polarisation charge effects in screening the impurity. These terms are dominated by the $\exp(-Z_I..)$ components which reduce the interaction as the impurity position Z_I increases. In other words the remote screening effect drops off roughly exponentially as the impurity is located farther from the source region.

As discussed in the previous chapter it is important for this scattering rate to return to the BH model when the atomistic impurity is located a large distance from the interface. Looking at the second and third terms in the square brackets of the matrix element, the exponentials of negative power, it is clear that in the limit of $Z_I \rightarrow \infty$ these terms will tend to zero and will yield the BH model.

5.2.2 Scattering Rate

Having found the scattering matrix element in section 5.2.1 for the remotely screened impurity model, it is now possible to define the total scattering rate for use in MC simulation. This calculation will follow the method of Fermi's Golden Rule which defines the scattering transition probability for a carrier wave vector \mathbf{k} to a state \mathbf{k}' . In addition to the total scattering rate, the differential scattering cross-section and the momentum relaxation rate will be calculated for the new impurity scattering model.

Equation (5.9) is the Fermi Golden Rule for an elastic scattering event [110], which is a scattering event where the incoming and outgoing energies are equal. Here the overlap integral for electrons is assumed to equal one and the band structure is modelled by an ellipsoidal and non-parabolic model. The effective mass, which is a 2nd rank tensor with a single diagonal component, is represented by the isotropic density of states effective mass, $m_d^* = \sqrt[3]{m_l m_t^2}$, by making use of the Herring-Vogt transformation as discussed in section 3.2.3 [71]. The Herring-Vogt transformation allows the ellipsoidal bandstructure to be represented as a spherical bandstructure.

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} |\mathbf{H}_{\mathbf{k}'\mathbf{k}}|^2 \,\delta\Big(E\Big(\mathbf{k}^{*'}\Big) - E\big(\mathbf{k}^{*}\Big)\Big)$$
(5.9)

Here \mathbf{k}^* is the carrier wave vector after the Herring-Vogt transformation into starred space (see section 3.2.3). This equation calculates the probability of scattering from a state \mathbf{k} to a state \mathbf{k}' but the calculation must be completed using the Herring-Vogt transformed vectors, hence the change of the vector in this expression. Inserting the scattering matrix element of equation (5.8) into equation (5.9) for the probability of scattering gives

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} \left(\frac{eQ}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}}\right)^{2} \frac{1}{\Omega^{2}} \left(\frac{4\pi}{\mathbf{q}_{\perp}^{2} + q_{z}^{2} + k_{c}^{2}}\right)^{2} \delta\left(E\left(\mathbf{k}^{*'}\right) - E\left(\mathbf{k}^{*}\right)\right)$$

$$\times \left[1 + \exp\left(-2Z_{I}\sqrt{\mathbf{q}_{\perp}^{2} + k_{c}^{2}}\right) - 2\cos\left(q_{Z}Z_{I}\right)\exp\left(-Z_{I}\sqrt{\mathbf{q}_{\perp}^{2} + k_{c}^{2}}\right)\right]$$
(5.10)

There are various quantities which can be obtained from the scattering probability, $P(\mathbf{k}, \mathbf{k}')$, which allow various different properties of the scattering model to be analysed. The differential scattering cross-section is found using equation (5.11) which integrates the scattering probability over all \mathbf{k} space. Here v is the velocity of a state \mathbf{k} given by equation (5.12) where α is the nonparabolicity parameter.

$$\sigma = \frac{\Omega^2}{\nu(\mathbf{k})} \int_0^\infty d\mathbf{k}^{*'} \mathbf{k}^{*'2} P(\mathbf{k}, \mathbf{k}')$$
(5.11)

$$\nu(\mathbf{k}) = \frac{1}{\hbar} \frac{\partial E}{\partial \mathbf{k}} = \frac{\hbar k^*}{m_d \left(1 + 2\alpha E\right)}$$
(5.12)

The differential scattering cross-section is a function of angle and allows for analysis of the strength of a scattering centre. Using the scattering probability in equation (5.11) and completing the integral yields

$$\sigma = \left(\frac{Ze^2}{4\pi\varepsilon_{s_l}\varepsilon_0}\right)^2 \frac{2^2 m_d^2}{\hbar^4} (1+2\alpha E)^2 \frac{1}{\left(\mathbf{q}^2+k_c^2\right)^2} \times \left[1+\exp\left(-2Z_I\sqrt{\mathbf{q}_\perp^2+k_c^2}\right) - 2\cos\left(q_Z Z_I\right)\exp\left(-Z_I\sqrt{\mathbf{q}_\perp^2+k_c^2}\right)\right]$$
(5.13)

Here the impurity charge, Q, has been replaced by the number of free unit charges of the II, Z, multiplied by the electron charge, e.

The total scattering rate and the momentum relaxation rate are calculated in a similar manner to each other and are specified in equations (5.14) and (5.15) respectively. The momentum relaxation rate includes a weighting term by the change in the momentum of a scattering event, hence the extra term within the integral on the RHS of equation (5.15).

$$\Gamma(\mathbf{k}) = \sum_{\mathbf{k}^{*'}} P(\mathbf{k}, \mathbf{k}') = N_k \int_{\mathbf{k}^{*'}} P(\mathbf{k}, \mathbf{k}') d\mathbf{k}^{*'}$$
(5.14)

$$\frac{1}{\tau_m(\mathbf{k})} = \sum_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') \left(1 - \frac{\mathbf{k}^{*'}}{\mathbf{k}^*} \cos \theta_s \right) = N_k \int_{\mathbf{k}^{*'}} P(\mathbf{k}, \mathbf{k}') \left(1 - \frac{\mathbf{k}^{*'}}{\mathbf{k}^*} \cos \theta_s \right) d\mathbf{k}^{*'}$$
(5.15)

$$N_k = \frac{\Omega}{\left(2\pi\right)^3} \tag{5.16}$$

where θ_s is the angle between the incident and scattered wave-vector. The exact definition of the scattering angle is left to section 5.2.3.

The total scattering rate can be found by substituting (5.10) into equation (5.14) and after expanding the integral using spherical co-ordinates, the scattering rate is given as

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{s_i}\varepsilon_0}\right)^2 \frac{1}{\Omega} \frac{2^2}{\hbar} \int_0^{2\pi} d\varphi_k \int_0^{\pi} d\theta_k \sin \theta_k \int_0^{\infty} dk^{*'} k^{*'2} \frac{1}{\left(\mathbf{q}^2 + k_c^2\right)^2} \\ \times \left[1 + \exp\left(-2Z_I \sqrt{\mathbf{q}_{\perp}^2 + k_c^2}\right) - 2\cos\left(q_Z Z_I\right) \exp\left(-Z_I \sqrt{\mathbf{q}_{\perp}^2 + k_c^2}\right)\right]$$
(5.17)
$$\times \delta\left(E\left(\mathbf{k}^{*'}\right) - E\left(\mathbf{k}^{*}\right)\right)$$

The integral over the range of final electron wave vectors, k', can be completed easily due to the Dirac delta function which ensures energy conservation. This leaves the integral over the final angles $\theta_{k'}$ and $\varphi_{k''}$. These integrals have not been completed here and will be discussed in the next section. After some algebraic manipulation and multiplication by the II density of the unit volume, $N_I \Omega$, the total scattering rate is found as

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{s_l}\varepsilon_0}\right)^2 N_I \frac{2^2 m_d k^*}{\hbar^3} (1+2\alpha E) \int_0^{2\pi} d\varphi_{k'} \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}}{\left(\mathbf{q}^2 + k_c^2\right)^2} \times \left[1 + \exp\left(-2Z_I \sqrt{\mathbf{q}_{\perp}^2 + k_c^2}\right) - 2\cos\left(q_Z Z_I\right) \exp\left(-Z_I \sqrt{\mathbf{q}_{\perp}^2 + k_c^2}\right)\right]$$
(5.18)

Here and in the following expressions the magnitude of the incoming vector after the Herring-Vogt transformation is written as k^* for simplicity. The momentum relaxation rate for the scattering process can be defined by inserting the momentum relaxation weighting term into the $\theta_{k'}$ integral of the total scattering rate and using the elastic scattering definition that $E(\mathbf{k'}) = E(\mathbf{k})$.

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^2 m_d k^*}{\hbar^3} (1+2\alpha E) \int_0^{2\pi} d\varphi_k \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}(1-\cos\theta_s)}{(\mathbf{q}^2+k_c^2)^2} \times \left[1+\exp\left(-2Z_I \sqrt{\mathbf{q}_{\perp}^2+k_c^2}\right) - 2\cos\left(q_Z Z_I\right) \exp\left(-Z_I \sqrt{\mathbf{q}_{\perp}^2+k_c^2}\right)\right]$$
(5.19)

5.2.3 Scattering Reference Frame

The II scattering models of BH and Conwell-Weisskopf (CW) have scattering potentials which are spherically symmetric. In a spherically symmetric scattering potential the angle of the incoming carrier doesn't affect the potential the carrier interacts with. This allows the scattering and device reference frames to be aligned reducing the complexity of the model. The scattering reference frame being that used to describe the scattering event and the device reference frame describing the physical device for the direction of carrier momentum.

In the remotely screened model developed here the scattering potential given by equation (5.5) is not spherically symmetric. An example of the potential contours is given in Figure 5.1 below. There is rotational symmetry around the Z-axis due to the assumption that the II is located at $\mathbf{R}_{I} = 0$ (see Figure 5.4 and Figure 5.5). To properly describe the scattering event for this model the scattering reference frame must be fixed and be consistent with the potential model.



Figure 5.1: Fixed scattering reference frame with example case of remotely screened impurity potential contours and incoming/outgoing carrier wave-vectors.

The fixed scattering reference frame used here is identical to the reference frame defined within the calculation of the potential (see Figure 4.1) and is closely related to the device reference frame which is aligned on the same axes within a cartesian coordinate system. An example of the fixed scattering reference frame is given by Figure 5.1 with the remotely screened II given by the red

5.2 Single Interface Scattering Model

circle and the incoming carrier by the blue circle. Figure 5.1 also shows example potential contours from a II located close to the source interface where the potential contours become slightly egg-shaped with a flat spot close to the source region, demonstrating the non-spherical nature.

Also demonstrated in this figure is an example scattering carrier with the incoming angle θ_k and outgoing angle $\theta_{k'}$ (the angle θ_k assumes a rotation of π around the Z axis). It is clear that the incoming angle of the carrier will affect the scattering potential that it sees and will alter the scattering rate.

For the Monte Carlo simulator used in this work, which is a 3D bulk Monte Carlo simulator it is ideal to express the wave-vectors in the spherical co-ordinate system. This ensures compatibility with the existing code and scattering processes.



Figure 5.2: Definition of the fixed scattering reference frame using a spherical co-ordinate system.

Figure 5.2 gives the fixed reference frame for this scattering model in more detail using the spherical co-ordinate system. Here the X-Y plane denoted in the figure represents the R-plane of the cylindrical co-ordinate system and of the R plane in Figure 5.1, the angle φ_k provides the angle of rotation around the Z-axis and θ_k gives the angle of the vector from the Z-axis. This is the

generic spherical co-ordinate reference system rotated such that it matches the axis definition of the scattering potential given by Figure 4.1.

Using the fixed reference frame, given in detail by Figure 5.2, the momentum transfer wavevectors of equations (5.3)-(5.4) can now be defined. Expanding the components in terms of Cartesian co-ordinates initially gives

$$\mathbf{q}_{\perp}^{2} = \left|\mathbf{k}_{\perp}^{\prime} - \mathbf{k}_{\perp}\right|^{2} = \left|k_{\perp}^{\prime} - k_{x}\right|^{2} + \left|k_{\perp}^{\prime} - k_{y}\right|^{2}$$
$$= \left(k_{\perp}^{\prime 2} + k_{x}^{2} - 2k_{\perp}^{\prime} \cdot k_{x}\right) + \left(k_{\perp}^{\prime 2} + k_{y}^{2} - 2k_{\perp}^{\prime} \cdot k_{y}\right)$$
(5.20)

$$q_{Z}^{2} = \left| k'_{Z} - k_{Z} \right|^{2} = k'_{Z}^{2} + k_{Z}^{2} - 2k'_{Z} \cdot k_{Z}$$
(5.21)

Using the following textbook transforms [96] and allowing the magnitude of the vector \mathbf{k} to be written simply as k

$$k_x = k \sin \theta_k \cos \varphi_k \tag{5.22}$$

$$k_{v} = k \sin \theta_{k} \sin \varphi_{k} \tag{5.23}$$

$$k_z = k \cos \theta_k \tag{5.24}$$

the scattering momentum transfer wave-vectors can be expressed in spherical terms as

$$\mathbf{q}_{\perp}^{2} = k^{2} \sin^{2} \theta_{k} + k^{2} \sin^{2} \theta_{k'} - 2kk' \sin \theta_{k} \sin \theta_{k'} \cos(\varphi_{k} - \varphi_{k'}) = k^{2} \left(\sin^{2} \theta_{k} + \sin^{2} \theta_{k'} - 2\sin \theta_{k} \sin \theta_{k'} \cos(\varphi_{k} - \varphi_{k'}) \right)$$
(5.25)

$$q_Z^2 = k^2 \cos^2 \theta_k + k^{\prime 2} \cos^2 \theta_{k'} - 2kk' \cos \theta_k \cos \theta_{k'}$$

= $k^2 \left(\cos \theta_k - \cos \theta_{k'}\right)^2$ (5.26)

These forms assume an elastic scattering process such that the incoming energy is conserved and so $E(\mathbf{k}) = E(\mathbf{k}')$ which is applicable to the II scattering process. Using equation (5.25) and (5.26), the full momentum transfer wave-vector can be written

$$\mathbf{q}^{2} = \mathbf{q}_{\perp}^{2} + q_{Z}^{2}$$

= $2k^{2} \left(1 - \cos \theta_{k} \cos \theta_{k'} - \sin \theta_{k} \sin \theta_{k'} \cos \left(\varphi_{k} - \varphi_{k'} \right) \right)$ (5.27)

The effect that the incoming carrier has on the scattering model can now be analysed using the momentum transfer relations defined above. Substituting equations (5.25)-(5.27) into the total scattering rate of equation (5.18) gives the model including the incoming wave-vector dependence.

Care must be taken with the momentum relaxation rate as the weighting term depends on the scattered angle between the carriers. As the angle between the incident and scattered wave-vector is defined here within a fixed reference frame system, the momentum relaxation weighting term must be given as

$$1 - \cos\theta_s \equiv 1 - \cos\theta_k \cos\theta_{k'} - \sin\theta_k \sin\theta_{k'} \cos(\varphi_k - \varphi_{k'})$$
(5.28)

Substitution of the momentum transfer relations and the weighting term of equations (5.25)-(5.28) into the momentum relaxation rate, equation (5.19), gives the incoming angle dependent momentum relaxation rate.

To test these relations are correct it is ideal to initially check that in the limit of $Z_I \rightarrow \infty$ the model returns to a spherically symmetric case. That is, the BH limit of the model which has no remote screening interaction. In Figure 5.3 the scattering rate is plotted over all incoming angles for such a limit.



Figure 5.3: Plot of the remotely screened scattering rate at $Z_1k_c \gg 4$ over all possible incoming carrier angles.

For this figure the impurity density matches the screening density and is given as $n = N_I = 10^{17} cm^{-3}$. The incoming carrier energy is calculated as the average energy at a lattice temperature of T = 300K using equation (5.29), yielding $\langle E \rangle \simeq 40 meV$.

$$\left\langle E \right\rangle = \frac{3}{2} k_B T \frac{\mathfrak{F}_{\frac{1}{2}}(\eta) + \frac{25}{4} \alpha k_B T \mathfrak{F}_{\frac{1}{2}}(\eta)}{\mathfrak{F}_{\frac{1}{2}}(\eta) + \frac{15}{4} \alpha k_B T \mathfrak{F}_{\frac{1}{2}}(\eta)}$$
(5.29)

where \mathfrak{F}_j is the Fermi-Dirac integral of order j [54] and η is the reduced Fermi level given by $\eta = E_F / k_B T$. It is very clear that the scattering rate in Figure 5.3 is independent of the incoming carrier angle. This is expected as the scattering model returns to the classic BH case.

Plotting the incoming angle dependent scattering and momentum relaxation rates for the remotely screened model at an impurity distance closer to the interface will show the anisotropic behaviour. Using the same impurity density and average carrier energy as Figure 5.3 with an impurity located at $Z_I = 1\lambda_c$, the scattering rate is then plotted in Figure 5.4 and the momentum relaxation rate in Figure 5.5.



Figure 5.4: Plot of the remotely screened scattering rate at $Z_I k_c \approx 1$ over all possible incoming carrier angles.

The anisotropic behaviour of the incoming carrier angle on the scattering model is demonstrated in these figures with a change in the rates with the incoming angle θ_k . Referring to Figure 5.1 (and given in more detail in Figure 5.2), the angle θ_k is the angle of the carrier from the Z-axis. For an angle of $\theta_k = 0$ the carrier is aligned with the Z-axis and is travelling away from the source
interface, and for an angle of $\theta_k = \pi$ the carrier is aligned with the Z-axis and travelling towards the interface.

It is interesting to note that for $\theta_k = 0$ and $\theta_k = \pi$ the scattering and momentum relaxation rates are identical. This behaviour can be understood from the magnitude-squared of the scattering matrix element, equation (5.8), being an even function around q_z . Therefore the rates for $\theta_k = x\pi$ will be identical to that of $\theta_k = \pi - x\pi$ for $0 \le x \le 1$.



Figure 5.5: Plot of the remotely screened momentum relaxation rate at $Z_{I}k_{c} \approx 1$ over all possible incoming carrier angles.

In both Figure 5.4 and Figure 5.5 the rates reach a maximum at $\theta_k = \pi/2$ which corresponds to the point at which the Z-directed momentum transfer, q_z , is at a minimum (the carrier angle is perpendicular to the Z-axis). Examining the scattering matrix element of equation (5.8) again, the third term of the remote screening coefficient (the terms within the square brackets) contains the q_z component as a parameter of the cosine function. When this component, the Z-directed momentum transfer, becomes small the frequency of the cosine function will decrease rapidly. At high frequency the cosine function averages to a value close to zero, but at low frequency the function average becomes much larger. Therefore, as $q_z \to 0$ the third term of the coefficient will increase in value and will therefore increase the scattering rate.

The rotational symmetry around the Z-axis is highlighted in these plots, where the φ_k angle doesn't alter the scattering or momentum relaxation rate in any way. This is plotted in more detail in Figure 5.6 where the momentum relaxation rate has been plotted at three different θ_k angles

showing the constant nature of the φ_k dependence. This is expected as the II is assumed to always lie on the same radial position as the carrier (at the origin) due to the assumption in the potential derivation that $\mathbf{R}_{\mathbf{I}} = 0$.



Figure 5.6: Momentum relaxation rate showing dependence on φ_k for incoming carrier dependent model at $n = N_I = 10^{17} cm^{-3}$.

As mentioned previously the scattering model used in this work is simplified by removing the incoming angle dependence. As this work is to analyse the effect that remote screening has on device performance, the incoming angle dependence is considered to be unnecessary additional complexity. The simplification employed within this model is to assume that the incoming carrier is aligned with the Z-axis such that $\theta_k = 0$ which drastically reduces the momentum transfer wave-vector definitions. Using this simplification in equations (5.25)-(5.27) gives

$$\mathbf{q}_{\perp}^2 = \mathbf{k}^{\prime 2} \sin^2 \theta_{k'} \tag{5.30}$$

$$q_Z^2 = \mathbf{k}^{\prime 2} \left(1 - \cos \theta_{k'} \right)^2 \tag{5.31}$$

$$\mathbf{q}^{2} = \mathbf{q}_{\perp}^{2} + q_{Z}^{2} = 2\mathbf{k}^{\prime 2} \left(1 - \cos \theta_{k}\right)$$
(5.32)

It is important to state here that this assumption means that all scattering events assume that the carrier is aligned such that it is on the Z-axis. Although scattering events will occur that assume the wrong scattering potential, this simplification provides on average a good estimate of the overall scattering effect. A more detailed examination of the effect of the incoming carrier dependence is given below along with a comparison of the models to examine the quality of the

simplification. For reference, the use of the Z-aligned (incoming carrier) simplification, for want of a better name, means that the momentum relaxation weighting term becomes

$$1 - \cos \theta_s = 1 - \cos \theta_{k'} \tag{5.33}$$

Below the incoming carrier dependent scattering model is examined for three different impurity concentrations with a corresponding comparison between the models. The impurity and screening concentrations are equal and the average carrier energy is calculated using equation (5.29) for the scattering & momentum relaxation rate plots.



Figure 5.7: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_I = 10^{15} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model.

Starting with Figure 5.7 by examining the incoming carrier dependence on the (a) scattering and (b) momentum relaxation rates for the lowest concentration of $n = N_I = 10^{15} cm^{-3}$. Here the plots show the rates over a range of impurity positions and as expected, become smaller with decreasing distance from the source interface. Both plot (a) and (b) of this figure are plotted against the angle θ_k of the incoming carrier and show the large shift in the scattering model at $\theta_k = \pi/2$.

Figure 5.8 shows (a) the ratio of mobility and (b) the momentum relaxation rate ratio between the incoming angle dependent model and the Z-aligned model. Here the mobility is calculated from the momentum relaxation rate using the Kubo-Greenwood formula of equation (5.34) [33].

$$\mu = \frac{2e}{3n} \int_{0}^{\infty} dE \rho(E) E \tau_m(E) m_c(E)^{-1} \left(\frac{\partial f_0}{\partial E_F}\right)$$
(5.34)

Here *e* is the electronic charge, *n* is the electron density, $\rho(E)$ is the density of states, $m_c(E)$ is the conductivity effective mass, f_0 is the equilibrium Fermi function and E_F is the Fermi energy. The electron density and the density of states can be written for non-parabolic bands as

$$n = 2\left(\frac{m^* k_B T}{2\pi\hbar^2}\right)^{\frac{3}{2}} \left[\mathfrak{F}_{\frac{1}{2}}(\eta) + \frac{5}{2}\alpha k_B T\mathfrak{F}_{\frac{3}{2}}(\eta)\right] \text{ for } \eta = \frac{E_f}{k_B T}$$
(5.35)

$$\rho(E) = \frac{3(2m^*)^{\frac{3}{2}}}{\pi^2 \hbar^3} E^{\frac{1}{2}} \left(1 + \frac{5}{2}\alpha E\right)$$
(5.36)

In Figure 5.8, the data is arranged such that in plot (a) it is ratio of the Z-aligned model to the full incoming angle model and (b) the ratio of the complete model to the Z-aligned model. Therefore for values greater than one, the Z-aligned model is (a) over-estimating the mobility or (b) under-estimating the momentum relaxation rate of the full incoming carrier model. This is arranged such that the plots show similar behaviour to each other as the mobility calculation depends on the inverse of the momentum relaxation rate.



Figure 5.8: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position.

Plot (a) of Figure 5.8 shows the mobility ratio over a range of impurity positions against the incoming carrier angle with the Z-axis, θ_k . For an impurity located at $Z_I = 4\lambda_c$ the mobility ratio is one at all incoming carrier angles which is expected as the remote screening nature at this distance is very small. As the impurity moves closer to the interface, the effect of the incoming carrier angle becomes much larger and the simplified Z-aligned model over-estimates the mobility by up to ~20%.

The behaviour of the mobility is shown in plot (b) of Figure 5.8. It should be noted that in this plot the momentum relaxation rate for the complete incoming carrier angle model is fixed at an angle of $\theta_k = \pi/2$, the alignment which yields the largest change from the Z-aligned model. This is the case for all the momentum relaxation ratio plots in this section, that is plots (b) of Figure 5.8, Figure 5.10 and Figure 5.12.

Plot (b) shows that the momentum relaxation time, the inverse of the momentum relaxation rate, is over-estimated by the Z-aligned model for $\sim 0.03\lambda_c < Z_I < 4\lambda_c$ with a peak at $Z_I = 0.1\lambda_c$, consistent with the mobility ratio. For values less than this the momentum relaxation rate is underestimated. Although not shown in the mobility ratio plot for this density, this behaviour is repeated for the mobility. There also is some small oscillation in the momentum relaxation ratio around $Z_I = 0.1\lambda_c$.

This oscillating behaviour can be understood by examining the scattering matrix element of equation (5.8), specifically the remote screening coefficient within square brackets. Expanding the exponential terms of this coefficient into a power series and retaining only the first order terms gives

$$f_{sgl} = 2 \left(1 - k_C Z_I \sqrt{1 + \frac{q_L^2}{k_C^2}} \right) \left(1 - \cos(q_Z Z_I) \right)$$
(5.37)

Plot (b) of Figure 5.8 shows that the oscillatory behaviour occurs for small $k_C Z_I$, which mean that the value of the term given by the first bracket of equation (5.37) will be close to one. As discussed earlier in this section, the q_Z component has a minimum for an incoming carrier angle at $\theta_k = \pi/2$ which will lead to a smaller argument of the cosine function. Therefore, for incoming carrier angles close to perpendicular to the Z-axis (small q_Z) and impurity positions close to the interface (small Z_I), the frequency of the cosine function will be greatly reduced. This will cause an increasingly oscillatory behaviour of the coefficient and scattering model, leading to the apparent *flip* from increased scattering to reduced scattering.

To obtain a quantitative measure of the quality of the simplification, an average of the mobility ratio has been taken over a range of impurity positions taking the worst incoming angle case for the complete model. This range of impurity positions, Z_1 , is the complete range of positions where remote screening is effective. Using the mean value theorem for integrals, the following expression is used to obtain this average.

$$\left\langle R_{\mu}\right\rangle = \frac{1}{10\lambda_{c}} \int_{0}^{10\lambda_{c}} dZ_{I} \frac{\mu_{z}}{\mu(\theta_{k} = \pi/2)}$$
(5.38)

where μ_Z is the mobility of the Z-aligned model and $\mu(\theta_k)$ is the complete incoming angle mobility model. The upper limit of $10\lambda_c$ is chosen such that the remote screening effect is negligible, correct at this impurity distance from the interface. Evaluating equation (5.38) for the impurity density of $N_I = 10^{15} cm^{-3}$ yields an average of 1.01325, or an over-estimation of 1.3% of the mobility over the range of impurity positions. Of course, this is for the worst-case scenario, assuming that all carriers scatter with $\theta_k = \pi/2$.

Continuing the examination of the Z-aligned simplification with a higher density of $n = N_I = 10^{17} cm^{-3}$ in Figure 5.9 and Figure 5.10. In plots (a) and (b) we see the consistent reduction in scattering strength with decreasing Z_I . Also evident is the effect of the incoming angle on the scattering model with a shift near the $\theta_k = \pi/2$ point. Although at this higher density the change or *flip* in the scattering behaviour is clear in plot (b) for an impurity position of $Z_I = 0.1\lambda_C$.



Figure 5.9: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_I = 10^{17} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model.

In Figure 5.10 the (a) mobility ratio and (b) momentum relaxation ratios are plotted for this impurity concentration. The underestimation of the momentum relaxation at $Z_I = 0.1\lambda_c$ is very noticeable and is highlighted in the mobility ratio plot of Figure 5.10. The oscillatory behaviour of the scattering model is clear within plot (b) of Figure 5.10 when plotted over the range of impurity positions. For impurity positions greater than $Z_I \approx 0.3\lambda_c$ the mobility is over-estimated and for impurity positions closer to the interface the mobility is under-estimated. This change in behaviour

occurs for electrons interacting with the II at angles perpendicular to the Z-axis where the magnitude of the scattering potential oscillates with the cosine function at small $k_C Z_I$.



Figure 5.10: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position.

At this impurity density of $n = N_I = 10^{17} cm^{-3}$ the oscillatory behaviour is more compressed over the range of impurity positions, Z_I . Referring to the earlier discussion regarding the oscillatory behaviour, this compression in the fluctuation can be understood from the larger value of $k_C Z_I$. That is, the screening length will be smaller at higher concentrations leading to larger values of k_C and reduced oscillation. Completing the average mobility ratio using equation (5.38) yields an average overestimation of roughly 1.4% for this density.



Figure 5.11: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_I = 10^{19} cm^{-3}$ over a range of impurity positions for the incoming carrier dependent remote screening model.

Continuing the evaluation of the Z-aligned approximation at an impurity density of $n = N_1 = 10^{19} cm^{-3}$, the reduction in the oscillatory behaviour seems to be further enhanced with the

increasing density. The scattering and momentum relaxation rates of Figure 5.11 don't show a significant shift with the incoming carrier angle.



Figure 5.12: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over impurity position.

The mobility ratio, plot (a) of Figure 5.12 shows a majority under-estimation of the mobility over the range of impurity positions plotted. Looking at the momentum relaxation rates of plot (b), there is a very small region of overestimation around $Z_I = 3 - 5\lambda_C$ and a large region of underestimation. It is clear that at this high impurity density the oscillations of the model are reduced in comparison to lower densities but lead to underestimation of the mobility. Completing the average mobility ratio from equation (5.38) for this density gives an average underestimation of around 2%.

In summary the alignment of the incoming carrier to the Z-axis reduces the complexity of the scattering model and preserves the remote screening effect. For remotely screened II scattering the average mobility using the Z-aligned model is within 1.5-2% of the complete model over the region that remote screening is effective.

5.2.3.1 Z-Aligned Model

Use of the Z-aligned momentum transfer wave-vectors of equations (5.30)-(5.32) in the scattering model allows the integral over the $\varphi_{k'}$ angles to be completed. As there is no longer a $\varphi_{k'}$ dependence, the integral over those angle can be completed to yield 2π . For reference, the Z-aligned scattering model equations can then be written as

5.2 Single Interface Scattering Model

$$\sigma(\theta_{k'}) = \left(\frac{Ze^2}{4\pi\varepsilon_{s_l}\varepsilon_0}\right)^2 \frac{2^3\pi m_d^2}{\hbar^4} (1+2\alpha E)^2 \frac{1}{\left(\mathbf{q}^2+k_c^2\right)^2} \times \left[1+\exp\left(-2Z_I\sqrt{\mathbf{q}_\perp^2+k_c^2}\right) - 2\cos\left(q_Z Z_I\right)\exp\left(-Z_I\sqrt{\mathbf{q}_\perp^2+k_c^2}\right)\right]$$
(5.39)

for the differential scattering cross section,

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^3\pi m_d k^*}{\hbar^3} (1+2\alpha E) \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}}{\left(\mathbf{q}^2+k_C^2\right)^2} \times \left[1+\exp\left(-2Z_I \sqrt{\mathbf{q}_{\perp}^2+k_C^2}\right) - 2\cos\left(q_Z Z_I\right)\exp\left(-Z_I \sqrt{\mathbf{q}_{\perp}^2+k_C^2}\right)\right]$$
(5.40)

for the total scattering rate and

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{s_l}\varepsilon_0}\right)^2 N_I \frac{2^3\pi m_d k^*}{\hbar^3} (1+2\alpha E) \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}(1-\cos\theta_{k'})}{\left(\mathbf{q}^2+k_c^2\right)^2} \times \left[1+\exp\left(-2Z_I \sqrt{\mathbf{q}_{\perp}^2+k_c^2}\right) - 2\cos\left(q_Z Z_I\right) \exp\left(-Z_I \sqrt{\mathbf{q}_{\perp}^2+k_c^2}\right)\right]$$
(5.41)

for the momentum relaxation rate.

5.2.4 Analysis of Scattering Rate

Using the different methods of utilising the scattering probability the remotely screened impurity scattering process will now be analysed. Looking at the differential scattering cross-section given by equation (5.39) first, which is plotted in a set of polar plots over the scattering angle $\theta_{k'}$ in Figure 5.13. These plots are for two different screening densities at different carrier energies and impurity locations.

Plot (a) in the figure corresponds to a moderate screening density of approximately $n = 10^{17} cm^{-3}$. At this screening density the differential cross-sectional scattering area is very large which will lead to large scattering rates. In the screening conditions of plot (a) it is clear to see that the scattering process favours forward scattering events. That is, events for which the electron scatters through angles of less than $\pi/2$. The differential scattering cross-section for scattering angles greater than $\pi/2$ or back-scattering, depicted by the negative component of the horizontal axis, is very small in comparison. At higher energies the tendency to forward scatter is increased as shown by the dotted lines in plot (a) of Figure 5.13. The back-scattering component is extremely small in this case and the range of angles for forward scattering is greatly decreased to a narrow range around zero degrees. From this behaviour it can be deduced that the scattering interaction has a lesser effect at higher carrier energies.



Figure 5.13: Polar plots of the differential scattering cross-section for an electron at low (solid lines) and high energy (dashed lines) at various impurity locations.

At the higher screening density of $n = 10^{19} cm^{-3}$ plotted in (b), the differential scattering cross-section is considerably smaller. At low carrier energies the scattering cross-section in plot (b) has an almost isotropic nature with no particular scattering angle favoured. This suggests that impurity scattering will have a larger effect at this screening density and carrier energy due to the wide range of probable scattering angles. For higher carrier energies the scattering cross-section shifts towards a forward-angle scattering preference which is comparable with the situation at the lower screening density of plot (a).

This can be understood by examining the differential scattering cross section given by equation (5.39), in particular the terms involving an angular dependence. These angular dependent terms have been rewritten here as a separate function, f_{θ} ,

$$f_{\theta} = \frac{1}{\mathbf{q}^4 \left(1 + k_c^2 / \mathbf{q}^2\right)^2}$$

$$\times \left[1 + \exp\left(-2q_{\perp} Z_I \sqrt{1 + k_c^2 / \mathbf{q}_{\perp}^2}\right) - 2\cos\left(q_Z Z_I\right) \exp\left(-q_{\perp} Z_I \sqrt{1 + k_c^2 / \mathbf{q}_{\perp}^2}\right)\right]$$
(5.42)

where the momentum transfer wave-vectors are given by equations (5.30)-(5.32). This function highlights the relation between the screening density and the carrier energy given by the terms $(1+k_c^2/\mathbf{q}^2)$ and $(1+k_c^2/\mathbf{q}_{\perp}^2)$. When the screening density is large, the inverse screening length k_c is large and these terms are also very large (\gg 1) which causes the function f_{θ} to become very small and the scattering more anisotropic. Conversely when the screening density is low along with the inverse screening length, the terms are very small (\ll 1) and the function will lead to more isotropic scattering. This behaviour is of course dependent on carrier energy with higher carrier energies always leading to more anisotropic scattering.

With decreasing distance between the impurity and the source interface, Z_I , the scattering crosssection is reduced in both cases of Figure 5.13. As the remote screening interaction is expected to increase the screening of an impurity, as discussed in Chapter 4, the reduction in scattering crosssection is logical. The increased screening of the scattering impurity centre as it moves towards the interface clearly reduces the scattering cross-section and will lead to a reduction in the scattering rate.

Plots of the scattering and momentum relaxation rates, equations (5.40) and (5.41), for the same conditions in the above polar plots are given in Figure 5.14 and Figure 5.15. In both of the figures below the II concentration is taken to equal the screening density, that is $n = N_1$.



Figure 5.14: (a) Scattering and (b) momentum relaxation rates under the low screening conditions of $n = N_1 = 10^{17} cm^{-3}$ for various impurity locations.

As discussed above, the scattering differential cross-section is reduced with decreasing Z_1 and becomes less effective at higher energies. This behaviour is clearly seen in the plots of Figure 5.14

which give the scattering rates (a) and momentum relaxation rates (b) for the lower concentration of $n = N_I = 10^{17} cm^{-3}$. Figure 5.15 of the higher screening density $n = N_I = 10^{19} cm^{-3}$ also shows this common behaviour which is expected for the remotely screened impurity scattering model.



Figure 5.15: (a) Scattering and (b) momentum relaxation rates under the high screening conditions of $n = N_1 = 10^{19} cm^{-3}$ for various impurity locations.

Comparing the scattering rates between the two figures shows around an order of magnitude decrease at the higher concentration following the trend in the radius of the differential scattering cross-section.

In part (b) of Figure 5.14 the momentum relaxation rate for carriers greater than 10meV doesn't show much effect of remote-screening as the impurity position is varied between $Z_I = \{0.5..4\}\lambda_c$. Looking at the high energy polar plot of this case, plot (a) of Figure 5.13, the differential scattering cross-section varies dramatically in radius around scattering angles close to zero degrees.

This behaviour can be understood by looking at the momentum relaxation rate given by equation (5.41), particularly the weighting term, $(1-\cos\theta_{k'})$. For small angle forward scattering this weighting term becomes very small and the momentum is relaxed on much longer scale. Therefore, in this case the scattering rate will be affected by the decreasing scattering cross-section radius but the momentum relaxation rate will show little change as the range of scattering angles remains close to zero.

Using the momentum relaxation rate of equation (5.41) the electron mobility can be analytically calculated providing more insight into the remote screening effect. Using the Kubo-Greenwood formula of equation (5.34) allows the electron mobility to be calculated. Here the ratio between the

5.2 Single Interface Scattering Model

mobilities of the remotely screened model and the BH model are plotted to allow comparison. The BH model is found as the limit $Z_1 \rightarrow \infty$ of the remotely screened model, given as equation (5.43). Noting of course, that the BH model is independent of Z_1 .

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^3\pi m_d k^*}{\hbar^3} (1+2\alpha E) \int_0^{\pi} d\theta_k \cdot \frac{\sin\theta(1-\cos\theta_{k^*})}{\left(\mathbf{q}^2+k_c^2\right)^2}$$
(5.43)

In the figures below the ratio between the mobilities is given for different doping concentrations where the screening density is assumed to equal the doping concentration. In both figures the ratio is taken as the remotely screened mobility over the BH mobility.



Figure 5.16: Ratio between the mobilities of the remotely screened and Brooks-Herring models. The impurity position from the source interface, Z_i , is given in (a) units of the respective channel screening length, λ_c , and (b) nanometres from the source interface.

In plot (a) of Figure 5.16 the ratio between the mobilities of the two models is plotted over the impurity distance from the source interface in units of the respective channel screening length. This follows the representation used in the plots of the differential scattering cross-section and the scattering/momentum relaxation rates. Plot (a) of Figure 5.16 shows that remote screening effect begins to strongly affect the impurity limited mobility at positions less than two screening lengths from the source interface. Beyond two screening lengths the remotely screened model is a close match with the BH model.

For impurity positions closer to the interface the remotely screened mobility increases heavily as the effect of polarisation charges becomes stronger. In particular, the mobility for the high channel density given by the green curve shows a dramatic increase to over 30x the BH mobility. This increase is very large and is a side-effect of the strong-screening limit. Using equation (3.36), the screening length for the screening/dopant concentration of $n = N_I = 10^{19} cm^{-3}$ is $\lambda_C = 1.39 nm$. As

discussed at the end of the previous chapter, the strongly screened model over-estimates the remote-screening effect for impurities located closer than 1nm from the source interface. Therefore at this density the point that the strongly screened model begins to overestimate the remote screening effect corresponds to $Z_I \approx 0.7 \lambda_C$.

The figures in this section have all been plotted with a length scale normalised to the channel screening length which allows the scattering model to be analysed and compared over a wide range of screening densities and impurity concentrations as well as comparison with the previous figures in this section. Despite this, it is important here to also examine the mobility using a fixed length scale. In plot (b) of Figure 5.16 the mobility ratio between the remotely screened and BH model is given over a physical impurity distance from the source interface.

Using a physical length scale for each screening/background doping concentration in plot (b) of Figure 5.16 provides an example closer to a real scenario of the remote screening effect on the impurity limited mobility. This figure highlights the dependence of the channel screening density, where the lower the screening density the stronger the mobility increase due to remote screening. This plot does not follow the trend of Figure 5.16 which reports that the higher the screening density, the larger the remote screening effect. This reversal in behaviour between the figures can be understood from the screening length.

Using equation (3.36) the screening length can be found as $\lambda_c = \{126.27, 12.64, 1.39\} nm$ for the screening densities of $n = \{10^{15}, 10^{17}, 10^{19}\} cm^{-3}$ respectively. As discussed above, the remote screening effect starts to alter the mobility at roughly $Z_I = 2\lambda_c$. Therefore, at the highest screening density the remote screening effect will only alter impurities located within ~ 2.8nm whereas at the lowest density we can expect a shift at distances up to ~ 250nm. For the lowest density involved at one nanometre from the interface, Z_I corresponds to roughly $8 \times 10^{-3} \lambda_c$ which leads to a large increase in the mobility of plot (b).

5.3 Double Interface Scattering Model

The double interface scattering model is found using an identical procedure to the single interface model. A scattering rate is defined using Fermi's Golden Rule approach based on the strongly screened impurity potential found for this model in the last chapter. The scattering potential is again not spherically symmetric and the Z-aligned incoming carrier simplification is applied.

This section will start with the calculation of the scattering matrix element in section 5.3.1. Following this the differential scattering cross-section, total scattering rate and the momentum relaxation rate will be obtained in section 5.3.2. The incoming carrier Z-alignment simplification is discussed in section 5.3.3 and finally the double interface remotely screened scattering model will be analysed in section 5.3.4.

5.3.1 Scattering Matrix Element

The strongly screened double interface potential found in section 4.3 is used to calculate the scattering matrix element for this model. The potential for the double interface remotely screened impurity potential is valid only in the channel region, that is the region of $0 < Z < L_c$. The calculation for the scattering matrix element must be integrated over this region only. Correcting the Z-space integral of equation (5.2) gives the following form for the scattering matrix element

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{0}^{L_{c}} dZ \int_{0}^{2\pi} d\phi \int_{0}^{\infty} d\mathbf{R} \ eV(\mathbf{R}, Z) \exp(-i\mathbf{q}_{\perp} \cdot \mathbf{R}) \exp(-iq_{Z}Z)$$
(5.44)

Substituting the scattering potential into the matrix element definition and completing the integrals gives, after some lengthy manipulation

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{eQ}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}} \frac{2\pi}{\Omega} \frac{1}{K_{c}^{2}} \left\{ 2\exp\left(-iq_{z}Z_{I}\right) - \exp\left(-Z_{I}K_{c}\right) \left(1 + \frac{iq_{z}}{K_{c}}\right) - \exp\left(-iq_{z}L_{c}\right)\exp\left((Z_{I} - L_{c})K_{c}\right) \left(1 - \frac{iq_{z}}{K_{c}}\right) - \exp\left(-Z_{I}K_{c}\right) \left(1 - \exp\left(-iq_{z}L_{c}\right)\exp\left(-L_{c}K_{c}\right)\right) \left(1 - \frac{iq_{z}}{K_{c}}\right)A_{m} - \exp\left(-Z_{I}K_{c}\right) \left(\exp\left(-iq_{z}L_{c}\right)\exp\left(L_{c}K_{c}\right) - 1\right) \left(1 + \frac{iq_{z}}{K_{c}}\right)A_{n} \right\}$$

$$(5.45)$$

Here the coefficients A_m and A_n are given by the equations (5.46)-(5.47) below and the terms K_C, K_S and K_D are defined in equation (5.48).

$$A_{m} = \frac{\exp(2L_{C}K_{C}) - \exp(2Z_{I}K_{C})}{\exp(2L_{C}K_{C}) - 1}$$
(5.46)

$$A_{n} = \frac{\exp(2Z_{I}K_{C}) - 1}{\exp(2L_{C}K_{C}) - 1}$$
(5.47)

$$K_{C} = \sqrt{\mathbf{q}_{\perp}^{2} + k_{C}^{2}}; \quad K_{S} = \sqrt{\mathbf{q}_{\perp}^{2} + k_{S}^{2}}; \quad K_{D} = \sqrt{\mathbf{q}_{\perp}^{2} + k_{D}^{2}}$$
(5.48)

Taking the term for the scattering matrix element, $\mathbf{H}_{\mathbf{k'k}}$, of equation (5.45) and completing the square of the magnitude to obtain

$$\left|\mathbf{H}_{\mathbf{k}'\mathbf{k}}\right|^{2} = \left(\frac{eQ}{4\pi\varepsilon_{Si}\varepsilon_{0}}\right)^{2} \frac{1}{\Omega^{2}} \left(\frac{4\pi}{\mathbf{q}_{\perp}^{2} + q_{Z}^{2} + k_{C}^{2}}\right)^{2} f_{dbl}\left(k_{C}, Z_{I}, L_{C}\right)$$
(5.49)

Here a new function defining the interaction with the source and drain regions, f_{dbl} , has been introduced and is given as

$$f_{dbl}(k_{C}, Z_{I}, L_{C}) = \frac{\exp(-2(L_{C} + Z_{I})K_{C})}{\left(\exp(2L_{C}K_{C}) - 1\right)^{2}} \left[\left(E_{C} - E_{S}\cos(q_{Z}Z_{I}) - E_{D}\cos(q_{Z}L_{C})\right)^{2} - \left(E_{S}\sin(q_{Z}Z_{I}) + E_{D}\sin(q_{Z}L_{C})\right)^{2} \right]$$
(5.50)

New coefficients have been formed to simplify this expression and are given by equations (5.51)-(5.53) below.

$$E_{c} = \exp((L_{c} + 2Z_{I})K_{c})(\exp(2(L_{c} - Z_{I})K_{c}) - 1)$$
(5.51)

$$E_{s} = \exp((L_{c} + Z_{I})K_{c})(\exp(2L_{c}K_{c}) - 1)$$
(5.52)

$$E_D = \exp\left(2L_C K_C\right) \left(1 - \exp\left(2Z_I K_C\right)\right) \tag{5.53}$$

5.3.2 Scattering Rate

The double interface remotely screened impurity scattering model can now be developed using the scattering matrix element found in the previous section. The probability of scattering can be found using Fermi's Golden rule, substituting in the scattering matrix element gives

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} \left(\frac{eQ}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 \frac{1}{\Omega^2} \left(\frac{4\pi}{\mathbf{q}_{\perp}^2 + q_z^2 + k_c^2}\right)^2 f_{dbl}\left(k_c, Z_l, L_c\right) \delta\left(E\left(\mathbf{k}^{*'}\right) - E\left(\mathbf{k}^{*}\right)\right) \quad (5.54)$$

The differential scattering cross-section can be calculated from this probability using equation (5.11) and defining the impurity charge to be Q = Ze, the number of unit charges of the impurity multiplied by the electronic charge.

$$\boldsymbol{\sigma} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 \frac{2^2 m_d^2}{\hbar^4} \left(1 + 2\alpha E\right)^2 \frac{1}{\left(\mathbf{q}^2 + k_c^2\right)^2} f_{dbl}\left(k_c, Z_I, L_C\right)$$
(5.55)

Substituting the scattering probability defined in equation (5.54) into the total scattering rate, equation (5.14) and completing the k' integral gives

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^2 m_d k^*}{\hbar^3} \left(1 + 2\alpha E\right) \int_0^{2\pi} d\varphi_{k'} \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}}{\left(\mathbf{q}^2 + k_C^2\right)^2} f_{dbl}\left(k_C, Z_I, L_C\right) \quad (5.56)$$

Similar to the single interface model, the final integrals over the $\theta_{k'}$ and $\varphi_{k'}$ components will be discussed in the following section. The momentum relaxation rate for the scattering process can be simply defined by inserting the weighting term for the change in momentum into the theta integral of equation (5.56), yielding

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 \frac{N_I 2^2 m_d k^*}{\hbar^3} (1+2\alpha E) \\ \times \int_0^{2\pi} d\varphi_k \int_0^{\pi} d\theta_k \frac{\sin\theta_{k'}(1-\cos\theta_s)}{\left(\mathbf{q}^2+k_c^2\right)^2} f_{dbl}\left(k_c, Z_I, L_c\right)$$
(5.57)

5.3.3 Scattering Reference Frame

The double interface model will also employ the simplified scattering reference frame in an approach identical to that used with the single interface remote screening model. That is the incoming carrier will be aligned with the Z-axis of the scattering reference frame reducing the angular dependency of the scattering model. Following a similar analysis to that of section 5.2.3, this simplification will be tested for the double interface scattering model.

The fixed scattering reference frame given by Figure 5.1 and Figure 5.2 remains unchanged for the double interface model. The momentum transfer wave vector relations presented in section 5.2.3 for the full incoming carrier case, equations (5.25)-(5.27), and the Z-aligned simplified relations given by equations (5.30)-(5.32) are again used.

The scattering rate, momentum relaxation rate and the mobility analysis are completed for three impurity concentrations, $N_I = \{10^{14}, 10^{16}, 10^{18}\} cm^{-3}$. Again as the potential assumes that the impurity is located at the origin of the radial axis, $R_I = 0$, there is rotational symmetry around the Z axis, the $\varphi_{k'}$ angle. The scattering and momentum relaxation rate plots assume an incoming carrier with average thermal energy given by equation (5.29) and the mobilities are calculated using the Kubo-Greenwood method of equation (5.34). To analyse the double interface model effectively the channel length, L_C , is varied with the impurity position located at exactly half the channel length, $Z_I = 0.5L_C$, in each of the cases below.



Figure 5.17: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{14} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model.

Starting with the lowest concentration of $N_I = 10^{14} cm^{-3}$ in Figure 5.17 and Figure 5.18, the incoming carrier model clearly shows an increased scattering effect for carrier angles around $\theta_k = \pi/2$. This is very much consistent with the single interface model and can be again related to the momentum transfer in the Z-axis given by q_Z . For incoming carrier angles close to perpendicular to the Z axis there will be minimal momentum transfer in the Z plane and the q_Z component will become very small. Looking at the double interface coefficient in the scattering matrix element, equation (5.50), for small values of this momentum transfer component the coefficient will be large and hence the scattering will be increased.



Figure 5.18: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length.

At this low density the simplified model overestimates the momentum relaxation over a wide range of channel lengths as shown in plot (b) of Figure 5.18. Completing an average over a wide range of channel lengths using the mean-value theorem for integrals given by equation (5.58) below, yields an average overestimation of around 1.8%.

$$\langle R_{\mu} \rangle = \frac{1}{\left(20\lambda_{C} - 10^{-5}\lambda_{C}\right)} \int_{10^{-5}\lambda_{C}}^{20\lambda_{C}} dL_{C} \frac{\mu_{Z}}{\mu(\theta_{k'} = \pi/2)}$$
 (5.58)

For the double interface model it is not possible to complete the average value from an effective channel length of zero and instead a value of $10^{-5} \lambda_c$ is chosen as a non-zero channel length. The upper limit of $20\lambda_c$ coincides with the upper limit of the single interface case (given by equation (5.38)), allowing $10\lambda_c$ from each interface.





Figure 5.19: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{16} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model.

Examination of the intermediate impurity density of $N_I = 10^{16} \text{ cm}^{-3}$ in Figure 5.19 and Figure 5.20 shows that the simplified model overestimates the scattering for channel lengths greater than $L_c \approx 0.2\lambda_c$. At this density the momentum relaxation ratios plotted in part (b) of Figure 5.20 also demonstrate that underestimation occurs at small channel lengths. Completing the average mobility ratio using equation (5.58) for this density yields an average overestimation of around 2.4%.

The momentum relaxation ratio oscillates very much like the single interface model discussed in section 5.2.3 and shows the increased compression of this oscillation at higher screening concentration. Whereas in the single interface model the scattering matrix element has a single cosine function, here the double interface scattering matrix element coefficient, equation (5.50), has a set of cosine and sine functions all dependent on the Z directed momentum transfer, q_Z .



Figure 5.20: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length.

At the highest impurity concentration given by Figure 5.21 and Figure 5.22 the behaviour of the remotely screened double interface model differs to the behaviour seen this far. The scattering rate and momentum relaxation rate plots at a channel length of $L_c = 2\lambda_c$ show a double oscillation. This double oscillation has peaks at $\theta_k = \pi/4, 3\pi/4$ and a trough at $\theta_k = \pi/2$ for the momentum relaxation rate of plot (b) in Figure 5.21 with a much less understated double oscillation around the $\theta_k = \pi/2$ point in both the scattering rate and mobility ratio.



Figure 5.21: (a) Scattering and (b) momentum relaxation rates for an impurity concentration of $N_1 = 10^{18} cm^{-3}$ over a range of channel lengths for the incoming carrier dependent, remote screening model.

This double oscillation at small channel lengths is very much an effect brought on by the set of cosine and sine functions in the scattering matrix element. Examining the f_{dbl} coefficient of the scattering matrix element of equation (5.50), the cosine and sine functions both depend on the Z-directed momentum transfer q_Z and either the impurity position, Z_I , or channel length, L_C .

5.3 Double Interface Scattering Model

As has been discussed previously the q_Z component will tend to zero as the incoming carrier angle is close to perpendicular to the Z axis. This reduces the frequency of these angular functions which lead to the oscillatory behaviour. Here the angular functions will have two separate frequencies depending on the impurity position and channel length as q_Z is reduced. In the cases plotted here the impurity position is taken to be always half the channel length, that is $Z_I = 0.5L_C$, but it will always be the case that $Z_I < L_C$.



Figure 5.22: (a) Ratio of mobilities between the remote screening models over incoming carrier angle θ_k and (b) momentum relaxation ratios between models over channel length.

It can be seen that although this has a considerable effect on the momentum relaxation rate, the effect on the mobility is relatively minimised. This of course is due to the fact that the momentum relaxation rate is integrated over all energy in the mobility calculation. For this higher density the average mobility ratio is calculated to give an average underestimation of around 1%.

5.3.3.1 Z-Aligned Model

Using the Z-aligned momentum transfer wave-vector relations of equations (5.30)-(5.32) allows the integral over the angle $\varphi_{k'}$ to be completed and yields the following final expressions for the double interface remotely screened scattering model.

$$\boldsymbol{\sigma} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 \frac{2^3\pi m_d^2}{\hbar^4} \left(1 + 2\alpha E\right)^2 \frac{1}{\left(\mathbf{q}^2 + k_c^2\right)^2} f_{dbl}\left(k_c, Z_l, L_c\right)$$
(5.59)

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2^3\pi m_d k^*}{\hbar^3} \left(1 + 2\alpha E\right) \int_0^{\pi} d\theta_k \cdot \frac{\sin\theta_{k'}}{\left(\mathbf{q}^2 + k_C^2\right)^2} f_{dbl}\left(k_C, Z_I, L_C\right)$$
(5.60)

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{Si}\varepsilon_0}\right)^2 \frac{N_I 2^3 \pi m_d k^*}{\hbar^3} (1+2\alpha E) \times \int_0^{\pi} d\theta_{k'} \frac{\sin\theta_{k'}(1-\cos\theta_s)}{(\mathbf{q}^2+k_c^2)^2} f_{dbl}(k_c, Z_I, L_c)$$
(5.61)

5.3.4 Analysis of Scattering Rate

An analysis of the scattering model defined in section 5.3.2 will now be completed in a similar format to that completed for the single interface model. Here, the analysis of the double interface model will be centred around the effect that a small channel length has on the remotely screened impurity scattering model.

Starting with the differential scattering cross-section which has been plotted in Figure 5.23. For these plots the screening density is taken as (a) $n = N_I = 10^{16} cm^{-3}$ and (b) $n = N_I = 10^{18} cm^{-3}$ which leads to screening lengths of $\lambda_C = 39.95nm$ and $\lambda_C = 4.04nm$ respectively.



Figure 5.23: Polar plots of the double interface scattering cross-section for varying channel lengths with a single impurity located mid-channel, $Z_1 = 0.5L_c$.

Similar to the behaviour of the single interface model, the low screening density differential scattering cross-section demonstrates mostly low-angle forward scattering. At high energies the differential scattering cross-section for this low density is restricted to a very small range of scattering angles around zero degrees. The momentum relaxing effect of such small scattering angles will be minimal.

The differential scattering cross-section for the low screening density, plot (a) of Figure 5.23 has an extremely large radius. As the channel length decreases, the remote screening effect reduces this radius and will reduce the scattering rate. The high screening density differential scattering cross-

section of plot (b) is somewhat more isotropic for low energy carriers than the low density plots. Plotted in Figure 5.25 are the scattering and momentum relaxation rates for the double interface remotely-screened impurity model with the low screening density conditions of the polar plots given above.



Figure 5.24: (a) Scattering and (b) momentum relaxation rates under the low screening conditions of $n = N_I = 10^{16} cm^{-3}$ over a range of channel lengths with a single impurity located mid-channel, $Z_I = 0.5L_c$.

As the differential scattering cross section predicted, the scattering rate does decrease with the reducing distance between the source and drain interfaces but the momentum relaxation rate is largely unaffected. Plot (b) of Figure 5.24 shows a small difference between the different channel lengths at the very low energies around 0.1meV which will have negligible effect in room temperature silicon. Nearly all the scattering at this low screening density is small angle forward scattering as seen by plot (a) of Figure 5.23, which is not altered by increasing the remote screening induced by the source and drain. Despite the small angle forward scattering, the momentum relaxation rate is high for low energy carriers but drops off very rapidly with energy.



Figure 5.25: (a) Scattering and (b) momentum relaxation rates under the high screening conditions of $n = N_I = 10^{18} cm^{-3}$ for varying channel lengths with a single impurity located mid-channel, $Z_I = 0.5L_c$.

Figure 5.25 shows (a) the scattering and (b) the momentum relaxation rates for the higher density. The scattering rate is clearly affected by the reduction in channel length as is the momentum relaxation rate for low energy carriers. Above approximately 20meV the momentum relaxation rates becomes largely unaffected by the increase in the remote screening effect as the tendency to small angle scattering becomes more dominant.

The Kubo-Greenwood formula of equation (5.34) is used to plot a comparison of the double interface remotely screened impurity mobility against the BH model of equation (5.43). The BH model being the limit of $Z_I \gg 0$ and $L_C \rightarrow \infty$ of the double interface model. In plot (a) of Figure 5.26 the ratio of the remotely screened mobility to the BH mobility is given over a range of channel lengths which have been normalised to the respective channel screening lengths. For the three densities plotted here, $n = N_I = 10^{14} cm^{-3}$, $10^{16} cm^{-3}$, $10^{18} cm^{-3}$, the screening lengths are calculated as $\lambda_C = 399.54 nm$, 39.95 nm, 4.04 nm respectively.



Figure 5.26: Ratio between the mobilities of the remotely screened and Brooks-Herring models. The channel length, L_c , is given in units of (a) the respective channel screening length, λ_c , and (b) nanometres where the impurity is always positioned at $Z_I = 0.5L_c$.

Plot (a) of Figure 5.26 clearly shows that beyond $L_c = 5\lambda_c$ the double interface remote screening has little effect on the impurity limited mobility. This plot also shows a large increase in the mobility at a channel lengths less than λ_c for the highest doped case represented by the green curve. This large increase in mobility is likely due to the strongly screened model over estimating the effect of remote screening and a quick calculation shows that at this point the impurity is located around $Z_1 = 2nm$. It is around this distance that the strongly screened model is expected to overestimate the effect, as discussed in the previous chapter.

Plot (b) of Figure 5.26 gives the mobility comparison using a fixed reference for the channel length at each density. This figure highlights the strength of the interaction for low screening density channels where the screening length is extremely long. For the lowest density, given by the blue curve, the screening length is almost 400*nm* which for the remote screening interaction to dissipate, would require a channel length of $5\lambda_c \approx 2\mu m$.

5.4 Sub-Threshold Impurity Scattering

It is well known that scattering rates that are based on the Coulomb potential can diverge unless they are well bounded. In the BH approach the Coulomb potential is screened by the mobile charge density which generally provides a good bound on the Coulomb potential. There are other approaches which use half the average inter-ion distance to confine the effect of Coulomb scattering such as the CW model and Ridley's Third-Body Exclusion (TBE) technique [47, 55]. These models are discussed in detail in the literature review of II scattering in Chapter 2.

5.4 Sub-Threshold Impurity Scattering

The remotely screened impurity scattering model that has been developed in this chapter follows a similar approach to that of BH in that it uses the mobile charge density to restrict the range of scattering. This approach works well when the mobile charge or screening density is greater than or equal to the II density. Under certain conditions such as a MOS capacitor or a MOSFET device at low gate bias, the screening density can be much lower than the fixed charge density and can approach intrinsic silicon carrier densities. At such low carrier concentrations the scattering rates for the BH approach, and similarly the remote screening model, will tend to very high values.



Figure 5.27: Plots of (a) the scattering rates and (b) the momentum relaxation rates for standard II scattering models at sub-threshold conditions with $N_1 = 10^{18} cm^{-3}$ and $n = 10^{12} cm^{-3}$.

The three aforementioned II scattering models (CW, BH and TBE) are plotted in Figure 5.27 with high background impurity density and a low screening density, a typical sub-threshold case. In these plots the difference between using the the CW and TBE approaches of the inter-ion separation distance as a cut-off and using the BH approach of screening by mobile charge is clear to see. Plot (a) of this figure shows that with very low screening densities the BH model begins to diverge and results in a scattering rate of roughly $\Gamma_{BH} \approx 10^{20} s^{-1}$. This is far too large for efficient numerical simulation considering that within a general bulk Monte Carlo simulation the timestep is around $\Delta t \approx 10^{-15} s$ which leads to almost one hundred thousand events per timestep. Considering that it is typical to have only a few events per timestep, the processing overhead of using the BH model in these conditions becomes very large.

Looking at the momentum relaxation rates for the differing models in plot (b) of Figure 5.27 highlights the effect of limiting the scattering model through the average inter-ion separation. This difference in the momentum relaxation rates will alter the impurity limited mobility in device simulations directly. Therefore use of the TBE or CW approaches will lead to an underestimation of the effect that II scattering has on the mobility.

5.4 Sub-Threshold Impurity Scattering

The method for simulating II scattering in the Monte Carlo simulator used here is the TBE model as it overcomes the divergent scattering rate problem with low screening densities. It has traditionally been used in this Monte Carlo simulator and the loss of accuracy incurred is neglected in favour of efficient simulation. The loss of accuracy can be highlighted by examining the TBE approach applied to the remote screening model developed in this work.

The loss of accuracy lies with the cut-off used by the TBE model which utilises the impact parameter definition, given by equation (5.62). The scattering events which have an impact parameter beyond half the inter-ion separation distance are regarded as being more probable to scatter from another scattering centre and therefore are excluded.

$$\pi b(\theta)^2 = 2\pi \int_{\theta}^{\pi} \sin \theta_b \ \sigma(\theta_b) d\theta_b$$
(5.62)

To understand why this cut-off approach can be considered inaccurate it is necessary to examine the impact parameter, in particular for the case of remotely screened impurity scattering. Substituting the single interface differential scattering cross-section of equation (5.39) into the impact parameter definition above gives

$$\pi b\left(\theta\right)^{2} = \left(\frac{Ze^{2}}{4\pi\varepsilon_{st}\varepsilon_{0}}\right)^{2} \frac{2^{4}\pi^{2}m^{*2}}{\hbar^{4}} \left(1+2\alpha E\right)^{2} \int_{\theta}^{\pi} d\theta_{b} \frac{\sin\left(\theta_{b}\right)}{\left(\mathbf{q}^{2}+k_{c}^{2}\right)^{2}} \times \left[1+\exp\left(-2Z_{I}\sqrt{\mathbf{q}_{\perp}^{2}+k_{c}^{2}}\right)-2\cos\left(q_{Z}Z_{I}\right)\exp\left(-Z_{I}\sqrt{\mathbf{q}_{\perp}^{2}+k_{c}^{2}}\right)\right]$$
(5.63)

In Figure 5.28 the impact parameter is plotted at various impurity positions for the sub-threshold conditions used in Figure 5.27. Also plotted is the average, half inter-ion separation distance given by $a = \frac{1}{2} (2\pi N_I)^{-\frac{1}{3}}$.



Figure 5.28: Impact parameter for single interface remotely screened impurity model against half the average inter-ion separation distance for sub-threshold conditions of $N_I = 10^{18} cm^{-3}$ with a low carrier density of $n = 10^{12} cm^{-3}$.

The plot highlights the large range of scattering events that will be excluded by the TBE model. In particular, the correction to the II scattering model introduced by the remote screening effect is excluded using this approach. Therefore in sub-threshold conditions the use of the TBE model will neglect a great deal of impurity scattering including the effect of remote screening. To properly account for all II scattering in sub-threshold conditions the full BH approach with remote screening must be modelled, thereby avoiding any underestimation of the importance of II scattering on the mobility.

In this work a new method has been developed to retain the accuracy of the BH approach to II scattering whilst controlling the behaviour in sub-threshold conditions where the scattering rate tends to large values. This new method restricts the scattering rate but maintains the original momentum relaxation rate, and is possible through the introduction of a simple numerical cap developed by Dr. Jeremy Watling [111]. This numerical cap will first be discussed for the BH model of scattering in section 5.4.1. In section 5.4.2 the application of this correction to the remotely screened models is presented. Finally in section 5.4.3 the model is tested with the Monte Carlo simulator used in this work.

5.4.1 Numerical Cap to II Scattering

This numerical cap to the II scattering process was designed to allow use of the complete BH formulation in Monte Carlo simulations without the drawback of large simulations times. Starting by defining the original BH scattering rate and momentum relaxation rate as Γ_{II}^{BH} and $1/\tau_m^{BH}$ respectively.

$$\Gamma_{II}^{BH} = K_{II} \int_{0}^{\pi} P(\theta) d\theta$$
(5.64)

$$\frac{1}{\tau_m^{BH}} = K_{II} \int_0^{\pi} P(\theta) (1 - \cos(\theta)) d\theta$$
(5.65)

Here the term in front of the theta integral has been separated into K_{II} given by equation (5.66). The scattering angle probability, $P(\theta)$, has also been separated and is given by equation (5.67) below.

$$K_{II} = \left(\frac{Ze^2}{4\pi\kappa\varepsilon_0}\right)^2 N_I \frac{8\pi m_d k^*}{\hbar^3}$$
(5.66)

$$P(\theta) = \frac{\sin(\theta)}{\left(2k^{*2}\left(1 - \cos(\theta)\right) + k_c^2\right)^2}$$
(5.67)

Now defining the conditions that the numerical correction must satisfy in equations (5.68) and (5.69) which state that the momentum relaxation rate of the numerically corrected scattering model must match the BH model but the scattering rate cannot increase beyond Γ_{II}^{max} .

$$\frac{1}{\tau_m^{BH}} = \frac{1}{\tau_m^{new}}$$
(5.68)

$$\Gamma_{II}^{new} = \begin{cases} \Gamma_{II}^{BH} & \text{for } \Gamma_{II}^{BH} < \Gamma_{II}^{max} & (a) \\ \Gamma_{II}^{max} & \text{for } \Gamma_{II}^{BH} > \Gamma_{II}^{max} & (b) \end{cases}$$
(5.69)

Here Γ_{II}^{max} is the fixed cap which for reasons discussed later must be at least greater than $1/2\tau_m^{BH}$. The solution to this problem is to introduce a minimum scattering angle to cap the scattering rate, θ_{\min} , and to define a correction function, $G(\theta_{\min})$, to retain the momentum relaxation time.

$$\Gamma_{II}^{new} = K_{II}G(\theta_{\min})\int_{\theta_{\min}}^{\pi} P(\theta)d\theta$$
(5.70)

$$\frac{1}{\tau_m^{new}} = K_{II} G(\theta_{\min}) \int_{\theta_{\min}}^{\pi} P(\theta) (1 - \cos(\theta)) d\theta$$
(5.71)

The correction factor G can be expressed through use of the defining condition of equation (5.68) and after some basic manipulation yields

$$G(\theta_{\min}) = \frac{\int_{0}^{\pi} P(\theta) (1 - \cos(\theta)) d\theta}{\int_{\theta_{\min}}^{\pi} P(\theta) (1 - \cos(\theta)) d\theta}$$
(5.72)

It is clear that as $\theta_{\min} \rightarrow 0$ that this expression for G will tend to one which is the correct limit and ensures that the new scattering rate will adhere to the bounding condition (a) of equation (5.69). This leaves the minimum scattering angle, θ_{\min} , which due to the self-consistency of the problem must be found numerically by a root-finding technique. Using the condition (b) of equation (5.69), the minimum scattering angle can be found by solving the following equation.

$$\Gamma_{II}^{\max} - \Gamma_{II}^{new} = 0$$

$$\Gamma_{II}^{\max} - K_{II}G(\theta_{\min}) \int_{\theta_{\min}}^{\pi} P(\theta) d\theta = 0$$
(5.73)

This expression can be simplified by substituting in the expression for G given by equation (5.72) and dividing throughout by the original BH momentum relaxation rate.

$$\Gamma_{II}^{\max} \tau_m^{BH} - H(\theta_{\min}) = 0 \tag{5.74}$$

where the expression H is given by

$$H(\theta_{\min}) = \frac{\int_{\theta_{\min}}^{\pi} P(\theta) d\theta}{\int_{\theta_{\min}}^{\pi} P(\theta) (1 - \cos(\theta)) d\theta}$$
(5.75)

This modification to the root-finding method by redefining the problem into a scaled maximum scattering rate, $\Gamma_{II}^{\max} \tau_m^{BH}$, and the function $H(\theta_{\min})$ also solves a secondary problem. Examining the $G(\theta_{\min})$ equation of (5.72), it is clear that as $\theta_{\min} \to \pi$ this function will tend to infinity. By re-arrangement of the final problem, it is now possible to define this limitation in more detail. Taking the limit $\lim_{\theta_{\min}\to\pi} H(\theta_{\min})$ it is found that the function tends to $\frac{1}{2}$. Therefore as long as the maximum scattering rate is $\Gamma_{II}^{\max} \tau_m^{BH} \ge 1/2$ (using equation (5.74)) the numerically corrected II

scattering model will provide the exact BH model at a reduced scattering rate. This is a considerable reduction in the scattering rate as the momentum relaxation rate can be many orders of magnitude smaller than the scattering rate [61].

5.4.2 Numerically Capped Single/Double Interface Models

The numerically corrected approach to II scattering outlined in the previous section can be applied to the newly developed remotely screened impurity models. The approach to the problem remains identical for the new models but the scattering angle probability, equation (5.67), for the new scattering models is changed appropriately. For the single interface model the angular probability dependence is simply the θ -integral of the scattering rate.

$$P(\theta) = \frac{\sin\theta}{\left(\mathbf{q}^2 + k_c^2\right)^2} \left(1 + \exp\left(-2Z_I\sqrt{\mathbf{q}_\perp^2 + k_c^2}\right) - 2\cos\left(q_Z Z_I\right)\exp\left(-Z_I\sqrt{\mathbf{q}_\perp^2 + k_c^2}\right)\right) \quad (5.76)$$

The probability for double interface model can be obtained in the same manner.

$$P(\theta) = \frac{\sin\theta}{\left(\mathbf{q}^2 + k_c^2\right)^2} f_{dbl}\left(k_c, Z_l, L_c\right)$$
(5.77)

Here the f_{dbl} expression is given by the equations (5.50)-(5.53).

For both the remotely screening II scattering models the limit of the $H(\theta_{\min})$ function is identical to that of the BH model, that is $\lim_{\theta_{\min}\to\pi} H(\theta_{\min}) = \frac{1}{2}$. This can be understood from the behaviour of the coefficients of the remotely screened models which simply scale the behaviour of the II scattering model.

5.4.3 Testing Numerical Capped II Scattering

The numerical correction reduces the scattering rate by restricting the range of angles through which the carrier can scatter. Introduction of a correcting coefficient $G(\theta_{\min})$ allows the new model to retain the original momentum relaxation rate by a self-consistent method of finding θ_{\min} . As this method alters the dynamics of the carrier scattering process, it is important to ensure that the energy and velocity are being properly modelled. This can be tested by simulating the energyand velocity-field data between the BH and the new numerical II model. With typical data for the

5.4 Sub-Threshold Impurity Scattering

energy- and velocity-field relations given for undoped silicon, it is not possible to compare this model directly with experiment as II scattering is obviously negligible in such samples. Here to induce II scattering the silicon has a donor doping concentration of $N_D = 10^{18} cm^{-3}$.



Figure 5.29: (a) Energy-field and (b) velocity-field curves for bulk silicon at a donor concentration of $N_p = 10^{18} cm^{-3}$.

Figure 5.29 shows the energy-field relation and the velocity-field relation for the doped silicon. The new numerical corrected II model matches perfectly the BH model in both figures verifying that the energy and velocity of carriers is unaffected by numerically capping the scattering process.

Introduction of a new II scattering model may also affect the calibration of the simulator with other experimental data such as with the bulk and universal mobility. As the Monte Carlo simulator used in this work has been tested using Ridley's TBE model with the doping concentration dependent correction discussed in section 3.3.3 for impurity scattering, it is important to ensure that the new model doesn't negatively alter the calibration with experimental data.

The bulk or doping concentration dependent mobility has been tested for silicon with the original II model against the new numerical II approach in plot (a) of Figure 5.30. The doping-concentration dependent correction factor has not been modified for use with the new II model and remains identical to that used in the TBE approach discussed in section 3.3.3. Experimental data in this plot is from Thurber [29].



Figure 5.30: (a) Bulk mobility and (b) universal mobility for silicon at $N_A = 2 \times 10^{18} cm^{-3}$ against experimental data at 300K.

The new II scattering model is a slightly better fit with the experimental bulk mobility than the TBE model. Between the concentrations of $10^{15} - 10^{18} cm^{-3}$ the new model is a closer match to experimental data.

The above plots show that the numerical cap to the BH model doesn't alter the characteristics of the BH model in the energy/velocity plots whilst improving the calibration with experimental data over the TBE model with the bulk mobility. To demonstrate the validity of the new model in sub threshold conditions it is necessary to simulate a more advanced structure such as the MOS capacitor. Simulations of the MOS capacitor are used in Monte Carlo simulations to calibrate the universal mobility with experimental data [28] as discussed in chapter 3.5.2. In the device used here the substrate is doped to $N_A = 2 \times 10^{18} cm^{-3}$ with a low horizontal field of E = 0.5 kV / cm. The parameters for each of the scattering mechanisms are as detailed in chapter 3.3. In plot (b) of Figure 5.30 the universal mobility has been plotted showing the improved behaviour of the numerically capped BH model with the Ridley TBE model. Again, the new model is closer to experimental data than Ridley's TBE model and demonstrates the improved mobility of using the BH model.

5.5 Simulator Implementation

The implementation of the newly developed scattering processes into the numerical Monte Carlo simulation follows the typical method used for most scattering mechanisms [31, 75]. To briefly summarise, the maximum scattering rate of each mechanism is tabulated in the scattering table which is used to stochastically select a mechanism at each scattering event. Once a scattering mechanism is chosen, the appropriate scattering process is evaluated where the carrier is scattered.

5.5 Simulator Implementation

Within the scattering process, an out-scattering angle is stochastically selected for the carrier which, for elastic scattering, conserves the electron energy.

In this section a brief discussion of the details of the numerical method used will be given starting with the method for the calculation of the scattering rate in section 5.5.1. The scattering process and the method used to select the scattering angle is discussed in 5.5.2. Also discussed in section 5.5.3 is the numerical implementation of the double interface scattering model coefficient. Calculation of the coefficient during simulation can incur numerical floating point overflows under certain conditions. A solution has been found and this will be discussed in this final section.

5.5.1 Scattering Rate

The remotely-screened impurity scattering models have been developed as a correction to the existing BH scattering model. Remote screening has been shown to reduce the magnitude of II scattering and it is known that it will yield the BH model in the appropriate limit. Therefore in the numerical Monte Carlo simulation it is only necessary to tabulate the BH model for the scattering table as it will always be the upper limit on the scattering rate.

For the numerical II correction the remotely screened scattering and momentum relaxation rates must be evaluated which will require numerical integration of the θ integrals in equations (5.40), (5.41), (5.56) and (5.57). The Romberg integration method as discussed in Numerical Recipes for Fortran, section 4.3 [112] is used to evaluate the integral. It is also required to find the minimum scattering angle, θ_{min} , for the numerical II correction. This involves finding the root of equation (5.74) and must be found numerically. Referring to the methods in Numerical Recipes in Fortran [112], the Van Wijngaarden-Dekker-Brent method of section 9.3 has been employed. These numerical routines were chosen for both their simplicity in use and their efficiency as discussed in the Numerical Recipes book. The precision in the results is found to be sufficiently high and of course can be fine-tuned to suit the requirements.

5.5.2 Scattering Process

The scattering process for remotely screened impurity scattering is relatively typical for II scattering. There are some modifications to the standard II process which are outlined below, the most important of these is that of finding the carrier position. When this scattering event is chosen from the scattering table, the process must first determine the carrier position in relation to the source and drain interfaces. If the carrier is located in the source or drain the standard BH scattering process is invoked.

In the case that the carrier is located in the channel, there is still the possibility that it is located in a region that remote screening has no effect. To combat this, the remote screening correction is only invoked if the carrier is within four screening lengths of the source interface in the single interface model and within four screening lengths of the source or drain in the double interface model. The value of four is chosen such that the difference between the scattering rate of the remotely screened model and the BH model is within a few percent, e.g. $1 - \exp(-4) \approx 0.98$.

For a carrier which is suitable for scattering with the remote screening correction, an out-scattering angle must be selected. The out-scattering angle can be selected form the probability distribution function for each model given by the equations (5.76) and (5.77). There are several methods available to select the scattering angle using random numbers which are discussed in appendix of the review paper by Jacoboni and Reggiani [75]. The BH model makes use of the direct technique which allows the scattering angle to be chosen directly from a single random number. As the probability distribution functions for the remotely screened impurity scattering models cannot be integrated analytically, the direct technique is of no use. Instead a rejection technique is chosen which unfortunately is less efficient than the direct technique as it requires repeated evaluation of the probability distribution function. To increase the efficiency of the rejection method, the probability distribution function (PDF) for the scattering is scaled to values between 0 and 1. Thereby allowing the upper value constant for the rejection method to be selected as the PDF maximum and ensuring a minimum number of rejections. To obtain the scaling factor requires finding the maximum value of the probability function which must be found numerically. Using Brent's method of section 10.2 in Numerical Recipes in Fortran the maximum value of this function can be easily evaluated.

5.5.3 Double Interface Coefficient Calculation

The f_{dbl} coefficient given by equations (5.50)-(5.53) contains many exponential terms whose components can be very large. This leads to numerical overflows during the numerical simulation. The solution to this problem is to use the approximation $\exp(x) - 1 \approx \exp(x)$ for large values of x. Within the IEEE double-precision format for floating-point numbers which is used in the numerical simulations in this work, the numerical precision for floating point numbers is $2^{\pm 53}$ with a range of $10^{\pm 308}$. The exponential function will overflow in this system for a power of roughly $x = \pm 308 \times \ln 10 = \pm 709.196$.

Rather than employing an approximation to the coefficient to all values of x or at values x > 709, it is best to introduce an approximation at the numerical precision. In other words, utilise an
approximation on the coefficient when the power is at the limit of numerical precision rather than the limit on range, therefore for $x = \pm 53 \times \ln 2 = \pm 36.7368$. Using this value with the IEEE doubleprecision system, the approximation that $\exp(x) - 1 \approx \exp(x)$ can be used and becomes exact when employed above powers of |x| > 36.7368.

The f_{dbl} coefficient given by equations (5.50)-(5.53) contains three different $\exp(x)-1$ terms which can be simplified using the approximation discussed above. These three terms are

$$\left(\exp\left(2L_{c}K_{c}\right)-1\right) \tag{5.78}$$

$$\left(\exp\left(2(L_c - Z_I)K_c\right) - 1\right) \tag{5.79}$$

$$\left(1 - \exp\left(2Z_I K_C\right)\right) \tag{5.80}$$

To attain the maximum accuracy it is important to approximate only the necessary terms to avoid overflow. It follows that with three conditions that there are $2^3 = 8$ cases which can easily be tested in the final program code and the suitably simplified expression for f_{dbl} chosen. As an example, the case where all three terms can be approximated is given here. Using the simplifying approximation on the equations (5.78)-(5.80), substitution back into the coefficient and rearranging the terms yields

$$f_{dbl} = 1 + \exp(-2Z_{I}K_{C}) - 2\cos(q_{Z}Z_{I})\exp(-Z_{I}K_{C}) + 2\cos(q_{Z}L_{C})\exp(-L_{C}K_{C}) -2\cos(q_{Z}(Z_{I}-L_{C}))\exp((Z_{I}-L_{C})K_{C}) + \exp(2(Z_{I}-L_{C})K_{C})$$
(5.81)

This term is greatly simplified from the complete form and interestingly has some similarity with the single interface coefficient, the terms within the square brackets of equation (5.40). The first three terms of the above expression are identical to the single interface coefficient, which can be found exactly by allowing $L_c \rightarrow \infty$.

5.6 Conclusion

In this chapter a new scattering rate, entitled remotely screened impurity scattering, has been developed for numerical device simulation within the Monte Carlo technique. This newly developed scattering rate introduces polarisation charge effects induced from the highly doped source and drain regions on impurity scattering in the channel. Remotely screened impurity

5.6 Conclusion

scattering has been developed in two different forms, given as either the single or double interface case referring to the inclusion of one or two highly doped regions respectively. The single interface case allows the inclusion of a single polarisation charge on the screening of an impurity and the double interface case allowing multiple polarisation charges to be induced between the source and drain regions.

Remote screening of IIs has been shown to increase the II-limited mobility considerably when the II is located close to the highly-doped source and/or drain regions. The strength of remote screening becomes very strong when the impurity is located within one channel screening length of the highly-doped regions. In the context of real device channel lengths, the lower the impurity density in the channel leading to larger channel screening lengths, the stronger the effect of remote screening. In short channel length devices, taking into account remote screening from both the source and the drain, the II-limited mobility can be increased by up 500%.

Due to the anisotropic nature of the scattering potential and the context of this work, a simplified approach has been utilised within the scattering model to reduce the complexity whilst allowing an analysis of the effect of remote screening. This simplification, here entitled the Z-aligned model, assumes that the incoming carrier is aligned with the Z-axis of the scattering reference frame thereby removing the anisotropy. Essentially the Z-aligned model assumes that the scattering potential is isotropic such that it can be used with the typical formulations of Monte Carlo scattering approach. The simplification was shown here using an analysis of incoming carrier angle to lead to an average (negligible) error of at most 2%. The average being taken over the domain where remote screening is effective and assuming that the incoming carrier is at the worst case alignment for anisotropy.

The developed scattering mechanism has been developed to fully replace the existing II scattering mechanism within the channel region of a MOSFET and although possible, for computational efficiency a transition has been introduced to merge with the classic BH approach. This transition is based on the roll-off of the polarisation charges at increasing distance from the source and drain interfaces. This transitional distance corresponds to roughly four channel screening lengths from the interface where the effect of polarisation charges on the screening is considered to be very small. Beyond this transitional distance, the II scattering model will revert to the classic BH approach, improving computational efficiency by removing the need to evaluate the complex remote screening expressions.

The existing approach to II scattering in the MC simulator used within this work made use of Ridley's TBE approach which has been shown to be non-ideal due to the removal of low screening density II scattering. The TBE approach is particularly bad when modelling low effective field

5.6 Conclusion

universal mobility where the screening density is much lower than the impurity density. To resolve this problem and allow a complete analysis of the effect of remote screening on II scattering, a new approach has been developed which allows low screening density II scattering to be modelled.

This new approach makes use of a numerical cap to II scattering which has been introduced to solve the problem of diverging scattering rates with very low screening densities. This problem exists with BH scattering and the approach used here for the remote screening model. Through restriction of the minimum scattering angle and use of a self-consisten correction function, the scattering rate can be numerically capped whilst still maintaining the momentum relaxation rate. Thereby all II scattering can be modelled efficiently and with scattering rates which are within reason for MC simulation.

Chapter 6 Results and Discussion

6.1 Introduction

The objective of this PhD research is to examine the remote screening effect that the highly doped source and drain regions have on channel ionized impurity (II) scattering. In Chapter 5 a scattering model suitable for Monte Carlo (MC) simulation has been developed which includes the remote screening effect. In this chapter this newly developed model has been applied to simulations of two n-type MOSFET devices.

Remote screening of channel dopants by the source and drain regions has been shown in the previous chapter to reduce the effect of II scattering. This reduction in II scattering is evident close to the source and drain interfaces and is considered negligible at a distance greater than four channel screening lengths from an interface. As the remote screening effect is heavily dependent on channel screening length, where larger screening lengths increase the strength of the remote screening, this effect will be more dominant at lower gate voltages. Lower gate voltages of course lead to lower screening densities in the channel region.

Hence, the remote screening of II scattering is expected to increase carrier transport performance through the channel region particularly near the source and drain interfaces. This increase in performance is expected to be greater at lower gate voltages and will be reduced as the gate voltage and channel screening concentration is increased. This increase in channel transport performance can also be described as reduction in the control that doping has on the channel. At low gate voltages II scattering helps to turn the device off by reducing the channel mobility. With the introduction of remote screening to the II scattering mechanism, the low gate voltage channel mobility may be increased leading to larger off-state currents.

The in-house MC simulator which was discussed in Chapter 3 is utilised for the device simulations. The scattering parameters and models remain identical to that of the calibrated simulator in chapter 3.3 with the exception of the II scattering model. For these device simulations the numerically capped Brooks-Herring (BH) model presented in section 5.4 is employed in place of the Third-Body Exclusion (TBE) II scattering model for the reasons discussed in the previous chapter.

A self-consistent MC simulation is completed for each of the two devices as discussed in Chapter 3. The simulation parameters for the two MC device simulations completed here are kept constant between the simulations with a timestep of $\Delta t = 10^{-17} s$ and a total simulation period of 10 ps (at

6.1 Introduction

low gate voltages the simulation period is increased up to 20ps to reduce the statistical error). A transient period of 2ps is allowed before statistics are gathered and the non-linear Poisson solution is solved every 250 timesteps or 2.5 fs. A total of 95000 particles are utilised in each MC simulation to represent the charge carriers.

The first of the devices to be simulated with the remotely screened impurity scattering model is a bulk MOSFET device. This bulk device was first developed and published by Toshiba in 2001 [113, 114] and is designed as a high-performance device for sub-50nm CMOS applications. This bulk MOSFET device is highly doped and has an advanced doping profile including a super-steep retrograde (SSR) doping profile and halo implants. This complex channel doping profile is implemented to reduce the short-channel effects that occur at this scale such as threshold voltage roll-off and punch-through. It has been used here to represent a typical current generation bulk MOSFET device which can be scaled successfully for use in future generations [115]. The industry roadmap, the International Technology Roadmap for Semiconductors (ITRS) 2008 update [1], predicts the bulk device structure to be continued in production until at least 2012.

The second device to be simulated is a device that is proposed for future technology generations. This device is a Ultra-Thin Body Double-Gate (UTB DG) MOSFET device which has been developed as part of the PullNano European project as a template device. The PullNano project used this template device to compare the a wide variety of different device simulation techniques used within the PullNano European consortium [116]. This device has been designed for a future low standby power technology (LSTP) generation and is constructed with a gate length of 22nm. The DG structure is currently predicted by the ITRS as the "ultimate MOSFET device" that can be scaled to the end of the roadmap [1].

The UTB DG device has highly doped source and drain region and unlike the bulk MOSFET, has a low doped channel which will increase channel screening lengths. Originally developed with a high- κ dielectric which has been replaced with the equivalent oxide thickness (EOT) in traditional silicon dioxide for the purposes of this simulation study. This allows the additional complex scattering mechanisms that need to be introduced with a high- κ dielectric to be neglected.

In these devices the effect of remote screening from the highly doped source and drain regions is considered but the remote screening from the highly doped or metal gate(s) has been left to research beyond the thesis. Within this work, the analysis of remote screening from the source and drain has been considered to be the more dominant component and has been studied first. Carriers travel through the source and drain remote screening regions to contribute to conduction and therefore are far more likely to feel the effect of remote screening. The effect of gate remote

6.2 35nm Bulk Device

screening is still considered an important interaction and will be studied in the future. The gate remote screening effect will become stronger in future device such as the double gate device where gate oxides are become very thin and metal gates are commonplace.

The structure of the chapter will begin with the bulk MOSFET in section 6.2 followed by the UTB DG device in section 6.3 and finishing with the conclusions of the simulations in section 6.4. For the device simulation sections 6.2 & 6.3, the internal structure is repeated. The device simulation sections begin with an initial presentation of the device structure and calibration, followed by a brief study of the expected effect of remote screening. Next within the section, the numerical results of the MC simulations for the particular device are discussed in detail and finally the results are summarised.

6.2 35nm Bulk Device

The 35nm bulk MOSFET device used here has been developed to replicate a published device structure from Toshiba [113, 114]. This device has been chosen as it represents a realistic device that is used within the industry in CMOS applications. The device structure has been reconstructed from published experimental data within the commercial TCAD software Sentaurus [117], and was completed as part of the PhD thesis of Fikru Ademu-Lema, a researcher from this group [118]. Full details of the calibration of the commercial tools along with the extraction of the device structure, doping profiles and characteristics are provided in [118]. The device has since been applied within the research group in Drift-Diffusion (DD) simulation. This required the device structure and doping profiles to exported from Sentaurus for application with the in-house DD code. Another researcher from this group, Gareth Roy, completed this process and calibrated the DD simulator with Sentaurus as part of his PhD thesis [119].

Thankfully due to the construction of the Monte Carlo (MC) simulator used within this work, the device structure can be directly exported from the DD simulator and applied without difficulty to the MC simulator. As MC calibration is completed through non-device specific experimental data as discussed in Chapter 3, no further calibration is required.

This section will begin by presenting the device structure and doping profile for this bulk device along with the calibration data showing the match between the simulators in section 6.2.1. Following this in section 6.2.2 will be a brief analysis of the expected behaviour from the introduction of remote screening to ionized impurity scattering in simulation of this device. Finally the results of the MC simulation with remote screening will be analysed in detail within section 6.2.3.

6.2.1 Device Structure and Calibration

The structure and dimensions of the bulk device used with the MC simulator are given in Figure 6.1. It should be noted that this diagram is not drawn to scale but provides a generic overview of the structure. The remote screening plane is given in the upper-left corner of Figure 6.1 and the positions of the remote screening interfaces at Z=0 and Z= L_c are marked at the edges of the source/drain regions.



Figure 6.1: Structure of the bulk MOSFET device.

The printed gate length of 35nm defines the device scale and has a metallurgical channel length, measured from the net doping profile, of approximately 26nm. This device has been designed for the 45nm technology node and MOSFET's of this scale are already in production. The oxide thickness for this device is 1.4nm and is a silicon oxi-nitride dielectric with a dielectric constant of $\varepsilon_r = 5.45$. The gate in the MOSFET device of Figure 6.1 used within the DD and MC simulators, is a metal gate with a work-function of $\phi_G = 3.945 eV$. This deviates from the original specification of a poly-Si gate as discussed in the original paper [113, 114] and is shown below to have little effect on the calibration.

The net doping profile is given in Figure 6.2 for this bulk device which, as mentioned earlier, was produced in the commercial TCAD process tool Sentaurus [117] by Fikru Ademu-Lema [118]. Use of the commercial process tool has allowed the advanced doping profile to be accurately replicated from the published data of the Toshiba device.



Figure 6.2: Density plot of bulk MOSFET net doping with the positive scale denoting donor doping and the negative acceptor doping.

In Figure 6.2 the advanced doping profile in the substrate can be clearly seen with n-type doping in red and p-type doping in blue. The device has an indium doped retrograde channel profile with the peak density just below the source/drain wells. This is coupled with boron Halo extensions in the channel to improve the punch through characteristics without heavily increasing the channel doping concentration. The source and drain wells are arsenic doped, with shallow extensions to reduce short-channel effects and deep junctions to reduce access resistance.

This channel doping profile has been directly imported into the DD simulation (and MC simulation) from Sentaurus and the technique is discussed within the PhD thesis of Gareth Roy [119]. It should be noted that this net doping profile figure, which is exported from Sentaurus, includes a poly-Si gate which has been neglected within the DD and MC simulations. Calibration of the commercial TCAD software with the in-house DD simulator is completed and discussed in detail within Gareth Roy's thesis and here the results are simply repeated. Calibration is completed through the I_D-V_G curves and is shown in Figure 6.3.



Figure 6.3: (a) Linear and (b) semi-log I_D -V_G curves for the bulk MOSFET device showing the comparison between Sentaurus, Drift-Diffusion and Monte Carlo simulation.

In plot (a) we see the excellent match between DD and the commercial Sentaurus software at both the low drain voltage, $V_D = 50mV$, and the high drain voltage of $V_D = 1V$. For reference the DD simulation has been completed with the original poly-Si gate and the metal gate to demonstrate the negligible difference between the results. Also shown is the comparison between the MC and DD simulation of the metal gate device at a low drain bias of $V_D = 0.1V$ and a high drain bias of $V_D = 1V$. As is expected, the MC simulation is very close to the DD results at low drain bias and shows an increase in the drain current at high drain bias. This increase is due to the nonequilibrium transport of carriers within the MC simulation.

As MC simulation is calibrated to other more generic experimental data such as energy-/velocityfield curves, bulk mobility and universal mobility, further calibration is not required. Although it is important to ensure that the MC results are similar to the DD simulation, it is indicative to check that the curves match in low-field conditions. This is easier to show in a semi-log plot of the I_D-V_G curves given in plot (b) of Figure 6.3. In plot (b) the match of the MC with the DD solution is clear at the low-field conditions given for $V_D = 0.1V$. The solutions between DD and MC also match well for low gate voltages at the high drain bias where the non-linear transport effects such as velocity overshoot are not important.

In both these plots for the $I_D - V_G$ curves, the range of gate voltages was restricted to a minimum of $V_G = 0.4V$. This is the case as MC is unable to accurately model the device for lower gate voltages as the statistical noise in the data overcomes the results.

An important step in the use of the remote screening model for ionized impurity scattering is the position of the source and drain interfaces. In a device with a realistic doping profile this can

become a complex decision as abrupt transitions in doping are not found. Instead a simple estimate has been made based on the net doping profile.



Figure 6.4: (a) Net doping profile with signed log scale and (b) net doping difference plot for the bulk MOSFET.

Plot (b) of Figure 6.4 demonstrates a difference plot of the net doping profile (given by the solid red line). In the channel of the bulk MOSFET the point at which the net doping is at the steepest gradient is chosen at the point that the source and drain interfaces are to be located. In Figure 6.4 the positions chosen for the source and drain interfaces are depicted by the dashed green lines, where the source interface is assigned at Z = 47nm and the drain interface at Z = 73nm.

Plot (a) of Figure 6.4 also demonstrates a signed log plot of the net doping profile which highlights that the point at which the net doping is at the steepest gradient coincides with the metallurgical p-n junction. This strengthens the chosen positions of the source and drain interfaces as being closest to the original definition used in the scattering model calculation of abrupt interfaces.

6.2.2 Analytical Estimate of Remote Screening Effect

Before moving to MC simulations of the bulk MOSFET device it is important to examine what effect remotely screened ionized impurity scattering is expected to have on the device performance. A simple estimate is possible by using analytical calculation of the mobility in conjunction with data from DD simulation of the device. The DD data used within this mobility estimate is the same data used to initialise the MC simulation.

The mobility analysis completed here is not a typical mobility analysis, which normally might be completed to characterise uniformly doped silicon under various electric fields. Here the analysis of mobility is used to estimate the effect that the remotely screened II scattering mechanism has in the context of channel performance. In a channel of the device the mobility is spatially dependent on various factors such as doping density, carrier concentration and the electric field. Such spatial dependence makes definition of the mobility in a MOSFET channel extremely challenging. Here it must be stressed that this analysis is a simple, rough estimate for the channel mobility to allow greater insight into the behaviour of remote screening in a device context.

To compute the spatially dependent mobility, a 1D profile of the channel impurity and electron concentration data at each mesh point is obtained from the DD simulation data. The Kubo-Greenwood formula [33], given by equation (5.34), allows the mobility to be calculated at each mesh point along this channel profile for the scattering mechanisms considered. The total mobility can then be approximated through use of Mathiessen's rule [30], equation (6.1),

$$\frac{1}{\mu_{tot}} = \sum_{i} \frac{1}{\mu_{i}} \tag{6.1}$$

with μ_i denoting the *i*-th mobility mechanism. As mentioned above, the total mobility can be calculated at each mesh point along the channel. In this particular device the mesh points are evenly spaced at 0.5nm intervals.

Referring to the textbook definition of the mobility, it is defined as the proportionality of the carrier velocity gained by carriers in between scattering events to the electric field [15]. The average time between collisions is also known as the mean free time and can be expressed as the mean free path, $l = \tau v_{th}$, with the inclusion of the average thermal velocity of electrons. The use of the thermal velocity can only be considered a very rough estimation which although only applies at equilibrium has been used in literature to define an estimated mean free path (see [120]). Therefore, the mobility can be said to be a value defining the relation between carrier velocity and the electric field over a mean free path.

Taking the average thermal velocity as $v_{th} = (8k_BT/\pi m_C)^{\frac{1}{2}} = 0.9907 \times 10^7 \text{ cm/s}$ [51] and assuming the mesh spacing is the mean free path, the mean free time can be approximated as $\tau = 0.5 \times 10^{-9}/0.9907 \times 10^5 \approx 5 \times 10^{-15} \text{ s}$. This mean free time is an order of magnitude smaller than has been reported in the literature [33] and highlights the problem that calculating the mobility on the mesh spacing interval will lead to overestimation. To combat this problem a moving median of the total mobility is taken with a span calculated from an estimate of the mean free path. A simple estimate for the mean free path can be obtained using the average ensemble momentum relaxation time, defined as equation (6.2) [121], for the mean free time between scattering events.

$$\left\langle \tau_{m}^{tot} \right\rangle = \frac{\int_{0}^{\infty} dE \tau_{m}^{tot}(E) E^{\partial f_{0}} / \partial E}{\int_{0}^{\infty} dE E^{\partial f_{0}} / \partial E}$$
(6.2)

where f_0 is the equilibrium Fermi distribution and τ_m^{tot} is given as

$$\frac{1}{\tau_m^{tot}(E)} = \sum_i \frac{1}{\tau_m^i(E)}$$
(6.3)

with τ_m^i denoting the *i*-th scattering mechanism.

Considering that the Kubo-Greenwood formula is valid for linear, low-field systems [122] and Mathiessen's rule is valid for independent scattering mechanisms [30], the use of this approach in bulk MOSFETs can only be regarded as an estimate [71]. Despite all this, the purpose of this analysis is to evaluate the remote screening effect on channel performance and the methodology discussed here is considered a reasonable approach.

The scattering mechanisms included in this mobility calculation include the acoustic and optical phonons along with II scattering. Three cases are presented, each case has a different II scattering model which will be either the BH model discussed in section 2.3, the single interface remotely screened II model from section 5.2.3.1, or the double interface remotely screened II model given in section 5.3.3.1. In all cases ellipsoidal, non-parabolic bands are assumed and for the purposes of a simple evaluation, only the first silicon band is included (the X valleys).

For the intra-valley acoustic phonon model, the approach discussed in section 3.3.1 is used here. The intervalley optical phonon scattering mechanisms presented in section 3.3.2 are also used here. This includes the scattering between equivalent X-valleys and the f- and g-type intervalley optical phonons. The material constants and transport parameters used in the phonon scattering calculations are those listed in Table 2 of Chapter 3.

As mentioned earlier the DD data is used to initialise the MC simulation and for the mobility calculations presented here, the impurity and electron density is all that is required. For this analysis, only the low drain solutions at three separate gate voltages will be analysed. At each gate voltage, the average ensemble momentum relaxation rate is calculated and the mean free path is estimated to obtain the span of the moving median.

Presented in Figure 6.5 is (a) the carrier/impurity concentrations and (b) the calculated mobility for the different ionized impurity models with phonon scattering for $V_G = 0V$. The average ensemble momentum relaxation rate is calculated using equation (6.2) in conjunction with the sum of all momentum relaxing processes, equation (6.3), which for all the phonons and the BH model yields $\langle \tau_m^{tot} \rangle = 1.56 \times 10^{-14} \, s$. Hence the estimated mean free path is given as $l_{\tau_m} = 1.55 nm$.

For this the lowest gate voltage of $V_G = 0V$, the carrier concentration is very much lower than the impurity density in the channel. These are conditions where II scattering typically dominates [27] and will lead to a high II scattering rate.



Figure 6.5: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 0V$ and a drain voltage of $V_D = 0.1V$.

Plot (b) of Figure 6.5 shows a low mobility along the channel length consistent with the large II scattering for the BH case. It is clear in this plot that the remotely screened II models cause a large increase in the mobility over the entire length of the channel. At the source end of the channel both the single and double interface models corresponds to an increase of 196%, with an increase of 117% at the drain end of the channel for the double interface model.

The large increase in the channel mobility with the remotely screened models can be understood from the low carrier (screening) density in the channel as shown by the red line in plot (a) of Figure 6.5. The longer the screening length in the channel, the stronger the remote screening effect which reduces the momentum relaxing effect of II scattering.



Figure 6.6: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 0.4V$ and a drain voltage of $V_D = 0.1V$.

At a higher gate voltage of $V_G = 0.4V$, the average ensemble momentum relaxation rate is calculated as $\langle \tau_m^{tot} \rangle = 3.27 \times 10^{-14} s$ which gives a estimated mean free path of $l_{\tau_m} = 3.24 nm$. The electron concentration at this gate voltage, shown in plot (a) of Figure 6.6, is almost an order of magnitude larger than the impurity concentration. This will decrease the strength of the II scattering and the mobility will increase, clear from plot (b) which is approximately four times larger than that of $V_G = 0V$.

The increase in screening density at this higher gate voltage has the effect of reducing the strength of remote screening on II scattering. The effect on the channel mobility due to the remote screening models is consistently smaller, on the order of 20% at the source and 7-8% at the drain. There is still an increase in the mid channel mobility but it is reduced to \sim 5% improvement.



Figure 6.7: Plot of (a) the impurity and electron density, and (b) the impurity and phonon limited mobility at a gate voltage of $V_G = 1V$ and a drain voltage of $V_D = 0.1V$.

For the highest gate voltage of $V_G = 1V$ shown in Figure 6.7, the average momentum relaxation rate is $\langle \tau_m^{tot} \rangle = 5.59 \times 10^{-14} s$ and the estimated mean free path is $l_{\tau_m} = 5.54 nm$. At such a high gate

voltage the mobility here is dominated by phonon scattering as the electron concentration rises to well above the impurity concentration. The characteristic increase in mobility around the source and drain interface is still present although much less dramatic than at lower gate voltages. At the source end of the channel there is a peak increase due to remote screening of 10%, a consistent increase of 2% along the channel and a peak increase at the drain of 4%. This reduction in the effect of remote screening is expected at high gate voltages where the screening density is greatest.

From this analysis, the effect of remote screening on the device mobility will be larger at lower gate voltages where the carrier concentration is near to or lower than the impurity concentration. It predominantly has the largest effect in a region next to the source and/or drain interfaces, with the increase at the source interface less affected by the drain bias. As the gate voltage increases, the screening density increases and the effectiveness of remotely screened II scattering is reduced.

6.2.3 I_D-V_G Behaviour with Remote Screening

Implementing the remotely-screened impurity scattering rates developed in Chapter 5 into the MC simulator will allow the effect on device performance to be thoroughly analysed. MC simulations have been completed for the bulk MOSFET device using the newly developed, numerically capped methodology for II scattering, discussed in section 5.4.

The analytical analysis in the previous section suggests that the remote screening effect should increase the channel mobility for lower gate voltages. Ideally this increase in mobility should lead to an increase in the drain current at lower gate voltages with a decreasing effect as the gate voltage is increased. Given this evaluation of the remote screening effect, the bulk MOSFET device has been simulated for various voltage points to produce an $I_D - V_G$ curve. Such a curve can be used to assess the sub-threshold leakage current, the threshold voltage itself and the drive current. As mentioned previously, statistical noise means that simulations must have a gate voltage of $V_G \ge 0.4V$.

In Figure 6.8 the simulation data is plotted for four different cases of channel II scattering. The cases are: the numerically capped BH (Num BH) model throughout the device; no II scattering in the channel region with Num BH in the source/drain (No Chn II); single interface Remotely Screened Brooks Herring scattering (RSBH Sgl) in the channel with Num BH in source/drain; double interface remotely screened BH scattering (RSBH Dbl) in the channel and Num BH in source/drain.



Figure 6.8: (a) Linear and (b) semi-log I_D -V_G curves for the bulk MOSFET with remotely screened impurity scattering.

The Num BH case is the baseline case with the *generic* II scattering model and is denoted by the solid black line in the two figures. As an upper limit on the effect of II scattering in the channel, simulations have been completed where II scattering has been neglected for the channel region and is shown in the figures as the red line.

The effect of neglecting the channel II scattering is clear, with a noticeable increase in the drain current. The increase at low drain bias is an almost constant increase at each gate voltage point, roughly $30\mu A / \mu m$. The increase in current at high drain bias increases slightly with gate voltage, roughly $40\mu A / \mu m$ at low gate voltage to $70\mu A / \mu m$ at high gate. Therefore in the bulk device the channel impurity scattering plays a role in characterising the device current drive and sub-threshold leakage.

The remotely screened models are also included in the figures for the single interface (green line) and the double interface (blue line) cases. The remote screening models clearly have negligible effect at any gate voltage. The small increases at low gate voltage of the low drain bias curve are within the error bars on the plot and therefore are not conclusive. In the following sub-sections the results will be analysed in more detail for the four different II cases discussed above. Initially, results of the low drain case of $V_D = 0.1V$ will be analysed followed by the high drain case of $V_D = 1V$.

6.2.3.1 Low Drain Bias

The results for $V_D = 0.1V$ will be examined in greater detail for two gate voltage points, one at a low gate voltage and one at the highest gate voltage of $V_G = 1V$. Given that MC simulation is prone to statistical noise at low gate voltages it is important to examine a low gate voltage data

point where this noise is minimal. Examining the error bars at low gate voltage points of $V_G = 0.4V, 0.5V, 0.6V$ in more detail in Figure 6.9 will allow the data point with minimal noise to be chosen.



Figure 6.9: I_D -V_G characteristics for the bulk MOSFET with error bars for the low gate voltage points at $V_D = 0.1V$.

The error bars in the MC simulator are calculated from the standard deviation in the mean of the variable, the drain current in this case, where the statistical dependence has been included through an estimation of the correlation time [123]. The error is calculated using the following equation [124]

$$err = \sqrt{\frac{\sigma^2}{N} \left(1 + 2\sum_{k=1}^{N-k} \left(1 - \frac{k}{N} \right) \rho_k \right)}$$
(6.4)

where σ is the standard deviation (σ^2 the variance), *N* is the number of steps in the simulation and ρ_k is the estimated *k* -th- lag autocorrelation (which is dimensionless). The sum represents an estimation of the autocorrelation function. The *k*-th lag autocorrelation is estimated using the sample autocorrelation, $\rho_k = \gamma_k / \gamma_0$ where the autocovariance, γ_k , can be estimated by (γ_0 is the *k*=0 lag autocovariance, essentially the sample variance, σ^2)

$$\gamma_k = \frac{1}{N} \sum_{t=1}^{N-k} \left(z_t - \overline{z} \right) \left(z_{t+k} - \overline{z} \right)$$
(6.5)

Here z_t represents the variable at a time step t and \overline{z} represents the sample mean over the entire time series. This method is discussed in more detail within the textbook by Box [124]. It should be noted that the error bars plotted are a 1σ case representing a confidence interval (assuming a normal distribution of error) of roughly 68%.

With a more detailed examination on the low gate voltage points in Figure 6.9, the remotely screened II scattering model appears to show a slight increase at $V_G = 0.4V, 0.5V$ and oddly a slight decrease at $V_G = 0.6V$. At a gate voltage of 0.4V the statistical noise, given by the error bars, make this result difficult to examine and validate. At the gate voltage point of 0.5V the remotely screened models are a close match, and slightly larger than the Num BH model. The error bars at this data point are also small enough to be considered not to dominate the results. It is this data point of $V_G = 0.5V$ which has been chosen to analyse the low gate voltage behaviour of the II models.

It is important to ensure that the remotely screened II models are operating correctly and that their behaviour is as expected. This is best shown by plotting a profile of the II scattering tally taken at each mesh point. That is, for each mesh square, the number of II scattering events was counted for each of the II scattering cases. This II scattering tally has been plotted in Figure 6.10 with the II scattering tally data given in arbitrary units.



Figure 6.10: lonized impurity scattering tally along the channel for the bulk MOSFET at (a) $V_G = 0.5V$ and (b) $V_G = 1V$ with $V_D = 0.1V$.

The plots in Figure 6.10 show that there is a finite number of II scattering events in the channel region which, away from the source and drain interfaces, is similar between the remotely screened and BH II models. For the case of no channel II scattering, the tally drops off abruptly at the source and drain interfaces as is expected. Also expected is the drop off in scattering events for the single and double interface RSBH model next to the source and drain interfaces respectively. As

6.2 35nm Bulk Device

discussed in Chapter 4, the remotely screened II models reduces the II scattering rate located close to the interfaces which explains the reduced number of events seen above.

It is also interesting to note that at the higher gate voltage, plot (b) in Figure 6.10, the II scattering tally is larger in the channel than the lower gate voltage case, plot (a), for all channel II scattering models. This is contrary to the fact that at lower screening densities the II scattering rate will increase. This can be understood from the increased carrier density at the high gate voltage increasing the number of scattering carriers in the channel and consequently, the number of events.

Having established that the scattering models are operating correctly, it is now essential to examine what effect each model has on the carrier transport. This is best completed by examining the carrier velocity and density along the channel. Figure 6.11 presents (a) the electron velocity and (b) density for the gate voltage of 0.5V and Figure 6.12 the (a) velocity and (b) density for a gate voltage of 1V.



Figure 6.11: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 0.5V$ and $V_D = 0.1V$.

For $V_G = 0.5V$ the electron velocity along the channel in plot (a) of Figure 6.11 increases within the source region and is rapidly accelerating as it crosses the source interface at 47nm. At this point, the model which neglects channel II scattering (No Chn II, the red line) has an increased velocity which remains larger throughout the channel length and into the drain region at 73nm. The standard and remotely screened II scattering cases remain fairly close to each other throughout the channel region. There is a slight increase in carrier velocity mid channel, around the 55-60nm region, with remote screening but this is very much within the region suffering from statistical noise. Plot (b) of Figure 6.11 gives the corresponding electron density along the channel. The electron density is identical between all four cases of channel II scattering in the MC simulation. Hence, it can be said that the larger drain current of the No Chn II case is due simply to the increased electron velocity through the channel. It also explains the mild increase of the remote screening models drain current as the electron velocity in plot (a) of Figure 6.11 is very slightly higher over the standard case. This result is partially obscured by the noise in the data which, if we look at the electron density, can be understood. Given the channel electron density of roughly $n = 1 \times 10^{19} cm^{-3}$ of the multiplying this a rough estimate channel and by volume, $26nm \times 10nm \times 1nm = 2.6 \times 10^{-19} cm^3$, corresponds to around 2-3 electrons in the channel volume. With so few electrons in the channel contributing to the current density, noise in the results is unavoidable.

The electron velocity and density for the highest gate voltage of $V_G = 1V$ in Figure 6.12 demonstrates similar results for the different II scattering cases. Here the electron velocity in the channel for the case without II scattering is higher than the standard case, but is not as significant an increase as the lower gate voltage case. In fact, the peak velocity for this higher gate voltage is lower than the velocities shown in Figure 6.11 for the low gate voltage point.



Figure 6.12: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 1V$ and $V_D = 0.1V$.

The relative decrease in velocity can be explained by the electron density given by plot (b) of Figure 6.12. At this higher gate voltage the electron density is around $n_e = 5 \times 10^{19} cm^{-3}$ which is approximately five times the number of electrons present in the channel region. Referring to the simple expression for current density, which can be written

$$J = e n_e v_d \tag{6.6}$$

where *e* is the electronic charge, n_e the carrier density and v_d the drift velocity. Hence, given that the drain current at $V_G = 1V$ is roughly 2.5 times larger and the electron density is 5 times larger than the $V_G = 0.5V$ case, the velocity at high drain need only be half the magnitude of the low drain results. Examining the velocity of the low drain case given in Figure 6.11, the peak is roughly $v_d = 0.6 \times 10^7 cm/s$ and for the high drain case the peak is roughly $v_d = 0.3 \times 10^7 cm/s$ which coincides with this basic theory.

The energy profiles for the carriers are quite similar between the different channel II scattering cases. As II scattering is an elastic process, the carrier energies in the channel will not be directly modified by changes in this model. Although with the reduction in II scattering, other scattering processes could become more prominent such as inelastic phonon scattering which potentially could modify the carrier energies, this has not been the case here.



Figure 6.13: Energy profile along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering at (a) $V_G = 0.5V$ and $V_D = 0.1V$ and (b) $V_G = 1V$ and $V_D = 0.1V$.

At low drain bias the remotely screened II scattering model has little effect on the drain current. Considering the analytical mobility analysis completed in section 6.2.2, the remotely screened models was expected to increase the channel performance, particularly at the source interface. Therefore examination must move to the final scattering mechanism, interface roughness (IR) scattering which was not considered in the analytical mobility analysis. IR scattering is found to have a strong performance defining factor in the bulk MOSFET device with the removal of this scattering mechanism from the device leads to a 52% increase in drain current at $V_G = 0.5V$, and a 88% increase at $V_G = 1V$.

An estimate to the IR scattering mobility can be evaluated using the IR scattering model and estimating the effective vertical electric field, E_{eff} , in the channel of the device from the MC simulation results. Taking an estimate of the effective field in the middle of the channel, Z = 60nm, for both gate voltages gives $E_{eff} \approx 1.09 MV/cm$ and $E_{eff} \approx 1.68 MV/cm$ respectively. Using these values in the IR momentum relaxation rate, equation (6.7) below, it is possible to compute the corresponding mobility with the Kubo-Greenwood formula, equation (5.34).

$$\frac{1}{\tau_m^{IR}} = \frac{e^2 m_d}{2\hbar^3} \left(E_{eff} \Delta_{rms} L_e \right)^2 \left(1 + 2\alpha E \right) \int_0^{2\pi} d\varphi \frac{1 - \cos\varphi}{\left(1 + \frac{1}{2} L_e^2 q^2 \right)^{\frac{3}{2}}}$$
(6.7)

here $\Delta_{rms} = 0.35nm$ is the RMS amplitude of surface fluctuations and $L_e = 1.3nm$ is the correlation length. This IR scattering model is the Ando model with an exponential autocorrelation function [58] as discussed in section 3.3.4, here presented as a momentum relaxation rate. Numerical calculation gives an IR mobility of $\mu_{IR} = 275 cm^2/Vs$ at $V_G = 0.5V$, and $\mu_{IR} = 92 cm^2/Vs$ at $V_G = 1V$.

Referring to analytical mobility analysis, the remotely screened model was expected to provide a peak increase of mobility at the source end of the channel of 20% at $V_G = 0.4V$ and 10% at $V_G = 1V$ when considering only phonon and II scattering. Using Mathiessen's rule, given by equation (6.1), the IR mobility can be included in the estimate. The effect of remote screening at Z = 48nm with IR scattering reduces from an increase of approximately 14% to 4.5% at low gate, and from approximately 4% to 0.5% at high gate (referring to the analytical mobility plots in Figure 6.6 and Figure 6.7). It is clear from this result that the inclusion of the IR scattering greatly reduces the effectiveness of remote screening on II scattering.

Further explanation for the negligible effect of remote screening can be understood from the regions over which remote screening has an effect on II scattering. This region is defined by the channel screening length which can be evaluated using the degenerate, non-parabolic Debye-Hückel model, equation (3.36). Examining plot (b) of Figure 6.11 and Figure 6.12, the electron densities in the channel can be approximated as $n = 2 \times 10^{19} cm^{-3}$ for $V_G = 0.5V$ and $n = 5 \times 10^{19} cm^{-3}$ for $V_G = 1V$ which gives screening lengths of $\lambda_c = 1.06nm$ and $\lambda_c = 0.79nm$ respectively. As discussed in the previous chapter (see Figure 5.16), the scattering model for remote screening has little effect beyond two channel screening lengths (and is negligible at $4\lambda_c$) which leads to a region of around 1.5-2nm from each interface where remote screening can be

considered to have a strong effect. This is less than 10% of the metallurgical channel length, which is not an appreciable portion of the channel.

Taking a simple estimate on the number of probable scattering events in this small remote screening region is possible and will highlight the importance of II scattering. For both gate voltages, the carrier drift velocity is around $v_d = 0.25 \times 10^7 \text{ cm/s}$ which means the average carrier will traverse the remote screening region in approximately $\tau_{tr} \approx 2nm/0.25 \times 10^7 \text{ cm/s} = 80 \text{ fs}$. Evaluating the BH scattering rate from section 3.3.3 and the remotely screened model (RSBH) from section 5.2.3.1 for the conditions outlined above, yields rates for both gate voltages on the order of $\Gamma_{BH} \approx 5 \times 10^{12} \text{ s}^{-1}$ and $\Gamma_{RSBH} \approx 2 \times 10^{11} \text{ s}^{-1}$ respectively. Taking these estimates for the scattering rate and transition time, the carriers travelling through the remotely screened region will have a probable number of scattering events of approximately $n_{\Gamma_{BH}} \approx \Gamma_{BH} \tau_{tr} = 0.4$ for the BH model and $n_{\Gamma_{BH}} \approx \Gamma_{BH} \tau_{tr} = 0.2$ for the RSBH model.

Therefore, within the remote screening region (here defined to be $\sim 2\lambda_c$ from either interface), the BH II model has a probability of scattering which is reduced by a factor of two in the RSBH II model. Although there is a factor of two reduction in probable scattering events, there are still relatively few events within the all important region of remote screening.

6.2.3.2 High Drain Bias

For the detailed analysis at high drain bias, the gate voltage points of $V_G = 0.4V$ and $V_G = 1V$ are chosen. At the higher drain voltage of $V_D = 1V$, the statistical noise in the MC simulation data is less of a factor. Examining the error bars for the high drain $I_D - V_G$ curves in Figure 6.14, this reduction in noise is clear.



Figure 6.14: I_D -V_G characteristics for the bulk MOSFET with error bars for the low gate voltage points at $V_D = 1V$.

A similar procedure is followed for the presentation and discussion of the high drain results as was taken in the low drain bias analysis. The II scattering tally for both gate voltages is plotted in Figure 6.15. Following this the electron velocity and density are plotted in Figure 6.16 for the 0.4V gate voltage and in Figure 6.17 for the higher gate voltage of 1V.



Figure 6.15: lonized impurity scattering tally along the channel for bulk MOSFET at (a) $V_G = 0.4V$ and (b) $V_G = 1V$ with $V_D = 1V$.

Figure 6.15 shows that the inclusion of remote screening reduces the II scattering tally at the source and drain interfaces. Also evident is the effect of the high drain voltage which biases the channel and reduces the II scattering tally towards the drain. With the large bias across the channel, the electron concentration is very much reduced at the drain end of the channel (see Figure 6.16 and Figure 6.17), causing the reduction in the II scattering tally.

The low gate voltage electron density, plot (b) in Figure 6.16, shows that the No Chn II case induces a higher density at the drain end of the channel. This is contrary to the behaviour of this scattering case at the lower drain voltage discussed in section 6.2.3.1, where increases in drain current could be explained solely by increases in channel velocity.



Figure 6.16: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 0.4V$ and $V_p = 1V$.

The increase in drain current of the No Chn II scattering case, as shown in Figure 6.14, is partly explained by the increased electron velocity in plot (a) of Figure 6.16, but more so by the significant increase in electron density in plot (b). With no II scattering in the channel region, the number of scattering events which can randomise the carrier trajectory will be smaller. This reduction in carrier trajectory randomisation will lead to a higher velocity of carriers traversing the channel from source to drain thereby altering the continuity balance and affecting the electron density.



Figure 6.17: (a) Velocity profile and (b) electron density along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering models at $V_G = 1V$ and $V_D = 1V$.

At the highest gate voltage, shown in Figure 6.17, the electron densities are a close match for all the models. The velocity of the No Chn II case is marginally larger than the other models, especially at the peak velocity point at approximately 70nm. At this gate voltage and drain bias, the carriers are passing the source interface at the saturation velocity and continue to accelerate along the channel, that is velocity overshoot occurs over the entire channel length. This behaviour has a strong effect on controlling the drain current where the velocity at the source interface shows that the No Chn II case is approximately 4% higher than the Num BH case. Comparing drain currents from the linear $I_D - V_G$ curves of Figure 6.8 also shows a increase in the region of 4%.



Figure 6.18: Energy profile along channel of bulk MOSFET comparing standard and remotely screened ionized impurity scattering at (a) $V_G = 0.4V$ and (b) $V_G = 1V$, for $V_D = 1V$.

Again the average carrier energies along the channel length remain very close to each other between the different channel II scattering cases. The average energy reached by the ensemble is considerably larger than the low drain bias. Near the drain end of the channel, the average carrier energy is around 600meV, an almost ten-fold increase on the low drain bias energies.

At high drain bias the remotely screened models again show negligible effect on the drain current. Interface roughness scattering continues to play a strong role in defining the channel performance. Removing IR scattering from the simulations provides a 22% increase at low gate voltage and a 28% increase at high gate. It is expected, from the analysis involving the analytical mobility estimate at low drain, that IR scattering will reduce the effectiveness of remote screening.

The region where remote screening can be considered strong can be approximated from evaluation of the screening length at the source and drain end of the channel, again making use of equation (3.36). Electron densities can be approximated from the plots of Figure 6.16 and Figure 6.17, but in this case it needs to be approximated for the both the source and drain end of the channel due to

the high drain bias. For the lower gate voltage of $V_G = 0.4V$, the electron density is approximated at $n = 10^{19} cm^{-3}$ for the source end and $n = 10^{18} cm^{-3}$ for the drain end. Evaluating equation (3.36) yields the respective screening lengths of $\lambda_C = 1.4nm$ and $\lambda_C = 4nm$ for the source and drain ends of the channel. The source end of the channel will have a region in the order of $2\lambda_C \approx 3nm$ where remote screening is strongly effective. This corresponds well with the drop in the II scattering tally given in plot (a) of Figure 6.15. This region corresponds to just over 10% of the metallurgical channel length, yet there is no improved performance. At the drain end of the channel, the remote screening region is even larger and can be estimated at $2\lambda_C \approx 8nm$, which is over a third of the channel length.

This lack of performance improvement can be understood by examining the impurity scattering in the remote screening regions at the source and drain end of the channel. Starting by looking at the source end of the channel first, estimates will be made to gain insight into II scattering in the remote screening region. For the low gate bias, the electron velocity can be estimated from Figure 6.16, with a drift velocity of $v_d = 0.4 \times 10^7 \text{ cm/s}$ at the source interface. Given a remote screening region of approximately 3nm, the transit time for the carrier through this region can be estimated as $\tau_{tr} \approx 3nm/0.4 \times 10^7 \text{ cm/s} = 74 \text{ fs}$. The BH and RSBH scattering rates can be evaluated from equations (5.64) and (5.40) from Chapter 5, using the conditions outlined above, yielding rates of $\Gamma_{BH} \approx 10^{13} \text{ s}^{-1}$ and $\Gamma_{RSBH} \approx 6 \times 10^{12} \text{ s}^{-1}$ at the source interface. Therefore a carrier has approximately $n_{\Gamma_{RSBH}} = \Gamma_{RSBH} \times \tau_{tr} = 0.42$ probable scattering events for the BH and RSBH models respectively, in the remote screening region at the source interface. The relative number of probable II scattering events is almost a factor of two decrease in number of events, but due to the few II events in this small remote screening region the effect is almost negligible.

At the drain interface the scenario is quite different due to the high drain voltage. The drift velocity from Figure 6.16 is $v_d = 3.3 \times 10^7 \text{ cm/s}$ at the drain interface with a corresponding transit time through the 8*nm* remote screening region of $\tau_{tr} \approx 24 \text{ fs}$. The average carrier energy, plotted in plot (a) of Figure 6.18, is around 0.45 eV which is very large. Evaluating the BH and RSBH scattering rates gives $\Gamma_{BH} \approx 6 \times 10^{13} \text{ s}^{-1}$ and $\Gamma_{RSBH} \approx 9 \times 10^{12} \text{ s}^{-1}$ at the drain interface. Estimating the probable number of scattering events at the drain end of the channel from this data, $n_{\Gamma_{BH}} = \Gamma_{BH} \times \tau_{tr} = 1.44$ and $n_{\Gamma_{RSBH}} = \Gamma_{RSBH} \times \tau_{tr} = 0.2$, shows a greater difference between the models. Despite this large difference between probable number of scattering events, there is still no noticeable change in the channel performance characteristics. This negligible shift in channel performance at the drain end is due to the very high carrier energy. Evaluation of the differential scattering cross section highlights the dominance of small angle, forward scattering at these high carrier energies. Such scattering events typically have only a small effect on the momentum relaxing effect of scattering.

At high gate voltages, the increased screening of the high carrier density reduces the remote screening regions to around 1.6nm at the source interface ($\lambda_c = 0.79nm$) and 2.5nm at the drain interface ($\lambda_c = 1.25nm$). This in conjunction with the high carrier velocity and increased energy greatly reduces the effectiveness of II scattering. The number of probable scattering events for each electron in either the source or drain remote screening region becomes very small, on the order of $n_{\Gamma} \approx 0.02 - 0.04$ per electron. This behaviour is expected at this high gate, high drain condition where the carrier density and average energy are at peak levels.

6.2.4 Summary

The bulk MOSFET device has an advanced channel doping profile, demonstrated in Figure 6.2, that includes a retrograde channel with halo implants which help to reduce short-channel effects. The channel is very highly doped at around $N_I \approx 10^{18} cm^{-3}$ (with source/drain wells at $N_I \approx 10^{20} cm^{-3}$) which will lead to a large amount of channel impurity scattering.

In this device the source and drain interface position have been simply chosen as the point of steepest gradient of the net doping, shown in Figure 6.4. This choice of position happens to coincide with the metallurgical p-n junction and gives the source interface at Z = 47nm and the drain interface at Z = 73nm.

Using initial DD simulation data, an analytical analysis of the phonon and impurity limited mobility was completed for low drain voltage conditions. This analysis provides insight into the effect that remote screening has on impurity scattering in a device context, and highlights the expected effect of remote screening on the channel performance. At very low gate voltages when the channel carrier density is low, remote screening leads to a large increase in carrier mobility. For $V_G = 0V$, the analysis predicts a channel mobility increase of over 100% along the channel. As the carrier density increases with gate voltage, the effect of remote screening is minimised.

At the minimum MC simulation gate voltage of $V_G = 0.4V$, the analytical mobility analysis predicts a peak increase of 20% at the source end of the channel and 10% at the drain end of the channel. Due to the high screening, this region of increased mobility is restricted to a few important nanometre's at each interface. At the highest gate voltage simulated, $V_G = 1V$, the effect

6.2 35nm Bulk Device

of remote screening is very small with a maximum increase of 10% at the source interface and a consistent 2% increase along the channel length.

MC simulation of the bulk MOSFET with the remotely screened II model demonstrated results that are much less dramatic than predicted with the analytical mobility analysis. The simulation of the newly developed II scattering model proved to offer negligible difference to the carrier transport in the channel region. Results are also obtained which removes the II scattering entirely from the channel to obtain an upper limit. These results show that in the bulk MOSFET, the II scattering process has a role in determining the channel performance and drain current.

Not considered in the analytical mobility estimate is the effect of IR scattering on the channel performance, which is found to strongly affect the bulk MOSFET device. At low drain bias, removal of IR scattering from the device increases the drain current by up to 88% at a gate voltage of $V_G = 1V$. A simple estimate for the IR limited mobility is made at low drain voltage to evaluate the effect on the remote screening. Through use of Mathiessen's rule, the IR limited mobility was added to the analytical mobility and found to reduce the effectiveness of remote screening from 14% to 4.5% at low gate and from 4% to 0.5% at high gate. IR scattering is shown from this simple estimate to be a dominant mechanism in the bulk MOSFET device and reduces any appreciable effect of remote screening.

The region in a device channel where remote screening is effective was also analysed and found to be around 2-3nm at the source end, but could be as much as 8nm at the drain end of the channel. For the metallurgical channel length of 26nm in this device, these remote screening regions can be an appreciable portion of the channel length. Examining the II scattering in these remote screening regions, it was shown that for the source end of the channel, probable II scattering events are quite low. Typically there is a probability of less than one II scattering event per carrier travelling through the remote screening region.

At low drain bias, the carriers have approximately $n_{\Gamma_{BH}} \approx 0.4$ probable scattering events with the BH model which halves to $n_{\Gamma_{RSBH}} \approx 0.2$ with remote screening. At high drain bias and low gate voltage conditions, the probability of II scattering is slightly higher with $n_{\Gamma_{BH}} \approx 0.75$ for BH which drops to $n_{\Gamma_{RSBH}} \approx 0.42$. Despite the consistent drop with the use of remote screening of II scattering, the number of events is too low to affect the channel performance characteristics.

At the drain end of the channel for high drain voltages, where the remote screening region is large and can be up to 8nm, the carrier energy is so large that II scattering has little effect. For high energy carriers, the scattering is dominated by small angle forward scattering events which have little momentum relaxing effect on the carriers.

6.3 22nm Double-Gate Device

The ultra-thin body double gate (UTB DG) device has been developed as a template device for the PullNano European research project. It was developed originally to compare device simulations amongst a consortium of European research groups [116]. Here this device has been reconstructed and calibrated using the commercial Sentaurus software to the original specification, then exported for use with the in-house DD simulator. As discussed before, the MC simulation uses the DD solution output as a starting point and requires no further calibration.

Repeating a similar analysis for the UTB DG device as for the bulk MOSFET, the section will start with a brief discussion of the device structure and the calibration between simulations in section 6.3.1. Using initial DD data, an analytical calculation of the carrier mobility with and without remote screening has been completed with discussion in section 6.3.2. Section 6.3.3 presents and examines the MC simulation data for the device with the newly developed II scattering mechanisms.

6.3.1 Device Structure and Calibration

The UTB DG device has been developed and optimised for low standby power (LSTP) applications for the 22nm technology node. The device has a 22nm metal gate (a work function of $\phi_G = 4.8eV$) with a 1.1nm silicon-dioxide insulator layer and a 10nm silicon body thickness as depicted in Figure 6.19. The remote screening plane definition is given in the lower-left corner of Figure 6.19 and the positions of the remote screening interfaces at Z=0 and Z=L_C are marked.



Figure 6.19: Structure of the Ultra-Thin-Body Double-Gate device.

This device was originally developed with a high- κ dielectric with effective oxide thickness (EOT) of 1.1nm but this has been disregarded here as the high- κ dielectric requires additional scattering mechanisms for accurate simulation. Instead, the oxide layer is replaced with silicon dioxide of 1.1nm so that the electrostatic nature of the insulator is retained without the additional overhead of dealing with the advanced gate stack of a high- κ dielectric.

The net doping profile for the structure is shown in Figure 6.20. The DG device has a low doped ptype channel with $N_A = 1.2 \times 10^{15} cm^{-3}$ and highly doped source/drain regions at $N_D = 5.2 \times 10^{19} cm^{-3}$.



Figure 6.20: Net doping concentration for the Ultra-Thin-Body Double-Gate device.

The source and drain doping is a constant doping level from the edges of the device up to 50.5nm and 84.5nm respectively, with a Gaussian function roll-off of the doping into the channel region. This gives a gate underlap of 6nm from the region of constant doping of the source and drain to the metal gate.



Figure 6.21: (a) Linear and (b) semi-log I_D -V_G calibration plots showing the comparison between Sentaurus, Drift-Diffusion and Monte Carlo simulation.

The DD simulation was calibrated against the commercial TCAD tool, Sentaurus [117] at both low and high drain voltages of $V_D = 0.1V, 1V$. The calibrated $I_D - V_G$ characteristics for both drain voltages are shown in Figure 6.21 along with the MC simulation output. The MC simulation is a close match with the DD simulation when non-equilibrium carrier transport effects are minimal, that is at low drain and/or low gate voltage. The effects of non-equilibrium transport are clear for the MC results for high drain at high gate voltage.

The source and drain interfaces must also be selected for the remotely screened II models and again the choice of interface position is taken here solely through the use of the difference plot of the net doping profile, plot (b) of Figure 6.22.



Figure 6.22: (a) Net doping profile with signed log scale and (b) net doping difference plot for the Ultra-Thin-Body Double-Gate device.

The position where the net doping is at its maximum gradient, highlighted by the difference plot, is chosen here as the position of the interface and is considered to be closest to the original definition of the interface as the point where the doping density changes abruptly from source/drain to channel concentrations. The source and drain interfaces, depicted in Figure 6.22 as the green dashed lines, are at 53.5nm and 81.5nm giving a novel channel length of around 28nm.

6.3.2 Analytical Estimate of Remote Screening Effect

An estimate of the remote screening effect on carrier transport through the channel of the DG device can be calculated using the Kubo-Greenwood formula for mobility [33]. The spatially varying channel mobility can be estimated using the momentum relaxation rates for phonons and BH and RSBH II models along with DD simulation data for impurity and electron concentrations.

The method used to obtain the channel mobility here is identical to that explained and utilised in section 6.2.2. To recap, Mathiessen's rule is used to sum the individual scattering mechanisms

along with an ensemble averaged momentum relaxation length to take a moving median of the spatially varying total mobility. The phonon relaxation rates presented in section 3.3.2 are used to calculate phonon mobilities independently. The II scattering models are calculated from expressions given in the previous chapter, Chapter 5, and include the BH, single interface RSBH and double interface RSBH models. The analysis has been completed for three gate voltages, $V_G = \{0V, 0.4V, 1V\}$, at the low drain voltage of $V_D = 0.1V$.

In Figure 6.23 the plots of (a) electron and impurity concentration and (b) phonon and impurity limited mobility are given for the lowest gate voltage of $V_G = 0V$. The ensemble average momentum relaxation rate is calculated using equation (6.2) to be $\langle \tau_m^{tot} \rangle \approx 5.6 \times 10^{-14} s$, giving a mean free path as $l_{\tau_m} \approx 5.6 m$.

The effect of the minimal impurity density in the channel region is shown in plot (b) of Figure 6.23, where the mobility is entirely phonon controlled between roughly 63-72nm. It is within this phonon controlled region that impurity concentration is just above $N_I = 10^{15} cm^{-3}$. At this low density the II limited mobility is extremely high and can be considered to have negligible effect. For example, for an impurity density of $N_I = 10^{15} cm^{-3}$ with an electron concentration of $n_e = 10^{10} cm^{-3}$, the impurity limited mobility is roughly $\mu_{II} \approx 150\ 000\ cm^2/Vs$.



Figure 6.23: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 0V$ and a drain voltage of $V_D = 0.1V$.

II scattering in the channel has only an effect within the region of the Gaussian roll-off of the source/drain doping at the edges of the channel. This region of II scattering controlled transport is clearly shown in plot (b) of Figure 6.23. The II controlled region constitutes approximately a 9nm region at each end of the channel where remote screening affects the channel performance.

At this low gate voltage where the electron density in the channel is very low, remote screening has a large effect in the mobility. At the source end of the channel both remotely screened models provide in the region of a 150% increase in mobility, and around 200% for the double interface model at the drain end of the channel.

The analytical mobility estimate for the gate voltage $V_G = 0.4V$, is plotted in Figure 6.24. The average ensemble momentum relaxation rate is calculated as $\langle \tau_m^{tot} \rangle = 6.2 \times 10^{14} s$ giving a mean free path of $l_{\tau_m} = 6.1 nm$.



Figure 6.24: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 0.4V$ and a drain voltage of $V_D = 0.1V$.

Increasing the gate voltage to $V_G = 0.4V$, the electron concentration in the channel of the device has increased by several orders of magnitude, as plotted in (a) of Figure 6.24. This increase in the screening concentration leads to a reduction in the strength of II scattering to control the mobility, as shown in the mobility, plot (b). This of course has the roll on effect of reducing the strength of remote screening on the II scattering. Despite the large increase in the screening within the channel, there is a peak increase of channel mobility of approximately 40% at the source and 60% at the drain for the single and double interface models, respectively.

Increasing the gate voltage to the highest simulated, $V_G = 1V$, increases further the channel electron concentration which is now several orders of magnitude bigger than the impurity concentration. At this gate voltage, the ensemble momentum relaxation rate is found to be $\langle \tau_m^{tot} \rangle = 7.8 \times 10^{14} s$, leading to a mean free path of $l_{\tau_m} = 7.8nm$.



Figure 6.25: Plot of (a) the impurity and electron density, and (b) the impurity limited mobility for the UTB DG MOSFET at a gate voltage of $V_G = 1V$ and a drain voltage of $V_D = 0.1V$.

The increase in screening will further reduce the effectiveness of remote screening in increasing the channel mobility, which is expected at high gate voltages. Although the effect of remote screening is minimised, it still leads to approximately a 40% increase in the mobility at either end of the channel (drain increase due to double interface model).

Summarising this simple analysis of the phonon and impurity limited mobility for the DG device, it is clear that the low doped channel region is significantly less controlled by impurity scattering than in the case of a typical bulk MOSFET. Impurity scattering plays a negligible role within the 9nm metallurgical p-n junction region in the centre of the device, although there is still significant II scattering surrounding the source and drain interfaces. Remote screening is shown to have a very large increase at the low gate voltage of $V_G = 0V$, in the region of 150-200%. This large increase in mobility at low gate voltages is predicted to increase the sub-threshold leakage of this MOSFET device. As the device as been developed to fulfil the LSTP requirements, the effect of remote screening could play a strongly detrimental role in the device performance.

As the screening strength increases, the strength of remote screening reduces, as is expected. At $V_G = 0.4V$, the new II scattering model is expected to increase the channel mobility near the interfaces by up to 40% at the source and 60% at the drain. Even at the highest gate voltage, there is an expected 40% increase in channel mobility within the region of II controlled mobility near the interfaces.

6.3.3 I_D-V_G Behaviour with Remote Screening

The UTB DG device has been simulated in the MC code with four different channel II scattering cases, identical to the bulk MOSFET simulation discussed previously in this chapter. These cases

are: Num BH – numerically capped Brooks-Herring scattering in the channel region as presented in section 5.4.1; No Chn II – all channel II scattering is removed; RSBH Sgl – remotely screened BH model for the single (source) interface case; RSBH Dbl – remotely screened BH model for the double interface case.

Simulations have been completed to obtain the $I_D - V_G$ characteristics for a range of gate voltages of $V_G = \{0.4V, ..., 1V\}$ and drain conditions $V_D = 0.1V, 1V$. The restricted range of gate voltages is again due to the statistical noise in MC simulation at low gate voltages, when the majority carrier concentration in the channel region becomes extremely low. Here Figure 6.26 present the results of the MC simulations for all four II scattering cases with error bars at each data point giving the statistical error due to noise.



Figure 6.26: (a) Linear and (b) semi-log I_D -V_G characteristics for the UTB DG MOSFET with remotely screened impurity scattering.

Examining the linear plot of the $I_D - V_G$ characteristics of plot (a) in Figure 6.26 show that II scattering plays a very minor role in determining the channel performance. Simulations with the channel II scattering turned off show that there is a very small increase in the drain current at high gate voltages. This increase in drain current corresponds approximately to a maximum increase of around $20-30\mu A / \mu m$ at the peak gate voltage.

The logarithmic plot of the drain current, plot (b) of Figure 6.26, further demonstrates the lack of impurity scattering in controlling the device performance. At the lowest gate voltage point, $V_G = 0.4V$, the statistical noise is dominant and examination of the results are not reliable due to this. At the higher voltage of $V_G = 0.5V$, the noise is still considerable but the results seem to corroborate the minimal affect of impurity scattering. These results strongly demonstrate that impurity scattering in the channel region of this particular device is very much a negligible effect in determining the device performance. Remote screening clearly has negligible effect in defining the
drain current. A brief analysis of several data points at low and high drain will be completed in the following sub-sections to confirm the behaviour seen in the $I_D - V_G$ plots.

6.3.3.1 Low Drain Bias

The low drain voltage analysis will examine two gate voltage points to confirm the behaviour shown in the $I_D - V_G$ characteristics. A low gate voltage point will be chosen below with the aid of the error bars in the $I_D - V_G$ plots, Figure 6.27. The high gate voltage will be the $V_G = 1V$ point which can be selected without error bar consideration as statistical noise is not a factor at high fields.



Figure 6.27: Semi-log plot of the low gate voltage I_D - V_G characteristics with error bars for the UTB DG MOSFET at $V_D = 0.1V$.

Figure 6.27 gives the low gate voltage drain currents in more detail with their corresponding error bars. The error bars at the lowest gate voltage of $V_G = 0.4V$ are particularly large and have been clipped to fit them on the plot, hence the arrow as the bottom cap. Considering the magnitude of the noise at this voltage this point is not analysed here. At the higher gate voltage of $V_G = 0.5V$ the error bars are still quite large despite the corroboration between the drain currents, and for this reason it also will not be analysed here. Instead, the data point at $V_G = 0.6V$ will be analysed given that results appear stable, given the small error bars.



Figure 6.28: lonized impurity scattering tally along the channel for UTB DG MOSFET at (a) $V_G = 0.6V$ and (b) $V_G = 1V$ with $V_D = 0.1V$.

Starting by looking at the II scattering tally for both gate voltage points in Figure 6.28 (note that the II scattering tally is given with arbitrary units). Here the II scattering tally is extremely low in the centre of the channel of the device which, referring to Figure 6.22, is expected as the dopant density is very low. In both plot (a) and (b) of Figure 6.28 the effect of remote screening on the II scattering tally is evident. There is a distinct reduction in the number of events at both of the interfaces, considering either the single or double interface model. This is consistent with the expectation of a reduced scattering rate with remotely screened II scattering.

Given the close match between the drain currents at low gate voltages of each of the II cases considered here (Figure 6.26), the velocities and carrier densities are expected to be almost identical between the models. At high gate voltages only the case that removes channel II scattering, No Chn II, is expected to demonstrates any shift in velocity/carrier density.



Figure 6.29: (a) Velocity profile and (b) electron density along channel of UTB DG device with $V_G = 0.6V$ and $V_D = 0.1V$.

The electron velocity and density for the low gate voltage point are shown in Figure 6.29. As expected, given that the drain currents match, the carrier densities and velocities are very close along the length of the channel for all four cases.



Figure 6.30: (a) Velocity profile and (b) electron density along channel of UTB DG device with $V_{g} = 1V$ and $V_{D} = 0.1V$.

At a gate voltage of $V_G = 1V$ the electron densities of plot (b), Figure 6.30 are identical. As expected the No Chn II case demonstrates an increase in electron velocity over an appreciable length of the channel, approximately a 5% increase in velocity over the Num BH case. This explains the increase in drain current at this data point displayed in the drain current plot of Figure 6.26.



Figure 6.31: Energy profile along channel of UTB DG device with (a) $V_G = 0.6V$ and (b) $V_G = 1V$ for $V_D = 0.1V$.

The UTB DG device shows negligible difference with the inclusion of remotely screened impurity scattering at low drain bias. This is explained by the low doped channel, which due to the small impurity concentration has minimal channel II scattering. This is highlighted by the very small

shift in drain current when channel II scattering is removed from the simulation. The small amount of impurity scattering at the source and drain end of the channel account for the modest increase in drain current at very high gate voltage. Of course, with high gate voltages the screening density in the channel is large and remote screening is minimised. For $V_G = 1V$ the electron density in the channel can be approximated at $n = 10^{19} \text{ cm}^{-3}$ and evaluating equation (3.36) gives a screening length of $\lambda_C = 0.92 \text{ nm}$.

The role of IR scattering has been studied in this device and is found to have a small effect in defining the channel performance. Removal of IR scattering in the device causes around a 2-4% increase in drain current. This lack of IR scattering is an advantage of the UTB DG device as the channel enters the volume inversion regime [125]. That is, for silicon body thicknesses between $3nm < t_{si} < 20nm$, the channel region becomes almost fully inverted at high gate voltage causing the conducting channel to be located in the middle of the body. This is known to reduce the influence of surface scattering events including IR scattering [125-127].

6.3.3.2 High Drain Bias

With the high drain bias the statistical noise in the results is reduced considerably, as shown in Figure 6.32. Again, analysis of the results will be undertaken for two gate voltages. The low gate voltage point will be $V_G = 0.5V$ and the high gate voltage at $V_G = 1V$. Here the results are expected to be close to each other except for the No Chn II case at high gate voltages, similar to the low drain results.



Figure 6.32: Semi-log plot of the low gate voltage I_D - V_G characteristics with error bars for the UTB DG MOSFET at $V_D = 1V$.

The II scattering tally is plotted for both gate voltage points in Figure 6.33. The distinctive behaviour of the remote screening models is evident at the source interface where the II scattering tally drops abruptly at the interface and at the drain in the high gate voltage plot of (b). At the drain interface of the low gate voltage in plot (a), this distinctive behaviour is not present and all the II scattering cases drop off rapidly.

A closer examination of plot (a) in Figure 6.33 shows that the numerically capped BH model (Num BH) matches with the remotely screened single interface model (RSBH Sgl) as is expected at the drain interface. The remotely screened double interface model (RSBH Dbl) follows the no channel II scattering case (No Chn II) which demonstrates that the II scattering at the drain is being correctly remotely screened.



Figure 6.33: lonized impurity scattering tally along the channel for UTB DG MOSFET at (a) $V_G = 0.5V$ and (b) $V_G = 1V$ with $V_D = 1V$.

This change in behaviour at the drain in plot (a) of Figure 6.33 is obviously due to a reduction in II scattering at the drain end of the channel. Given the high drain bias of $V_D = IV$, the channel inversion layer will be biased such that the carrier density at the drain end of the channel will be lower. Combined with the increased carrier energy due to the drain bias, plot (a) of Figure 6.36, this results in a reduced II scattering tally at the drain end of the channel. This behaviour is also shown to a lesser effect in plot (b) of Figure 6.33.

To confirm this justification a simple calculation can be completed, through use of equation (6.8), to obtain an estimate on the number of carriers present in the drain end of the channel.

$$n_e^{vol} = V \times n_e \tag{6.8}$$

Here n_e^{vol} is the number of electrons in a volume given by V with an electron density of n_e . Referring Figure 6.19 the volume of the to channel is given as $V = 34nm \times 10nm \times 1nm = 3.4 \times 10^{-19} cm^3$ (for the MC simulations completed here the device width is 1nm), but considering only the drain end of the channel this becomes $1.7 \times 10^{-19} cm^3$. The electron density is given in plot (b) of Figure 6.34 and for the drain end of the channel will be estimated at $1.5 \times 10^{16} cm^{-3}$. Evaluating equation (6.8) gives an electron number of 2.55×10^{-3} in the drain end of the channel which relates to roughly 4 superparticles (1 superparticle $\approx 5.7325 \times 10^{-4}$ electrons).



Figure 6.34: (a) Velocity profile and (b) electron density profile along channel of UTB DG device with $V_G = 0.5V$ and $V_D = 1V$.

The velocities of each of the four channel II scattering cases for the low gate voltage, plot (a) of Figure 6.34, are comparable given the relatively few electrons (or superparticles) that travel through the channel region. At the high gate voltage shown in Figure 6.35, the velocities are again very close to each other with the No Chn II case being the largest by a small margin. This increased velocity explains the drain current increase shown in Figure 6.26.



Figure 6.35: (a) Velocity profile and (b) electron density profile along channel of UTB DG device with $V_G = 1V$ and $V_D = 1V$.

Plot (a) of Figure 6.35 shows that the electrons are above the saturation velocity over a large proportion of the channel. This phenomenon, which is a feature of non-linear transport in MC simulations, provides an increase in drain current beyond the velocity saturated value [15]. An interesting point demonstrated in these figures is that the peak velocity and energy of the carriers at low gate voltage is much larger than that at high gate voltage. This can be explained through the density of carriers that traverse the channel at low or high gate voltage.



Figure 6.36: Energy profile along channel of UTB DG device with (a) $V_G = 0.5V$ and (b) $V_G = 1V$ for $V_D = 1V$.

At low gate voltage the potential barrier in the channel between the source and drain is high and only high energy carriers are able to make it into the channel. These high energy carriers that make it into the channel will encounter a large field accelerating them into the drain. It is these few carriers from the high energy tail of the carrier distribution that are being averaged to obtain the velocity plot in Figure 6.34. This is corroborated by the high average energy of the carriers in the channel shown in plot (a) of Figure 6.36.

6.3 22nm Double-Gate Device

In the high gate voltage case the potential barrier is much lower and a larger fraction of the carrier distribution can traverse the channel to the drain. As carriers with lower energy are able to contribute to the drain current, the average energy in the channel will decrease along with the average carrier velocity.

At high drain the overall effect of remote screening is negligible which again is due to the low impurity density in the channel. The effect of the impurity scattering in the small regions at each of the channel provide only a small factor in defining channel performance at high gate voltage. Due to the new models inherent dependence on low screening, the high gate voltage cancels the effect that remote screening has in this device.

6.3.4 Summary

The 22nm template UTB DG device was originally developed as a template device to allow a comparison to be made between a large set of different simulators from the European device modelling community [116]. The device structure consists of a 10nm thick silicon body with a 1.1nm EOT oxide layer, here modified to be silicon dioxide from the original specification of a high- κ material.

With highly doped source and drain contact regions and a low doped channel region, the definition of source and drain interface positions for the remotely screened models is again chosen as the point of steepest gradient in the roll-off of the source and drain doping into the channel. This gives the source and drain interfaces at positions of Z=53.5nm and Z=81.5nm respectively.

An initial estimate on the effect of remote screening has been completed by analytically computing the phonon and II limited mobility from initial DD simulation. This examination showed that II scattering in the centre of the channel plays little role in determining the channel performance where dopant concentration is at its lowest. Channel II scattering has a controlling factor in the device performance near the interfaces where the source and drain doping is still evident and coincides with the region that remote screening is active. Within these small regions at the interfaces, the analysis demonstrated that remote screening will increase the channel mobility between 40-60% at gate voltages greater than $V_G = 0.4V$.

The MC simulation of this device includes simulation of four different cases of channel II scattering that include the BH scattering model, no channel II scattering and the remotely screened models. Simulations were completed to produce $I_D - V_G$ curves for low and high drain biases for each of the four cases. The drain current curves established that channel II scattering in this device

6.4 Conclusion

is not a strongly controlling factor on the device performance. This is made particularly clear by the No Chn II case which removes all II scattering from the channel and had little effect on the drain current. Given this conclusion, the new remote screening models for II scattering have an understandably negligible effect on the device characteristics.

6.4 Conclusion

This chapter has taken the newly developed remote screening II scattering models presented in Chapter 5 and applied them in a fully self-consistent MC device simulator. The objective of this simulation study is to analyse what effect that remote screening has in the modern MOSFETs where the distance between the highly-doped source and drain regions is small. The MC device simulator has been employed to simulate two distinct MOSFET devices, one with a highly doped channel typical of the current generation MOSFETs and the second device a future generation MOSFET with low doping in the channel.

The first device is a bulk MOSFET which is forecast to continue in the industrial roadmap until at least 2012 [1]. Originally developed by Toshiba as a future device for the high-performance technology [113, 114], it has been reconstructed from published data by fellow researchers in the Device Modelling Group [118]. This device has been constructed with a metal gate, a 1.4nm thick silicon oxi-nitride insulator and a highly doped, advanced channel doping profile design. Calibration of the bulk MOSFET is completed between the industrial TCAD tool, Sentaurus [117], the in-house DD simulator and the MC simulator.

Although MC simulation incorporates non-equilibrium transport that is not present in DD, it suffers from the problems of statistical noise inherent to such a stochastical particle based approach. Due to this limitation, the bulk MOSFET device can only be accurately simulated for gate voltages greater than $V_G = 0.4V$. This is an important restriction to analysis of remotely screened II scattering as very low screening conditions, where remote screening is expected to be strongest, cannot be simulated. This is a limitation of using the MC methodology which can only be worked around by moving to a different simulation approach that doesn't suffer from stochastical noise.

An important step in the modelling of remote screening is the choice of position for the interfaces between the source and the drain. Abrupt doping transitions don't occur in realistic doping profiles and therefore in this work the gradient of the doping profile is used to select the position. The point at which the doping is changing most rapidly between the highly doped source/drain and the channel is selected as the remote screening interface.

6.4 Conclusion

An analytical analysis of the phonon and II limited mobility has been completed using the Kubo-Greenwood formula in conjunction with channel profile data from the DD solution of the bulk MOSFET. This analysis allows the effect of remote screening to be estimated at low screening conditions (low gate voltages) and provides a picture of the region that remote screening is effective in the device. The analysis highlights that remote screening will have a larger effect at low gate voltages where the screening density in the channel is low, as is expected. As the gate voltage is increased, so does the screening density and the effect of remotely screened II scattering is reduced to smaller regions at the edge of the interfaces. At very low gate voltages in particular, the II and phonon limited mobility in the channel region is increased by almost a factor of two.

Simulations of the bulk MOSFET in the MC simulator have been completed with and without the remotely screening impurity scattering, and also for the case of no impurity scattering in the channel. Results without II scattering in the channel of the bulk MOSFET show that in this device that the II scattering mechanism plays an important role in defining the channel performance. The drain current is substantially increased when no channel II scattering is included. However when the remotely screened II model is included, the simulation shows negligible effect at any particular gate voltage or drain bias.

This lack of improvement in channel performance with remotely screened II scattering can be explained partly by the strong role that interface roughness scattering has in defining channel performance. The analytical estimate using DD data for the phonon and II limited mobility suggested that remote screening may have up to a 20% increase in channel mobility within the remote screening regions at either end of the channel. It was discussed that the inclusion of IR scattering reduces this analytical estimate of the channel mobility to less than 5%. Secondly, the region over which remote screening is effective in the bulk MOSFET device is small with few II scattering events. Typically, the number of scattering events that occur within the remote screening with a small region where remote screening is effective leads to a negligible effect for the bulk MOSFET.

The second device to be simulated was the PullNano UTB DG device which is expected to be the device that allows scaling to the end of the industrial roadmap [1]. The UTB DG device has an low doped channel region with a 10nm thick silicon body and 1.1nm oxide layer. A difference plot of the net doping profile has again been used to find the source and drain interface positions.

An analytical estimate on the channel mobility has been completed and demonstrates the effect that remote screening has on the channel performance. At the lowest gate voltage, remote screening is shown to have an increase in channel mobility within the remote screening region at either end of the channel. Predictably, as the gate voltage increases the effect of remote screening is reduced. At the minimum MC simulation gate voltage, $V_G = 0.4V$, it was shown that remote screening can increase the mobility close to the interfaces by up to 60% and at the highest gate voltage this is reduced slightly to 40%. Despite this increase the remote screening region is limited to a small regions at either end of the channel with the centre of the channel dominated by phonon scattering.

MC simulation of the device with the remote screening models doesn't provide the expected increase in channel performance as was estimated by the mobility analysis. It is shown that impurity scattering in the channel plays an almost negligible role in defining the channel performance. This lack of II scattering effect can be understood from the channel doping profile, which is very low, on the order of $N_I \approx 10^{15} cm^{-3}$. At this low doping, II scattering is negligible and the phonon scattering is dominant.

In conclusion, it has been shown by extensive simulation of two distinct MOSFET devices, that remote screening plays no role in defining the channel performance in these devices at the simulated gate voltages. In the bulk device, the dominant IR scattering combined with high screening strongly reduces the effect of remote screening. With the UTB DG device, it is explained by the low dopant density in the channel removing the dependence of II scattering in the channel characteristics.

In sub-threshold conditions the remote screening model has been shown to have a considerable effect on channel mobility and is predicted to increase the leakage current when the device is off (at low gate voltages). The restriction introduced by MC simulations on the allowable gate voltages means that simulations cannot be completed to examine this effect on leakage current. Further work is required to study the effect of this new scattering model at lower gate voltages using a different simulation methodology which doesn't suffer from the statistical noise of MC.

Chapter 7 Conclusions

The aim of this work was to develop and test an advanced Coulomb scattering model that includes the effect of the highly-doped source and drain regions on channel ionized impurity (II) scattering for the simulation of nanoscale MOSFET devices. The highly-doped source and drain regions introduce additional screening of the channel IIs through polarisation charge effects, the aptly named remote screening within this work, which as channel lengths are scaled in MOSFETs will increase and strongly affect II scattering in the channel region.

Here scattering potentials are developed which represent a single channel II located next to one or two highly doped regions, the source and drain, for the single and double interface cases respectively. An exact analytical solution to the Linearized Thomas-Fermi (LTF) form of Poisson's equation for the system has been calculated and is shown to be an excellent match with a fully self-consistent numerical Poisson solution. These scattering potentials include the important polarisation charge effects induced from the source and drain regions which increase the screening of channel IIs, hence the descriptive title of remote screening.

It has also been demonstrated that the remote screening scattering potential will return to the Brooks-Herring (BH) scattering potential in the limit that the II is located a large distance from the highly-doped regions. This is an important limit to obtain as it ensures that the newly developed scattering potential not only returns to the classic solution, but that it will also work alongside existing II scattering approaches. This is of importance for integration into Monte Carlo (MC) simulation.

For the purposes of this work which is to undertake a study on the strength of remote screening in MOSFET devices using a MC simulation methodology, a simplified model of remote screening has been employed. The simplified model is obtained using the aptly named strong-screening limit which assumes the source and drain regions become metallic-like. This simplification is shown to be almost exact for IIs located greater than 1nm from an interface and represents a worst-case scenario for remote-screening interactions. This worst-case scenario providing an upper limit on the remote screening interaction and is suitable for the context of this work.

Utilising the strongly screened scattering potentials, a MC suitable scattering model has been calculated using Fermi's Golden Rule approach. Again, as the context of this work is to explore the strength of remote screening in device simulations, an assumption on the scattering potential has been made to reduce the complexity of the problem. The remotely screened scattering potential

7. Conclusions

is anisotropic in nature but the typical approach to the calculation of scattering rates in MC is for isotropic scattering potentials. In this work the scattering potential is assumed to be isotropic such that the scattering carrier always approaches the II aligned with the Z-axis of the scattering frame. Use of the Z-aligned simplification is shown to lead to a less than 2% error over the region where remote screening is effective.

An analysis of the remotely screened scattering models has shown that remote screening reduces the strength of II scattering near to the highly-doped source and drain regions. For impurities within 1-2 channel screening lengths of either interface, the effect of remote screening can reduce the scattering rate by up to several orders of magnitude. As channel lengths shrink, the remote screening effect induced from the source and drain increases, particularly when the channel screening length becomes larger than the channel length. In such cases, the II limited mobility can increase up to 300% over the standard model for channel lengths less than 20nm.

The existing approach for II scattering utilised in the MC simulation in this work made use of Ridley's Third-Body Exclusion (TBE) model which is far from ideal as it neglects the vast majority of II scattering events when screening densities are low. This becomes a distinct problem for the accurate modelling of MOSFET devices as channel screening densities can be several orders of magnitude lower than II density leading to an underestimation of the scattering effect. In this work a new approach has been developed which allows the full complement of II scattering rate to be modelled. This new approach utilises a self-consistent model that allows the scattering to be fully modelled improving the low screening density simulations whilst ensuring that the simulation analysis of remote screening is accurate.

The remotely screened II scattering model has been used to complete a simulation study of two MOSFET devices which are chosen to represent a the state of the art technology and possible future technology. The first device being a bulk MOSFET that is typical of those currently in production with high channel doping, and the second device is an low channel doped Ultra-Thin Body Double Gate (UTB DG) MOSFET that is a proposed structure for future technological generations. These two devices are chosen to represent the trends in channel doping design, that is the highly doped bulk MOSFET or the relatively undoped channel design of the DG design.

The bulk MOSFET device has a strong dependence on II scattering which was demonstrated by the clear improvement in drain current when this scattering mechanism was removed from the channel region. An analytical mobility analysis due to phonon and II scattering suggested that remote screening in this device will greatly improve the channel mobility at low gate voltages. This increase in low gate voltage channel mobility would lead to a detrimental affect on device

7. Conclusions

performance by increasing the off-state current of the MOSFET. MC simulation of the device with remote screening of channel IIs demonstrated that for gate voltages above 0.4V, the effect is negligible. This negligible result was shown to be due to a combination of the high Interface Roughness (IR) scattering in the device which damped any effect of remote screening and the small remote screening region at the respective gate voltages.

The UTB DG MOSFET device was demonstrated to have a minimal dependence on II scattering due to the low doping in the channel region. The analytical analysis highlighted the phonon scattering dominance in the channel which was reinforced by the minimal shift in drain current with the removal of channel II scattering. Again, MC analysis of remote screening has a negligible effect on the device performance which can be understood from the lack of II scattering in the channel.

To conclude, the remote screening for II scattering has been shown for higher gate voltages to have negligible effect in current generation and low channel doped devices. Remote screening of channel IIs is expected to have a large effect at very low gate voltages where screening densities are low. Unfortunately the MC simulation approach is not suitable to simulate such conditions and future work should aim to simulate such low screening conditions perhaps through a direct solution of the Boltzmann equation using spherical harmonics expansions [73, 128, 129].

7.1 Future Work

The study of remote screening with sub-threshold MOSFET conditions should be completed to fully understand the role of this effect. This will involve moving to a simulation methodology that doesn't suffer from statistical noise. An approach which is suitable is that of the spherical harmonics expansion of the distribution function enabling a deterministic solution of the Boltzmann Transport Equation [73, 128, 129]. This would allow the low gate voltage simulations of MOSFETs to be completed whilst including the remote screening scattering mechanism.

Further studies of MOSFET devices with smaller dimensions are required to confirm the effect of remote screening and could be completed using the simulator and remote screening scattering approach developed here. Example devices include the final scaled devices of each particular architecture such as a 14nm physical channel length bulk MOSFET, a 10.7nm physical channel length Silicon-On-Insulator MOSFET and a 8.1nm physical channel length DG device [1]. This would provide a comprehensive study on the effect of remote screening in nanoscale devices by extending the study to the end of roadmap scaled devices.

7.1 Future Work

The full anisotropic scattering potential should also be studied in simulations of MOSFETs which would allow the full scattering effect of the remotely screened II to be analysed. There are several approaches which could be used including the radial wave expansion of the Schrödinger equation [105] or perhaps use of a linear Boltzmann expansion to obtain scattering expressions [108]. It would also be beneficial for an exact description of the remote screening effect to remove the strong-screening limit from the scattering models. In this work an upper limit or worst-case condition for remote screening was sufficient to be able to examine the effect but could be extended to provide a more detailed study. This may be required for the ultra-small channel length devices predicted for the end of the ITRS roadmap.

Finally, the remote screening of channel ionized impurities from the gate region should also be studied. The gate contact in future generation MOSFETs will become a metal and in conjunction with oxide thickness scaling, will lead to strong remote screening of channel IIs. This interaction will be evident along the entire length of the channel, extending the remote screening regions to a larger portion of the device and possibly leading to a stronger effect. In particular, this should be studied for possible future multiple gate MOSFET architectures where the channel can be surrounded by metallic gate regions.

Appendix A Single Interface Potential Calculation

This appendix will provide a detailed outline of the single interface potential calculation discussed in section 4.2.1. An exact solution to the potential of a single ionized impurity located close to highly-doped region has been obtained by solving Poisson's equation. This is a complex problem which has been simplified through use of the linear Thomas-Fermi (LTF) approximation to allow an analytical solution to be found.

A.1 Linearized Poisson's Equation

Starting with the LTF form of the Poisson equation, equations (4.4)-(4.5) from section 4.2.1 which define the system depicted in Figure 4.1.

$$\nabla^2 \varphi_s - k_s^2 \varphi_s = 0 \quad \text{for } Z < 0 \tag{A.1}$$

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{\varepsilon_{Si} \varepsilon_0} \delta(\mathbf{r} - \mathbf{r}_I) \text{ for } Z > 0$$
(A.2)

where φ_s is the potential in the source region, Z < 0, and φ_c is the potential in the channel, Z > 0. . The $k_s = \lambda_s^{-1}$ and $k_c = \lambda_c^{-1}$ terms represent the inverse screening lengths in the source and drain regions respectively and the ionized impurity is located at position $\mathbf{r}_{\mathbf{I}}$ in the channel region.

As discussed in section 4.2.1, the potential solution will use cylindrical co-ordinates throughout. For this model we neglect any radial dependence on impurity position by assuming the impurity is located at the origin of the radial axis. This reduces the complexity of the model by allowing radial symmetry around the Z axis. Expanding the Dirac delta function of equation (A.2) in cylindrical co-ordinates, assuming that $\mathbf{R}_{I} = 0$

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{2\pi\varepsilon_{s_l}\varepsilon_0} \frac{\delta(\mathbf{R})}{R} \delta(Z - Z_I) \text{ for } Z > 0$$
(A.3)

Considering the channel potential initially, the solution will involve finding the general (homogenous) solution and the particular solution. This can be written as

$$\varphi_C = \varphi_C^G + \varphi_C^P \tag{A.4}$$

where φ_C^G is the general solution and φ_C^P the particular. The general solution for the channel potential can be found by allowing $Z \neq Z_I$ for which the RHS of equation (A.3) becomes

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = 0 \tag{A.5}$$

This is identical to the source term given by equation (A.1) and will only differ in the behaviour of the solution.

In Figure 7.1 the expected behaviour of the Z components of the potential is given. For the general solutions the expected behaviour is depicted by the red curves and the particular solution by the blue curves. The behaviour is defined by the boundary condition at large distance, Z, which is presented in more detail later in this appendix.



Figure 7.1: Expected behaviour of the Z dependent component of the potentials.

For the general solution, the potential is expected to decrease with distance from the sourcechannel interface. The particular solution will model the point charge with an increasing potential moving towards the impurity location at $Z = Z_I$.

A.2 General Solution

First solving the general (homogeneous) solution for the source and channel regions given by equations (A.1) and (A.5). Generalising between the source and channel forms, the equation can be written as

$$\nabla^2 \varphi_x - k_x^2 \varphi_x = 0 \tag{A.6}$$

A.2 General Solution

with x denoting either the source (S) and channel (C) region. Using the separation of variables technique [96] the potential solution can be expressed in cylindrical co-ordinates using a known form.

$$\varphi(R,Z) = F(R)\zeta(Z) \tag{A.7}$$

Here due to the cylindrical symmetry of this problem around the Z axis there will be no θ dependence in the solution and this component is a constant. Substituting this form of the potential back into equation (A.6) gives

$$\nabla^2 F \zeta - k_x^2 F \zeta = 0 \tag{A.8}$$

Expanding the Laplacian operator in cylindrical co-ordinates, again neglecting the term involving θ as there is no dependence on this in the solution gives

$$\left[\frac{1}{R}\frac{\partial}{\partial R}\left(R\frac{\partial}{\partial R}\right) + \frac{\partial^2}{\partial Z^2}\right]F\zeta - k_x^2F\zeta = 0$$
(A.9)

Rearranging this solution into separable components

$$\begin{bmatrix} \frac{1}{R} \frac{\partial}{\partial R} \left(R \frac{\partial}{\partial R} \right) \end{bmatrix} F \zeta = \zeta \begin{bmatrix} \frac{1}{R} \frac{\partial F}{\partial R} + \frac{\partial^2 F}{\partial R^2} \end{bmatrix}$$

$$= F \zeta \begin{bmatrix} \frac{1}{FR} \frac{\partial F}{\partial R} + \frac{1}{F} \frac{\partial^2 F}{\partial R^2} \end{bmatrix}$$
(A.10)

$$\left[\frac{\partial^2}{\partial Z^2}\right] F\zeta = F\zeta \left[\frac{1}{\zeta}\frac{\partial^2 \zeta}{\partial Z^2}\right]$$
(A.11)

Substituting these back into equation (A.9) and dividing throughout by $F\zeta$ gives

$$\frac{1}{FR}\frac{\partial F}{\partial R} + \frac{1}{F}\frac{\partial^2 F}{\partial R^2} + \frac{1}{\zeta}\frac{\partial^2 \zeta}{\partial Z^2} - k_x^2 = 0$$
(A.12)

Having now got a form of the equation in terms of R and Z only, the equation can be separated introducing a separation constant $-k^2$.

$$\frac{1}{FR}\frac{\partial F}{\partial R} + \frac{1}{F}\frac{\partial^2 F}{\partial R^2} = -k^2$$
(A.13)

$$\frac{1}{\zeta} \frac{\partial^2 \zeta}{\partial Z^2} - k_x^2 = k^2 \tag{A.14}$$

Rearranging these equations

$$R^{2}\frac{\partial^{2}F}{\partial R^{2}} + R\frac{\partial F}{\partial R} + k^{2}R^{2}F = 0$$
(A.15)

$$\frac{\partial^2 \zeta}{\partial Z^2} = \left(k^2 + k_x^2\right)\zeta \tag{A.16}$$

A general form of Bessel's differential equation [96]

$$\rho^2 \frac{\partial^2 y}{\partial \rho^2} + \rho \frac{\partial y}{\partial \rho} + \left(k^2 \rho^2 - n^2\right) y = 0$$
(A.17)

and its solution is

$$y = AJ_n(k\rho) + BN_n(k\rho) \tag{A.18}$$

Using this general form gives a solution to equation (A.15) of

$$F = AJ_0(kR) + BN_0(kR)$$
(A.19)

Solving the Z dependence from (A.16) which takes the form of a standard exponential solution

$$\zeta = \exp\left(-Z\sqrt{k^2 + k_x^2}\right) + \exp\left(Z\sqrt{k^2 + k_x^2}\right)$$
(A.20)

Substituting these solutions back into equation (A.7) using linear superposition of Bessel functions to obtain the final potential gives

$$\varphi_{x} = \int_{0}^{\infty} dk \ a(k) \left(AJ_{0}(kR) + BN_{0}(kR) \right) \left(\exp\left(-Z\sqrt{k^{2} + k_{x}^{2}} \right) + \exp\left(Z\sqrt{k^{2} + k_{x}^{2}} \right) \right)$$
(A.21)

Boundary conditions (BC) define the exact form of this solution for the source and channel regions. For the Bessel solution of the radial dependence, F(kR), the solution must be finite as $kR \rightarrow 0$. This means the Neumann function coefficient must be zero, B = 0, to satisfy this BC where the coefficient *A* will be solved later.

$$F = AJ_0(kR) \tag{A.22}$$

The expected behaviour of the Z dependence of this solution is given in Figure 7.1, which is defined from the BC that the potential must tend to zero as $Z \to \pm \infty$. Therefore for the source region $\zeta(Z \to -\infty) = 0$ and for the channel region must be $\zeta(Z \to \infty) = 0$. This yields the following solution for the source and channel

$$\zeta_{s}(Z) = \exp\left(Z\sqrt{k^{2} + k_{s}^{2}}\right) \text{ for } Z < 0$$
(A.23)

$$\zeta_C(Z) = \exp\left(-Z\sqrt{k^2 + k_C^2}\right) \text{ for } Z > 0 \tag{A.24}$$

The general solution to equation (A.6) can now be defined for the source and channel regions. Substituting equations (A.22)-(A.24) into equation (A.21) gives

$$\varphi_{s} = \int_{0}^{\infty} dk A_{s}\left(k\right) J_{0}\left(kR\right) \exp\left(Z\sqrt{k^{2}+k_{s}^{2}}\right) \text{ for } Z < 0$$
(A.25)

for the source region and

$$\varphi_C^G = \int_0^\infty dk A_C(k) J_0(kR) \exp\left(-Z\sqrt{k^2 + k_C^2}\right) \text{ for } Z > 0$$
(A.26)

for the channel region where the coefficients $A_s(k) = a_s(k)A_s$ and $A_c(k) = a_c(k)A_c$ include the coefficient of the radial Bessel function and are yet to be obtained.

A.3 Particular Solution

The particular solution for the channel potential can be found by solving equation (A.3). Simplifying the RHS of the potential equation using the following theorem, which can be found using equation (6.512.8) of Gradshteyn & Rhyzhiks book [130],

$$\frac{\delta(R)}{R} = \int_{0}^{\infty} dk \ k J_{0}(kR)$$
(A.27)

the equation becomes

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0} \int_0^\infty dk \ k J_0(kR) \delta(Z - Z_I)$$
(A.28)

Using a known form of the solution from the separation of variables method and using the general solution for the R dependence, the potential can be defined as

$$\varphi_C^P = \int_0^\infty dk \ k J_0(kR) \zeta(Z) \tag{A.29}$$

Taking the LHS of equation (A.28) separately, expanding the Laplacian operator and substituting equation (A.29) gives

$$\left[\frac{1}{R}\frac{\partial}{\partial R}\left(R\frac{\partial}{\partial R}\right) + \frac{\partial^2}{\partial Z^2} - k_c^2\right]_0^\infty dk \ kJ_0(kR)\zeta(Z)$$
(A.30)

Here we neglect the angular term as before due to the cylindrical symmetry around the Z axis. Attempting to factorise equation (A.30) into separable components yields

$$\int_{0}^{\infty} dk \ k \left[\left(\frac{\partial^2 J_0(kR)}{\partial R^2} + \frac{1}{R} \frac{\partial J_0(kR)}{\partial R} \right) \zeta + J_0(kR) \left(\frac{\partial^2 \zeta}{\partial Z^2} - k_c^2 \zeta \right) \right]$$
(A.31)

To separate the variables of this equation the terms must be independent of each other. The problem here is that we cannot factor out the Bessel function due to the first term within the square brackets. Thankfully this problem has been solved earlier. Referring to equation (A.15) of the general solution which can be re-written here, assuming that equation (A.22) still holds, as

$$\frac{\partial^2 J_0(kR)}{\partial R^2} + \frac{1}{R} \frac{\partial J_0(kR)}{\partial R} = -k^2 J_0(kR)$$
(A.32)

Using this solution in equation (A.31) and factorising gives

$$\int_{0}^{\infty} dk \ k \ J_{0}\left(kR\right) \left\{ \frac{\partial^{2} \zeta}{\partial Z^{2}} - \zeta \left(k^{2} + k_{C}^{2}\right) \right\}$$
(A.33)

Substituting this back into equation (A.28) yields

$$\int_{0}^{\infty} dk \ k \ J_{0}\left(kR\right) \left\{ \frac{\partial^{2} \zeta}{\partial Z^{2}} - \zeta \left(k^{2} + k_{c}^{2}\right) \right\} = \frac{-Q}{2\pi\varepsilon_{si}\varepsilon_{0}} \int_{0}^{\infty} dk \ k \ J_{0}\left(kR\right) \delta(Z - Z_{I})$$
(A.34)

Simplifying this equation by removing the common terms,

$$\frac{\partial^2 \zeta}{\partial Z^2} - \zeta \left(k_c^2 + k^2 \right) + \frac{Q}{2\pi \varepsilon_{si} \varepsilon_0} \delta(Z - Z_I) = 0$$
(A.35)

which provides a form which is dependent on Z only. The solution to this equation is challenging due to the singularity of the Dirac delta function at $Z = Z_I$. To work around this it is necessary to take limits around Z_I , allowing the delta function to be integrated. Integrating around $Z = Z_I - \eta$ to $Z = Z_I + \eta$ for $\eta \to 0$

$$\int_{Z_{I}-\eta}^{Z_{I}+\eta} \frac{\partial^{2} \zeta}{\partial Z^{2}} dZ - \left(k_{c}^{2} + k^{2}\right) \int_{Z_{I}-\eta}^{Z_{I}+\eta} \zeta dZ + \frac{Q}{2\pi\varepsilon_{si}\varepsilon_{0}} \int_{Z_{I}-\eta}^{Z_{I}+\eta} \delta(Z - Z_{I}) dZ = 0$$
(A.36)

The first integral can be completed to give

$$\int_{Z_{I}-\eta}^{Z_{I}+\eta} \frac{\partial^{2} \zeta}{\partial Z^{2}} dZ = \left[\frac{\partial \zeta(Z)}{\partial Z}\right]_{Z_{I}-\eta}^{Z_{I}+\eta} = \frac{\partial \zeta(Z_{I}+\eta)}{\partial Z} - \frac{\partial \zeta(Z_{I}-\eta)}{\partial Z}$$
(A.37)

The second integral can be completed using the limit on η

$$-(k_{C}^{2}+k^{2})\sum_{Z_{I}-\eta}^{Z_{I}+\eta}\zeta dZ = -(k_{C}^{2}+k^{2})[\zeta(Z)]_{Z_{I}-\eta}^{Z_{I}+\eta}$$

$$\lim_{\eta\to 0} \approx -(k_{C}^{2}+k^{2})2\eta\zeta(Z_{I}) = 0$$
(A.38)

Taking the third integral and using $\int_{a}^{b} \delta(Z - Z_{I}) dZ = 1$ when $a < Z_{I} < b$

$$\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0}\int_{Z_I-\eta}^{Z_I+\eta}\delta(Z-Z_I)dZ = \lim_{\eta\to 0}\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0}$$
(A.39)

Rewriting equation (A.36) with the result of these integrals

$$\frac{\partial \zeta (Z_I + \eta)}{\partial Z} - \frac{\partial \zeta (Z_I - \eta)}{\partial Z} = -\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0} \text{ for } Z \approx Z_I$$
(A.40)

For a correct solution of the particular case it is important to ensure that the solution is consistent for both $Z \approx Z_I$ and $Z \neq Z_I$. Looking at equation (A.35) for the case of $Z \neq Z_I$

$$\frac{\partial^2 \zeta}{\partial Z^2} = \left(k_c^2 + k^2\right) \zeta \quad \text{for } Z \neq Z_I \tag{A.41}$$

which the solution is already known (see equation (A.20)). This result for $Z \approx Z_I$, equation (A.40), suggests that there is a difference in the gradients of the solution around the impurity position, Z_I . Referring to the expected behaviour of the solution depicted in Figure 7.1, the potential is expected to change gradient around Z_I . The boundary conditions state that for $Z < Z_I$ the potential must tend to zero at $Z \rightarrow -\infty$ and for $Z > Z_I$ the potential must tend to zero at $Z \rightarrow \infty$. Using the solution to equation (A.41) with the BC's, the solution to the Z dependence of the particular solution can be found as

$$\zeta_a(Z) = \alpha \exp\left((Z - Z_I)\sqrt{k^2 + k_c^2}\right) \text{ for } Z < Z_I$$
(A.42)

$$\zeta_b(Z) = \beta \exp\left(-(Z - Z_I)\sqrt{k^2 + k_c^2}\right) \text{ for } Z > Z_I$$
(A.43)

To ensure continuity of the solution, the gradients of the potential must match at $Z = Z_I$. Setting equations (A.42)-(A.43) equal to each other and solving for the coefficients gives

$$\zeta_{a}(Z_{I} - \eta) = \zeta_{b}(Z_{I} + \eta)$$

$$\alpha \exp\left(-\eta \sqrt{k^{2} + k_{c}^{2}}\right) = \beta \exp\left(-\eta \sqrt{k^{2} + k_{c}^{2}}\right) \qquad (A.44)$$

$$\lim_{\eta \to 0} \alpha = \beta$$

A.3 Particular Solution

Using equation (A.40) to find a value for α by calculating the gradients of equations (A.42) & (A.43).

$$\frac{d\zeta_a(Z_I - \eta)}{dZ} = \sqrt{k^2 + k_c^2} \alpha \exp\left(-\eta \sqrt{k^2 + k_c^2}\right) = \lim_{\eta \to 0} \sqrt{k^2 + k_c^2} \alpha \qquad (A.45)$$

$$\frac{d\zeta_b(Z_I+\eta)}{dZ} = -\sqrt{k^2 + k_c^2} \alpha \exp\left(-\eta\sqrt{k^2 + k_c^2}\right) = \lim_{\eta \to 0} -\sqrt{k^2 + k_c^2} \alpha \qquad (A.46)$$

Substituting this into equation (A.40) yields the value for the α coefficient.

$$-\sqrt{k^{2} + k_{c}^{2}} \alpha - \sqrt{k^{2} + k_{c}^{2}} \alpha = -\frac{Q}{2\pi\varepsilon_{si}\varepsilon_{0}}$$

$$\alpha = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{1}{\sqrt{k^{2} + k_{c}^{2}}}$$
(A.47)

Completing the solution for the ζ function

$$\zeta(Z) = \frac{Q}{4\pi\varepsilon_{s_i}\varepsilon_0} \frac{1}{\sqrt{k^2 + k_c^2}} \exp\left(-\left|Z - Z_I\right|\sqrt{k^2 + k_c^2}\right)$$
(A.48)

and substituting this into equation (A.29) will provide the final particular solution of the channel potential.

$$\varphi_C^P = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_0} \int_0^\infty dk \ J_0(kR) \frac{k}{\sqrt{k^2 + k_C^2}} \exp\left(-\left|Z - Z_I\right|\sqrt{k^2 + k_C^2}\right)$$
(A.49)

The channel potential can then be found by substituting the general and particular solutions into equation (A.4) to give

$$\varphi_{C} = \int_{0}^{\infty} dk J_{0}(kR) \left\{ A_{C}(k) \exp\left(-Z\sqrt{k^{2}+k_{C}^{2}}\right) + \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2}+k_{C}^{2}}} \exp\left(-|Z-Z_{I}|\sqrt{k^{2}+k_{C}^{2}}\right) \right\}$$
(A.50)

A.4 Coefficients

The coefficients A_s and A_c of equations (A.25) and (A.50) can be found by defining the boundary conditions at Z = 0, the source – channel interface. The static electric field boundary conditions apply at this interface, defining that the electric field must be continuous perpendicular to the interface and that the normal electric flux be continuous across the interface.

$$\frac{\partial \varphi_{S}(Z=0)}{\partial \mathbf{R}} = \frac{\partial \varphi_{C}(Z=0)}{\partial \mathbf{R}}$$
(A.51)

$$\varepsilon_{si}\varepsilon_0 \frac{\partial \varphi_s \left(Z=0\right)}{\partial Z} = \varepsilon_{si}\varepsilon_0 \frac{\partial \varphi_c \left(Z=0\right)}{\partial Z}$$
(A.52)

Finding the derivate of the potentials for the first boundary condition

$$\frac{\partial \varphi_{S}}{\partial R} = \int_{0}^{\infty} dk A_{S} \left(-k J_{1} \left(k R \right) \right)$$
(A.53)

$$\frac{\partial \varphi_{C}}{\partial R} = \int_{0}^{\infty} dk \left(-kJ_{1}\left(kR\right)\right) \left\{ A_{C} + \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{C}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{C}^{2}}\right) \right\}$$
(A.54)

Setting equation (A.53) equal to equation (A.54) and re-arranging gives

$$A_{s} - A_{c} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$
(A.55)

The derivates for the second boundary condition, equation (A.52) are

$$\frac{\partial \varphi_s}{\partial Z} = \int_0^\infty dk A_s J_0(kR) \sqrt{k^2 + k_s^2}$$
(A.56)

$$\frac{\partial \varphi_{c}}{\partial Z} = \int_{0}^{\infty} dk J_{0} \left(kR \right) \left\{ A_{c} \left(-\sqrt{k^{2} + k_{c}^{2}} \right) + \frac{Q}{4\pi \varepsilon_{si} \varepsilon_{0}} k \exp \left(-Z_{I} \sqrt{k^{2} + k_{c}^{2}} \right) \right\}$$
(A.57)

Substituting these into the boundary condition and after some manipulation yields

A.5 Single Interface Solution

$$A_{s} \frac{\sqrt{k^{2} + k_{s}^{2}}}{\sqrt{k^{2} + k_{c}^{2}}} + A_{c} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$
(A.58)

It is now a matter of solving the two simultaneous equations to find the values for the coefficients. The source coefficient can be easily found by adding equation (A.55) to (A.58) giving

$$A_{s} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{2k}{\sqrt{k^{2} + k_{c}^{2}} + \sqrt{k^{2} + k_{s}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right)$$
(A.59)

To find the channel coefficient, the two equations must be subtracted

$$A_{C} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-Z_{I}\sqrt{k^{2} + k_{c}^{2}}\right) \frac{\sqrt{k^{2} + k_{c}^{2}} - \sqrt{k^{2} + k_{s}^{2}}}{\sqrt{k^{2} + k_{c}^{2}} + \sqrt{k^{2} + k_{s}^{2}}}$$
(A.60)

A.5 Single Interface Solution

Substituting the coefficients back into the solutions for the potential gives the final form for the source and channel potentials.

$$\varphi_{s} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0} \left(kR\right) \frac{2k}{K_{c} + K_{s}} \exp\left(-Z_{I}K_{c}\right) \exp\left(ZK_{s}\right)$$
(A.61)

$$\varphi_{C} = \frac{Q}{4\pi\varepsilon_{s_{I}}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0}(kR) \frac{k}{K_{C}} \left\{ \exp\left(-\left|Z - Z_{I}\right| K_{C}\right) + \frac{K_{C} - K_{s}}{K_{C} + K_{s}} \exp\left(-\left(Z + Z_{I}\right) K_{C}\right) \right\}$$
(A.62)

where

$$K_c = \sqrt{k^2 + k_c^2} \tag{A.63}$$

$$K_s = \sqrt{k^2 + k_s^2} \tag{A.64}$$

A.5 Single Interface Solution

Remembering here that the variable k is purely a separation constant introduced in the solution of the partial differential equation and should not be confused with the carrier wave vector variable of the same letter.

Appendix B Double Interface Potential Calculation

In this appendix the double interface potential solution will be presented in more detail than is provided in chapter 4.3.1. Due to the similarities in the procedure and the solution, the previous appendix will be referenced for part of the solution method. Here a potential solution is found for a single ionized impurity between two highly-doped regions.

B.1 Linearized Poisson's Equation

Starting with Poisson's equation for the source, channel and drain regions which has been linearized using the Thomas-Fermi approximation in section 4.3.1

$$\nabla^2 \varphi_s - k_s^2 \varphi_s = 0 \quad \text{for } Z < 0 \tag{B.1}$$

$$\nabla^2 \varphi_C - k_C^2 \varphi_C = -\frac{Q}{\varepsilon_{Si} \varepsilon_0} \delta(\mathbf{r} - \mathbf{r}_I) \text{ for } 0 < Z < L_C$$
(B.2)

$$\nabla^2 \varphi_D - k_D^2 \varphi_D = 0 \quad \text{for } Z > L_C \tag{B.3}$$

Expanding the Dirac delta function of equation (B.2) into cylindrical co-ordinates, allowing $\mathbf{R}_{I} = 0$

$$\nabla^2 \varphi_c - k_c^2 \varphi_c = -\frac{Q}{2\pi\varepsilon_{si}\varepsilon_0} \frac{\delta(\mathbf{R})}{R} \delta(Z - Z_I) \text{ for } Z > 0$$
(B.4)

The solution of the channel potential will involve solving a general and a particular solution. In this model there will be two general solutions corresponding to the regions on either side of the point charge in the channel. Expressing the channel potential in this form as

$$\varphi_{c} = \varphi_{c}^{G1} + \varphi_{c}^{P} + \varphi_{c}^{G2}$$
(B.5)

where the superscripts G1 denote the source end of the channel, G2 the drain end of the channel and P the particular solution of the point charge. The first step is to find the general solution which for the channel potential is found by allowing the right-hand side (RHS) of equation (B.4) to equal zero. As this form is identical to that of the source and drain regions a universal form can be written

$$\nabla^2 \varphi_x - k_x^2 \varphi_x = 0 \tag{B.6}$$

where x denotes either the source (S), channel (C), or drain (D) region. This solution is found for the single interface case in appendix section A.3 and can be used here, again using the same radial boundary condition for the Bessel function.

$$\varphi_x = \int_0^\infty dk \ a(k) J_0(kR) \left(\exp\left(-Z\sqrt{k^2 + k_x^2}\right) + \exp\left(Z\sqrt{k^2 + k_x^2}\right) \right)$$
(B.7)

Although this is a universal form for the solution each region has different behaviour as depicted in Figure 7.2. This figure represents the expected behaviour of the general (red curve) and particular (blue curve) Z dependent components of the potential in each of the regions.



Figure 7.2: Expected behaviour of the Z component of the potential with the source region Z < 0, the drain region $Z > L_c$ and the impurity at $Z = Z_l$.

From Figure 7.2 the behaviour of the Z dependence of the potential in the source and drain is depicted by the red curve in the Z < 0 and $Z > L_C$ regions respectively. Applying the boundary condition such that the potential tends to zero at large Z distance, $\lim_{Z\to\infty} \varphi_S = 0$ and $\lim_{Z\to\infty} \varphi_D = 0$, gives the following solutions for the source and drain.

$$\varphi_{s} = \int_{0}^{\infty} dk A_{s}(k) J_{0}(kR) \exp\left(Z\sqrt{k^{2}+k_{s}^{2}}\right) \text{ for } Z < 0$$
(B.8)

$$\varphi_D = \int_0^\infty dk A_D(k) J_0(kR) \exp\left(-Z\sqrt{k^2 + k_D^2}\right) \text{ for } Z > 0$$
(B.9)

B.2 Solution of Coefficients

For the channel potential there are two general solutions describing the regions on either side of the point charge. Referring to the red curves in Figure 7.2 the Z behaviour is expected to decrease from the source interface and increase towards the drain interface. This corresponds with the limiting boundary condition that the potential must tend to zero at large distances, $\lim_{Z \gg 0} \varphi_C^{G1} = 0$ and $\lim_{Z \ll L_C} \varphi_C^{G2} = 0$. Applying these boundary conditions to equation (B.7) provides the general solutions for the channel potential.

for the channel potential.

$$\varphi_{C}^{G1} = \int_{0}^{\infty} dk \ kA_{C}(k) J_{0}(kR) \exp\left(-Z\sqrt{k^{2} + k_{C}^{2}}\right)$$
(B.10)

$$\varphi_C^{G2} = \int_0^\infty dk \ kB_C(k) J_0(kR) \exp\left(Z\sqrt{k^2 + k_C^2}\right)$$
(B.11)

The particular solution for the double interface model is identical to that of the single interface model allowing the solution to be repeated here. Using equation (A.49)

$$\varphi_{C}^{P} = \frac{Q}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}} \int_{0}^{\infty} dk \ J_{0}\left(kR\right) \frac{k}{\sqrt{k^{2} + k_{C}^{2}}} \exp\left(-|Z - Z_{I}|\sqrt{k^{2} + k_{C}^{2}}\right)$$
(B.12)

Substituting the general and particular solutions for the channel potential into equation (B.5) and re-arranging gives the following form.

$$\varphi_{C} = \frac{Q}{4\pi\varepsilon_{s_{l}}\varepsilon_{0}} \int_{0}^{\infty} dk \ J_{0}(kR) \frac{k}{\sqrt{k^{2} + k_{c}^{2}}} \exp\left(-|Z - Z_{I}|\sqrt{k^{2} + k_{c}^{2}}\right) + \int_{0}^{\infty} dk J_{0}(kR) \left\{A_{C}(k) \exp\left(-Z\sqrt{k^{2} + k_{c}^{2}}\right) + B_{C}(k) \exp\left(Z\sqrt{k^{2} + k_{c}^{2}}\right)\right\}$$
(B.13)

B.2 Solution of Coefficients

To find the coefficients for the source, drain and channel potentials given by equations (B.8), (B.9) and (B.13) respectively, the electric field boundary conditions must be applied at both the Z = 0 and $Z = L_c$ interfaces. The boundary conditions for the source interface are

$$\frac{\partial \varphi_{S}(Z=0)}{\partial \mathbf{R}} = \frac{\partial \varphi_{C}(Z=0)}{\partial \mathbf{R}}$$
(B.14)

B.2 Solution of Coefficients

$$\varepsilon_{si}\varepsilon_0 \frac{\partial \varphi_s \left(Z=0\right)}{\partial Z} = \varepsilon_{si}\varepsilon_0 \frac{\partial \varphi_c \left(Z=0\right)}{\partial Z}$$
(B.15)

and for the drain interface are

$$\frac{\partial \varphi_{c} \left(Z = L_{c} \right)}{\partial \mathbf{R}} = \frac{\partial \varphi_{D} \left(Z = L_{c} \right)}{\partial \mathbf{R}}$$
(B.16)

$$\varepsilon_{si}\varepsilon_{0}\frac{\partial\varphi_{C}(Z=L_{C})}{\partial Z}=\varepsilon_{si}\varepsilon_{0}\frac{\partial\varphi_{D}(Z=L_{C})}{\partial Z}$$
(B.17)

Using the following expressions allow the potentials to be simplified. These simplified terms will be used throughout the remainder of this appendix.

$$K_c = \sqrt{k^2 + k_c^2} \tag{B.18}$$

$$K_s = \sqrt{k^2 + k_s^2} \tag{B.19}$$

$$K_D = \sqrt{k^2 + k_D^2} \tag{B.20}$$

Starting with the boundary condition at the source-channel interface and finding the derivates of the potentials:

$$\frac{\partial \varphi_s \left(Z=0\right)}{\partial R} = \int_0^\infty dk \ A_s \left(-kJ_1\left(kR\right)\right) \tag{B.21}$$

$$\frac{\partial \varphi_C \left(Z=0\right)}{\partial R} = \frac{Q}{4\pi \varepsilon_{si} \varepsilon_0} \int_0^\infty dk \left(-kJ_1 \left(kR\right)\right) \left\{\frac{k}{K_C} \exp\left(-Z_I K_C\right) + A_C + B_C\right\}$$
(B.22)

Setting these derivatives equal to one another and re-arranging gives

$$A_{s} - A_{c} - B_{c} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{c}} \exp\left(-Z_{I}K_{c}\right)$$
(B.23)

For the second source-channel interface electric field BC:

B.2 Solution of Coefficients

$$\frac{\partial \varphi_{S}(Z=0)}{\partial Z} = \int_{0}^{\infty} dk A_{S} J_{0}(kR) K_{S}$$
(B.24)

$$\frac{\partial \varphi_C \left(Z=0 \right)}{\partial Z} = \frac{Q}{4\pi \varepsilon_{s_i} \varepsilon_0} \int_0^\infty dk \ J_0 \left(kR \right) \left\{ k \exp\left(-Z_I K_C \right) - A_C K_C + B_C K_C \right\}$$
(B.25)

Substituting these into equation (B.15) and manipulating yields

$$A_{s} \frac{K_{s}}{K_{c}} + A_{c} - B_{c} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{c}} \exp\left(-Z_{I}K_{c}\right)$$
(B.26)

For the channel-drain interface electric flux BC:

$$\frac{\partial \varphi_{C} \left(Z = L_{C} \right)}{\partial R} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \int_{0}^{\infty} dk \left(-kJ_{1} \left(kR \right) \right) \left\{ \frac{k}{K_{C}} \exp\left(-\left| L_{C} - Z_{I} \right| K_{C} \right) +A_{C} \exp\left(-L_{C} K_{C} \right) + B_{C} \exp\left(L_{C} K_{C} \right) \right\}$$
(B.27)

$$\frac{\partial \varphi_D \left(Z = L_C \right)}{\partial R} = \int_0^\infty dk \ A_D \left(-kJ_1 \left(kR \right) \right) \exp \left(-L_C K_D \right)$$
(B.28)

Using these equation to find the electric field BC at the drain interface:

$$A_{D} \exp(-L_{C}K_{D}) - A_{C} \exp(-L_{C}K_{C}) - B_{C} \exp(L_{C}K_{C})$$

$$= \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{C}} \exp(-(L_{C}-Z_{I})K_{C})$$
(B.29)

For the channel-drain interface electric field BC:

$$\frac{\partial \varphi_{c} \left(Z = L_{c} \right)}{\partial Z} = \frac{Q}{4\pi \varepsilon_{si} \varepsilon_{0}} \int_{0}^{\infty} dk \ J_{0} \left(kR \right) \left\{ \left(-k \right) \exp \left(-\left(L_{c} - Z_{I} \right) K_{c} \right) +A_{c} \left(-K_{c} \right) \exp \left(-L_{c} K_{c} \right) +B_{c} K_{c} \exp \left(L_{c} K_{c} \right) \right\}$$
(B.30)

$$\frac{\partial \varphi_D \left(Z = L_C \right)}{\partial Z} = \int_0^\infty dk A_D J_0 \left(kR \right) \left(-K_D \right) \exp\left(-L_C K_D \right)$$
(B.31)

The final BC is then

$$A_{D} \frac{K_{D}}{K_{C}} \exp(-L_{C} K_{D}) - A_{C} \exp(-L_{C} K_{C}) + B_{C} \exp(L_{C} K_{C})$$

$$= \frac{Q}{4\pi\varepsilon_{st}\varepsilon_{0}} \frac{k}{K_{C}} \exp(-(L_{C} - Z_{I})K_{C})$$
(B.32)

Introducing the following expressions to simplify the RHS of the resulting simultaneous equations (B.23), (B.26), (B.29) and (B.32).

$$\gamma = \frac{Q}{4\pi\varepsilon_{s_l}\varepsilon_0} \frac{k}{K_c} \exp\left(-Z_I K_c\right)$$
(B.33)

$$\gamma^{L} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{c}} \exp\left(-\left(L_{c} - Z_{I}\right)K_{c}\right)$$
(B.34)

Using these four simultaneous equations to find the coefficients is too complex for a basic elimination method. Instead an inverse matrix method is employed to solve the coefficients, e.g. arranging the simultaneous equations into the matrices $\mathbf{A}\mathbf{X} = \mathbf{b}$ presented as equation (B.35), the solution for the coefficients can be found from $\mathbf{X} = \mathbf{A}^{-1}\mathbf{b}$.

$$\begin{bmatrix} 1 & -1 & -1 & 0 \\ 0 & -\exp(-L_C K_C) & -\exp(L_C K_C) & \exp(-L_C K_D) \\ K_s & K_C & -K_C & 0 \\ 0 & -K_C \exp(-L_C K_C) & K_C \exp(L_C K_C) & K_D \exp(-L_C K_D) \end{bmatrix} \begin{bmatrix} A_s \\ A_c \\ B_c \\ A_D \end{bmatrix} = \begin{bmatrix} \gamma \\ \gamma' \\ \gamma' \\ \gamma' \end{bmatrix}$$
(B.35)

Computing the inverse of the first matrix on the LHS by hand is very tedious, instead Wolfram's Mathematica [131] application has been utilised to solve the coefficients. Following the computation in Mathematica to solve the matrix problem above and after some manipulation the coefficients can be written as follows.

$$A_{s} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} 2k \exp(-Z_{I}K_{C})A_{m}$$
(B.36)

$$A_{C} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{C}} (K_{C} - K_{s}) \exp(-Z_{I}K_{C}) A_{m}$$
(B.37)

$$A_{m} = \frac{K_{C} \left(\exp(2L_{C}K_{C}) + \exp(2Z_{I}K_{C}) \right) + K_{D} \left(\exp(2L_{C}K_{C}) - \exp(2Z_{I}K_{C}) \right)}{\left(K_{C}^{2} + K_{S}K_{D} \right) \left(\exp(2L_{C}K_{C}) - 1 \right) + K_{C} \left(K_{S} + K_{D} \right) \left(\exp(2L_{C}K_{C}) + 1 \right)}$$
(B.38)

for the source interaction coefficients and

$$B_{C} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{k}{K_{C}} \left(K_{C} - K_{D}\right) \exp\left(-Z_{I}K_{C}\right) A_{n}$$
(B.39)

$$A_{D} = \frac{Q}{4\pi\varepsilon_{si}\varepsilon_{0}} 2k \exp\left(-Z_{I}K_{C}\right) \exp\left(L_{C}\left(K_{C}+K_{D}\right)\right)A_{n}$$
(B.40)

$$A_{n} = \frac{K_{C} \left(\exp(2Z_{I}K_{C}) + 1 \right) + K_{S} \left(\exp(2Z_{I}K_{C}) - 1 \right)}{\left(K_{C}^{2} + K_{S}K_{D} \right) \left(\exp(2L_{C}K_{C}) - 1 \right) + K_{C} \left(K_{S} + K_{D} \right) \left(\exp(2L_{C}K_{C}) + 1 \right)}$$
(B.41)

for the drain interaction coefficients.

B.1 Double Interface Solution

Substituting the coefficients found in the previous section into the potentials from section B.1, gives the finalised forms.

$$\varphi_{S} = \frac{Q}{4\pi\varepsilon_{Si}\varepsilon_{0}} \int_{0}^{\infty} dk \, 2k \, J_{0}(kR) \exp(ZK_{S}) \exp(-Z_{I}K_{C}) A_{m}$$
(B.42)

$$\varphi_{C} = \frac{Q}{4\pi\varepsilon_{S_{I}}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0}(kR) \frac{k}{K_{C}} \left\{ \exp\left(-\left|Z - Z_{I}\right|K_{C}\right) + \left(K_{C} - K_{S}\right) \exp\left(-\left(Z + Z_{I}\right)K_{C}\right)A_{m} + \left(K_{C} - K_{D}\right) \exp\left(-\left(Z - Z_{I}\right)K_{C}\right)A_{n} \right\}$$
(B.43)

$$\varphi_{D} = \frac{Q}{4\pi\varepsilon_{Si}\varepsilon_{0}} \int_{0}^{\infty} dk \, 2k J_{0}\left(kR\right) \exp\left(-\left(Z - L_{C}\right)K_{D}\right) \exp\left(-\left(Z_{I} - L_{C}\right)K_{C}\right)A_{n} \qquad (B.44)$$

Here the coefficients A_m and A_n are given by the equations (B.38) and (B.41) respectively, and the terms K_x are given by equations (B.18)-(B.20). Again, k is introduced as a separation variable and should not be confused with the carrier wave vector of the same symbol.

Appendix C Full Single Interface Scattering Model

This appendix will present the scattering model for the full single interface model which is used in section 4.2.5.2 to compare the mobility of the full and strongly screened models. This model is not used extensively in this research work but is included in the appendix for future reference.

The full or complete single interface model is based on the potential found in section 4.2.1 with the accompanying detailed calculation in Appendix A. Unlike the strongly screened model which assumes a metallic source and is used for scattering calculations in this PhD work, the model developed here will include the exact interaction of the doped source region of semiconductor.

Here an identical approach to that of Chapter 5 to calculate a scattering model will be used. A spherically-symmetric scattering potential using the Z-aligned simplification discussed in section 5.2.3 is assumed in the calculation which follows Fermi's Golden Rule approach assuming spherical, non-parabolic bands. Initially the scattering matrix element will be calculated from the scattering potential in section C.1. The scattering probability and scattering rate are then obtained in section C.2.

C.1 Matrix Element

The matrix element is found by completing the Fourier transform of the interaction potential over cylindrical co-ordinates, equation (5.6).

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{-\infty}^{\infty} dZ \int_{0}^{2\pi} d\phi \int_{0}^{\infty} d\mathbf{R} \ U_{S}(\mathbf{R}, Z) \exp(-i\mathbf{q}_{\perp} \cdot \mathbf{R}) \exp(-iq_{Z}Z)$$
(C.1)

For the complete model the full interaction potential calculated in detail in Appendix A will be used and is repeated here as equations (C.2)-(C.3).

$$\varphi_{S} = \frac{Q}{4\pi\varepsilon_{SI}\varepsilon_{0}} \int_{0}^{\infty} dk J_{0} \left(kR\right) \frac{2k}{K_{C} + K_{S}} \exp\left(-Z_{I}K_{C}\right) \exp\left(ZK_{S}\right)$$
(C.2)

$$\varphi_{C} = \frac{Q}{4\pi\varepsilon_{s_{i}}\varepsilon_{0}} \int_{0}^{\infty} dk \ J_{0}\left(kR\right) \frac{k}{K_{C}} \left\{ \exp\left(-\left|Z - Z_{I}\right|K_{C}\right) + \frac{K_{C} - K_{s}}{K_{C} + K_{s}} \exp\left(-\left(Z + Z_{I}\right)K_{C}\right) \right\}$$
(C.3)

where

$$K_c = \sqrt{k^2 + k_c^2} \tag{C.4}$$

$$K_s = \sqrt{k^2 + k_s^2} \tag{C.5}$$

It should be noted that in equations (C.2)-(C.5) the variable k is not the carrier wave vector used in general notation but is in fact a separation variable introduced in the solution of the potential. Using the following form, the source and drain interaction potentials can be substituted into equation (5.6).

$$U_{s}(\mathbf{R},Z) = eV(\mathbf{R},Z) = e(\theta(-Z)\varphi_{s} + \theta(Z)\varphi_{c})$$
(C.6)

Here *e* is the electronic charge and θ is the unit step function. Re-arranging the form of equation (5.6) taking care with the unit step functions of equation (C.6) gives

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{1}{\Omega} \int_{0}^{2\pi} d\phi \int_{0}^{\infty} d\mathbf{R} \exp\left(-i\mathbf{q}_{\perp} \cdot \mathbf{R}\right) \left\{ \int_{-\infty}^{0} dZ \ e\varphi_{S} \exp\left(-iq_{Z}Z\right) + \int_{0}^{\infty} dZ \ e\varphi_{C} \exp\left(-iq_{Z}Z\right) \right\} \quad (C.7)$$

where \mathbf{q}_{\perp} is the momentum transfer in the plane perpendicular to the Z-axis and q_Z is the momentum transfer along the Z-axis. After some lengthy integration and algebraic manipulation, the Fourier transform of the potentials is found to be

$$\mathbf{H}_{\mathbf{k}'\mathbf{k}} = \frac{eQ}{4\pi\varepsilon_{si}\varepsilon_{0}} \frac{2\pi}{\Omega} \Biggl[\frac{2}{K_{c} + K_{s}} \exp(-Z_{I}K_{c}) \frac{K_{s} + iq_{z}}{\mathbf{q}_{\perp}^{2} + q_{z}^{2} + k_{s}^{2}} + \frac{1}{K_{c}} \Biggl(\exp(-iq_{z}Z_{I}) \frac{2K_{c}}{\mathbf{q}_{\perp}^{2} + q_{z}^{2} + k_{s}^{2}} - \exp(-Z_{I}K_{c}) \Biggl(\frac{K_{c} + iq_{z}}{\mathbf{q}_{\perp}^{2} + q_{z}^{2} + k_{s}^{2}} - \Biggl(\frac{K_{c} - K_{s}}{K_{c} + K_{s}} \Biggr) \frac{K_{c} - iq_{z}}{\mathbf{q}_{\perp}^{2} + q_{z}^{2} + k_{s}^{2}} \Biggr) \Biggr]$$
(C.8)

where the terms K_x have now become

$$K_C = \sqrt{\mathbf{q}_\perp^2 + k_C^2} \tag{C.9}$$

$$K_s = \sqrt{\mathbf{q}_\perp^2 + k_s^2} \tag{C.10}$$
C.2 Scattering Rate

Before this scattering matrix element can be used in Fermi's Golden Rule, the magnitude-squared must be found. After some extensive manipulation, the solution is found as

$$\left|\mathbf{H}_{\mathbf{k},\mathbf{k}'}\right|^{2} = \left(\frac{eQ}{4\pi\varepsilon_{Si}\varepsilon_{0}}\right)^{2} \left(\frac{2\pi}{\Omega}\right)^{2} \frac{1}{\left(\mathbf{q}_{\perp}^{2} + q_{Z}^{2} + k_{C}^{2}\right)^{2}} f_{sgl}\left(k_{C}, Z_{I}\right)$$
(C.11)

where the f_{sgl} function can be expressed as

$$f_{sgl}(k_C, Z_I) = 4\left(\cos\left(q_Z Z_I\right) A_{sgl} + \sin\left(q_Z Z_I\right) B_{sgl}\right) + \exp\left(-2Z_I K_C\right) \left(C_{sgl} + D_{sgl}\right)$$
(C.12)

$$A_{sgl} = \cos(q_{z}Z_{I}) - \exp(-Z_{I}K_{C}) \times \left(1 - \frac{1}{K_{C} + K_{s}} \left((K_{C} - K_{s}) + 2K_{s} \left(\frac{q_{z}^{2} + K_{C}^{2}}{q_{z}^{2} + K_{s}^{2}} \right) \right) \right)$$
(C.13)

$$B_{sgl} = \sin(q_{Z}Z_{I}) + \exp(-Z_{I}K_{C})\frac{q_{Z}}{K_{C}}$$

$$\times \left(1 + \frac{1}{K_{C} + K_{S}} \left((K_{C} - K_{S}) - 2K_{C} \left(\frac{q_{Z}^{2} + K_{C}^{2}}{q_{Z}^{2} + K_{S}^{2}} \right) \right) \right)$$
(C.14)

$$C_{sgl} = 1 - \frac{1}{K_{c} + K_{s}} \left(2(K_{c} - K_{s}) - \frac{(K_{c} - K_{s})^{2}}{K_{c} + K_{s}} + 4K_{s} \left(\frac{q_{z}^{2} + K_{c}^{2}}{q_{z}^{2} + K_{s}^{2}} \right) \left(1 - \frac{K_{s}}{K_{c} + K_{s}} \left(\frac{q_{z}^{2} + K_{c}^{2}}{q_{z}^{2} + K_{s}^{2}} \right) - \left(\frac{K_{c} - K_{s}}{K_{c} + K_{s}} \right) \right) \right)$$
(C.15)

$$D_{sgl} = \frac{q_Z^2}{K_C^2} + \frac{q_Z^2}{K_C(K_C + K_S)} \left(\frac{2(K_C - K_S)}{K_C} + \frac{(K_C - K_S)^2}{K_C(K_C + K_S)} - 4\left(\frac{q_Z^2 + K_C^2}{q_Z^2 + K_S^2}\right) \left(1 - \frac{K_C}{K_C + K_S} \left(\frac{q_Z^2 + K_C^2}{q_Z^2 + K_S^2}\right) + (K_C - K_S)\right) \right)$$
(C.16)

C.2 Scattering Rate

The scattering rate is found using Fermi's Golden Rule approach given by equation (5.9), which describes the probability of scattering from a state \mathbf{k} to a state \mathbf{k}' for an elastic interaction.

C.2 Scattering Rate

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} |\mathbf{H}_{\mathbf{k}'\mathbf{k}}|^2 \,\delta\big(E(\mathbf{k}') - E(\mathbf{k})\big) \tag{C.17}$$

Substitution of the scattering matrix element into the Golden Rule gives

$$P(\mathbf{k},\mathbf{k}') = \frac{2\pi}{\hbar} \left(\frac{eQ}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 \left(\frac{2\pi}{\Omega}\right)^2 \frac{\delta(E(\mathbf{k}') - E(\mathbf{k}))}{\left(\mathbf{q}_{\perp}^2 + q_Z^2 + k_C^2\right)^2} f_{sgl}(k_C, Z_I)$$
(C.18)

where f_{sgl} is given by equations (C.12)-(C.16). The scattering and momentum relaxation rates (for an elastic interaction) can be evaluated from the scattering probability using equations (5.14) and (5.15) respectively.

$$\Gamma(\mathbf{k}) = \sum_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') = N_k \int_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') d\mathbf{k}'$$
(C.19)

$$\frac{1}{\tau_m(\mathbf{k}')} = \sum_{k'} P(\mathbf{k}, \mathbf{k}') (1 - \cos\theta) = N_k \int_{\mathbf{k}'} P(\mathbf{k}, \mathbf{k}') (1 - \cos\theta) d\mathbf{k}'$$
(C.20)

$$N_k = \frac{\Omega}{\left(2\pi\right)^3} \tag{C.21}$$

The scattering can be calculated to give

$$\Gamma(\mathbf{k}) = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2\pi m^* k}{\hbar^3} (1 + 2\alpha E) \int_0^{\pi} d\theta \frac{\sin\theta}{\left(\mathbf{q}^2 + k_c^2\right)^2} f_{sgl}\left(k_c, Z_I\right)$$
(C.22)

where the final integral is left to numerical integration. The momentum relaxation rate can be found by substituting the scattering weighting term into the integral.

$$\frac{1}{\tau_m(\mathbf{k})} = \left(\frac{Ze^2}{4\pi\varepsilon_{si}\varepsilon_0}\right)^2 N_I \frac{2\pi m^* k}{\hbar^3} (1 + 2\alpha E) \int_0^{\pi} d\theta \frac{\sin\theta(1 - \cos\theta)}{\left(\mathbf{q}^2 + k_c^2\right)^2} f_{sgl}\left(k_c, Z_I\right)$$
(C.23)

References

- [1] "International Technology Roadmap for Semiconductors," <u>http://www.itrs.net/home.html</u>, 2008.
- [2] F. Arnaud, J. Liu, Y. M. Lee, K. Y. Lim, S. Kohler, J. Chen, B. K. Moon, C. W. Lai, M. Lipinski, L. Sang, F. Guarin, C. Hobbs, P. Ferreira, K. Ohuchi, J. Li, H. Zhuang, P. Mora, Q. Zhang, D. R. Nair, D. H. Lee, K. K. Chan, S. Satadru, S. Yang, J. Koshy, W. Hayter, M. Zaleski, D. V. Coolbaugh, H. W. Kim, Y. C. Ee, J. Sudijono, A. Thean, M. Sherony, S. Samavedam, M. Khare, C. Goldberg, and A. Steegen, "32nm general purpose bulk CMOS technology for high performance applications at low voltage," in *Electron Devices Meeting*, 2008. *IEDM* 2008. *IEEE International*, 2008, pp. 1-4.
- [3] C. H. Diaz, K. Goto, H. T. Huang, Y. Yasuda, C. P. Tsao, T. T. Chu, W. T. Lu, V. Chang, Y. T. Hou, Y. S. Chao, P. F. Hsu, C. L. Chen, K. C. Lin, J. A. Ng, W. C. Yang, C. H. Chen, Y. H. Peng, C. J. Chen, C. C. Chen, M. H. Yu, L. Y. Yeh, K. S. You, K. S. Chen, K. B. Thei, C. H. Lee, S. H. Yang, J. Y. Cheng, K. T. Huang, J. J. Liaw, Y. Ku, S. M. Jang, H. Chuang, and M. S. Liang, "32nm gate-first high-k/metal-gate technology for high performance low power applications," in *Electron Devices Meeting, 2008. IEDM 2008. IEEE International*, 2008, pp. 1-4.
- [4] S. Natarajan, M. Armstrong, M. Bost, R. Brain, M. Brazier, C. H. Chang, V. Chikarmane, M. Childs, H. Deshpande, K. Dev, G. Ding, T. Ghani, O. Golonzka, W. Han, J. He, R. Heussner, R. James, I. Jin, C. Kenyon, S. Klopcic, S. H. Lee, M. Liu, S. Lodha, B. McFadden, A. Murthy, L. Neiberg, J. Neirynck, P. Packan, S. Pae, C. Parker, C. Pelto, L. Pipes, J. Sebastian, J. Seiple, B. Sell, S. Sivakumar, B. Song, K. Tone, T. Troeger, C. Weber, M. Yang, A. Yeoh, and K. Zhang, "A 32nm logic technology featuring 2nd-generation high-k + metal-gate transistors, enhanced channel strain and 0.171µm² SRAM cell size in a 291Mb array," in *Electron Devices Meeting, 2008. IEDM 2008. IEEE International*, 2008, pp. 1-3.
- [5] L. Kang, K. Onishi, Y. Jeon, L. Byoung Hun, C. Kang, Q. Wen-Jie, R. Nieh, S. Gopalan, R. Choi, and J. C. Lee, "MOSFET devices with polysilicon on single-layer HfO₂ high-K dielectrics," in *Electron Devices Meeting*, 2000. *IEDM Technical Digest. International*, 2000, pp. 35-38.
- [6] A. I. Kingon, J.-P. Maria, and S. K. Streiffer, "Alternative dielectrics to silicon dioxide for memory and logic devices," *Nature*, vol. 406, pp. 1032-1038, 2000.
- [7] R. Chau, J. Brask, S. Datta, G. Dewey, M. Doczy, B. Doyle, J. Kavalieros, B. Jin, M. Metz, A. Majumdar, and M. Radosavljevic, "Application of high-κ gate dielectrics and metal gate electrodes to enable silicon and non-silicon logic nanotechnology," *Microelectronic Engineering*, vol. 80, pp. 1-6, 2005.
- [8] M. Casse, L. Thevenod, B. Guillaumot, L. Tosti, F. Martin, J. Mitard, O. Weber, F. Andrieu, T. Ernst, G. Reimbold, T. Billon, M. Mouis, and F. Boulanger, "Carrier transport in HfO₂ metal gate MOSFETs: physical insight into critical parameters," *Electron Devices, IEEE Transactions on*, vol. 53, pp. 759-768, 2006.
- [9] H. R. Harris, S. E. Thompson, S. Krishnan, P. Kirsch, P. Majhi, C. E. Smith, M. M. Hussain, G. Sun, H. Adhikari, S. Suthram, B. H. Lee, H. H. Tseng, and R. Jammy, "Flexible, simplified CMOS on Si(110) with metal gate / high k for HP and LSTP," in *Electron Devices Meeting*, 2007. *IEDM 2007. IEEE International*, 2007, pp. 57-60.

- K. Mistry, C. Allen, C. Auth, B. Beattie, D. Bergstrom, M. Bost, M. Brazier, M. Buehler, A. Cappellani, R. Chau, C. H. Choi, G. Ding, K. Fischer, T. Ghani, R. Grover, W. Han, D. Hanken, M. Hattendorf, J. He, J. Hicks, R. Huessner, D. Ingerly, P. Jain, R. James, L. Jong, S. Joshi, C. Kenyon, K. Kuhn, K. Lee, H. Liu, J. Maiz, B. McLntyre, P. Moon, J. Neirynck, S. Pae, C. Parker, D. Parsons, C. Prasad, L. Pipes, M. Prince, P. Ranade, T. Reynolds, J. Sandford, L. Shifren, J. Sebastian, J. Seiple, D. Simon, S. Sivakumar, P. Smith, C. Thomas, T. Troeger, P. Vandervoorn, S. Williams, and K. Zawadzki, "A 45nm Logic Technology with High-k+Metal Gate Transistors, Strained Silicon, 9 Cu Interconnect Layers, 193nm Dry Patterning, and 100% Pb-free Packaging," in *Electron Devices Meeting, 2007. IEDM 2007. IEEE International*, 2007, pp. 247-250.
- [11] K. Rim, S. Narasimha, M. Longstreet, A. Mocuta, and J. Cai, "Low field mobility characteristics of sub-100 nm unstrained and strained Si MOSFETs," in *Electron Devices Meeting*, 2002. *IEDM* '02. *Digest. International*, 2002, pp. 43-46.
- [12] M. Ieong, B. Doris, J. Kedzierski, K. Rim, and M. Yang, "Silicon Device Scaling to the Sub-10-nm Regime," *Science*, vol. 306, pp. 2057-2060, December 17, 2004 2004.
- [13] S. S. Chung, E. R. Hsieh, D. C. Huang, C. S. Lai, C. H. Tsai, P. W. Liu, Y. H. Lin, C. T. Tsai, G. H. Ma, S. C. Chien, and S. W. Sun, "More strain and less stress- the guideline for developing high-end strained CMOS technologies with acceptable reliability," in *Electron Devices Meeting*, 2008. *IEDM 2008. IEEE International*, 2008, pp. 1-4.
- [14] P. Packan, S. Cea, H. Deshpande, T. Ghani, M. Giles, O. Golonzka, M. Hattendorf, R. Kotlyar, K. Kuhn, A. Murthy, P. Ranade, L. Shifren, C. Weber, and K. Zawadzki, "High performance Hi-K + metal gate strain enhanced transistors on (110) silicon," in *Electron Devices Meeting*, 2008. *IEDM 2008. IEEE International*, 2008, pp. 1-4.
- [15] Y. Taur and T. H. Ning, *Fundamentals of Modern VLSI Devices*, First ed.: Cambridge University Press, 1998.
- [16] G. G. Shahidi, "SOI technology for the GHz era," in VLSI Technology, Systems, and Applications, 2001. Proceedings of Technical Papers. 2001 International Symposium on, 2001, pp. 11-14.
- [17] V. Barral, T. Poiroux, F. Andrieu, C. Buj-Dufournet, O. Faynot, T. Ernst, L. Brevard, C. Fenouillet-Beranger, D. Lafond, J. M. Hartmann, V. Vidal, F. Allain, N. Daval, I. Cayrefourcq, L. Tosti, D. Munteanu, J. L. Autran, and S. Deleonibus, "Strained FDSOI CMOS technology scalability down to 2.5nm film thickness and 18nm gate length with a TiN/HfO₂ gate stack," in *Electron Devices Meeting*, 2007. *IEDM* 2007. *IEEE International*, 2007, pp. 61-64.
- [18] C. Fenouillet-Beranger, S. Denorme, B. Icard, F. Boeuf, J. Coignus, O. Faynot, L. Brevard, C. Buj, C. Soonekindt, J. Todeschini, J. C. Le-Denmat, N. Loubet, C. Gallon, P. Perreau, S. Manakli, B. Mmghetti, L. Pain, V. Arnal, A. Vandooren, D. Aime, L. Tosti, C. Savardi, F. Martin, T. Salvetat, S. Lhostis, C. Laviron, N. Auriac, T. Kormann, G. Chabanne, S. Gaillard, O. Belmont, E. Laffosse, D. Barge, A. Zauner, A. Tarnowka, K. Romanjec, H. Brut, A. Lagha, S. Bonnetier, F. Joly, N. Mayet, A. Cathignol, D. Galpin, D. Pop, R. Delsol, R. Pantel, F. Pionnier, G. Thomas, D. Bensahel, S. Deleombus, T. Skotnicki, and H. Mmgam, "Fully-depleted SOI technology using high-k and single-metal gate for 32 nm node LSTP applications featuring 0.179 μm² 6T-SRAM bitcell," in *Electron Devices Meeting, 2007. IEDM 2007. IEEE International*, 2007, pp. 267-270.
- [19] K. Henson, H. Bu, M. H. Na, Y. Liang, U. Kwon, S. Krishnan, J. Schaeffer, R. Jha, N. Moumen, R. Carter, C. DeWan, R. Donaton, D. Guo, M. Hargrove, W. He, R. Mo, R. Ramachandran, K. Ramani, K. Schonenberg, Y. Tsang, X. Wang, M. Gribelyuk, W. Yan,

J. Shepard, E. Cartier, M. Frank, E. Harley, R. Arndt, R. Knarr, T. Bailey, B. Zhang, K. Wong, T. Graves-Abe, E. Luckowski, D. G. Park, V. Narayanan, M. Chudzik, and M. Khare, "Gate length scaling and high drive currents enabled for high performance SOI technology using high- κ /metal gate," in *Electron Devices Meeting*, 2008. *IEDM 2008*. *IEEE International*, 2008, pp. 1-4.

- [20] E. J. Nowak, "Maintaining the benefits of CMOS scaling when scaling bogs down," *IBM Journal of Research & Development*, vol. 46, p. 169, 2002.
- [21] H. Xuejue, L. Wen-Chin, K. Charles, D. Hisamoto, C. Leland, J. Kedzierski, E. Anderson, H. Takeuchi, C. Yang-Kyu, K. Asano, V. Subramanian, K. Tsu-Jae, J. Bokor, and H. Chenming, "Sub 50-nm FinFET: PMOS," in *Electron Devices Meeting*, 1999. IEDM Technical Digest. International, 1999, pp. 67-70.
- [22] Y. Bin, C. Leland, S. Ahmed, W. Haihong, S. Bell, Y. Chih-Yuh, C. Tabery, H. Chau, X. Qi, K. Tsu-Jae, J. Bokor, H. Chenming, L. Ming-Ren, and D. Kyser, "FinFET scaling to 10 nm gate length," in *Electron Devices Meeting*, 2002. *IEDM '02. Digest. International*, 2002, pp. 251-254.
- Y. Fu-Liang, L. Di-Hong, C. Hou-Yu, C. Chang-Yun, L. Sheng-Da, H. Cheng-Chuan, C. Tang-Xuan, C. Hung-Wei, H. Chien-Chao, L. Yi-Hsuan, W. Chung-Cheng, C. Chi-Chun, C. Shih-Chang, C. Ying-Tsung, C. Ying-Ho, C. Chih-Jian, C. Bor-Wen, H. Peng-Fu, S. Jyu-Horng, T. Han-Jan, Y. Yee-Chia, L. Yiming, L. Jam-Wem, C. Pu, L. Mong-Song, and H. Chenming, "5nm-gate nanowire FinFET," in *VLSI Technology, 2004. Digest of Technical Papers. 2004 Symposium on*, 2004, pp. 196-197.
- [24] H. Kawasaki, M. Khater, M. Guillorn, N. Fuller, J. Chang, S. Kanakasabapathy, L. Chang, R. Muralidhar, K. Babich, Q. Yang, J. Ott, D. Klaus, E. Kratschmer, E. Sikorski, R. Miller, R. Viswanathan, Y. Zhang, J. Silverman, Q. Ouyang, A. Yagishita, M. Takayanagi, W. Haensch, and K. Ishimaru, "Demonstration of highly scaled FinFET SRAM cells with high-κ/metal gate and investigation of characteristic variability for the 32 nm node and beyond," in *Electron Devices Meeting*, 2008. *IEDM* 2008. *IEEE International*, 2008, pp. 1-4.
- [25] M. V. Fischetti, T. P. O'Regan, N. Sudarshan, C. Sachs, J. Seonghoon, K. Jiseok, and Z. Yan, "Theoretical Study of Some Physical Aspects of Electronic Transport in nMOSFETs at the 10-nm Gate-Length," *Electron Devices, IEEE Transactions on*, vol. 54, pp. 2116-2136, 2007.
- [26] J. D. Krauss and D. A. Fleisch, *Electromagnetics with Applications*, Fifth ed.: McGraw-Hill, 1999.
- [27] S. Takagi, M. Iwase, and A. Toriumi, "On the universality of inversion-layer mobility in nand p-channel MOSFETs," in *Electron Devices Meeting*, 1988. *IEDM* '88. *Technical Digest.*, *International*, 1988, pp. 398-401.
- [28] S. Takagi, A. Toriumi, M. Iwase, and H. Tango, "On the universality of inversion layer mobility in Si MOSFET's: Part I-effects of substrate impurity concentration," *Electron Devices, IEEE Transactions on*, vol. 41, pp. 2357-2362, 1994.
- [29] W. R. Thurber, R. L. Mattis, Y. M. Liu, and J. J. Filliben, "Resistivity-Dopant Density Relationship for Phosphorus-Doped Silicon," *Journal of The Electrochemical Society*, vol. 127, pp. 1807-1812, 1980.
- [30] M. Lundstrom, *Fundamentals of carrier transport* Second ed.: Cambridge University Press, 2000.

- [31] M. V. Fischetti and S. E. Laux, "Monte carlo analysis of electron transport in small semiconductor devices including band-structure and space-charge effects," *Physical Review B*, vol. 38, p. 9721, 1988.
- [32] K. Tomizawa, *Numerical Simulation of Submicron Semiconductor Devices*: Artech House, 1993.
- [33] M. V. Fischetti, "Effect of the electron-plasmon interaction on the electron mobility in silicon," *Physical Review B*, vol. 44, p. 5527, 1991.
- [34] H. Kosina and G. Kaiblinger-Grujin, "Ionized-impurity scattering of majority electrons in silicon," *Solid-State Electronics*, vol. 42, pp. 331-338, 1998.
- [35] K. Diff and K. F. Brennan, "Theory of electron-plasmon-scattering rate in highly doped bulk semiconductors," *Journal of Applied Physics*, vol. 69, pp. 3097-3103, 1991.
- [36] D. Esseni and A. Abramo, "Modeling of electron mobility degradation by remote Coulomb scattering in ultrathin oxide MOSFETs," *Electron Devices, IEEE Transactions on*, vol. 50, pp. 1665-1674, 2003.
- [37] F. Gamiz, J. B. Roldan, J. E. Carceller, and P. Cartujo, "Monte Carlo simulation of remote-Coulomb-scattering-limited mobility in metal--oxide--semiconductor transistors," *Applied Physics Letters*, vol. 82, pp. 3251-3253, 2003.
- [38] S.-i. Saito, K. Torii, M. Hiratani, and T. Onai, "Improved theory for remote-chargescattering-limited mobility in metal--oxide--semiconductor transistors," *Applied Physics Letters*, vol. 81, pp. 2391-2393, 2002.
- [39] S. Takagi and M. Takayanagi, "Experimental Evidence of Inversion-Layer Mobility Lowering in Ultrathin Gate Oxide Metal-Oxide-Semiconductor Field-Effect-Transistors with Direct Tunneling Current," *Jpn. J. Appl. Phys., Part 1*, vol. 41, pp. 2348-2352, 2002.
- [40] P. Toniutti, P. Palestri, D. Esseni, and L. Selmi, "Revised analysis of the mobility and I_{ON} degradation in high-κ gate stacks: Surface optical phonons vs. remote Coulomb scattering," in *Solid-State Device Research Conference*, 2008. ESSDERC 2008. 38th European, 2008, pp. 246-249.
- [41] Y. Nakabayashi, T. Ishihara, J. Koga, M. Takayanagi, and S. Takagi, "New findings on inversion-layer mobility in highly doped channel Si MOSFETs," in *Electron Devices Meeting*, 2005. *IEDM Technical Digest. IEEE International*, 2005, pp. 133-136.
- [42] W. J. Zhu and T. P. Ma, "Temperature dependence of channel mobility in HfO₂-gated NMOSFETs," *Electron Device Letters, IEEE*, vol. 25, pp. 89-91, 2004.
- [43] M. V. Fischetti and S. E. Laux, "Performance degradation of small silicon devices caused by long-range Coulomb interactions," *Applied Physics Letters*, vol. 76, pp. 2277-2279, 2000.
- [44] M. V. Fischetti, "Scaling MOSFETs to the Limit: A Physicists's Perspective," Journal of Computational Electronics, vol. 2, pp. 73-79, 2003.
- [45] M. V. Fischetti, "Long-range Coulomb interactions in small Si devices. Part II. Effective electron mobility in thin-oxide structures," *Journal of Applied Physics*, vol. 89, pp. 1232-1250, 2001.

- [46] F. Lime, C. Guiducci, R. Clerc, G. Ghibaudo, C. Leroux, and T. Ernst, "Characterization of effective mobility by split C(V) technique in N-MOSFETs with ultra-thin gate oxides," *Solid-State Electronics*, vol. 47, pp. 1147-1153, 2003.
- [47] E. Conwell and V. F. Weisskopf, "Theory of Impurity Scattering in Semiconductors," *Physical Review*, vol. 77, p. 388, 1950.
- [48] D. Chattopadhyay and H. J. Queisser, "Electron scattering by ionized impurities in semiconductors," *Reviews of Modern Physics*, vol. 53, p. 745, 1981.
- [49] N. Sclar, "Ionized Impurity Scattering in Nondegenerate Semiconductors," *Physical Review*, vol. 104, p. 1548, 1956.
- [50] S. Chandrasekhar, "Stochastic Problems in Physics and Astronomy," *Reviews of Modern Physics*, vol. 15, p. 1, 1943.
- [51] B. K. Ridley, *Quantum Processes in Semiconductors*, Fourth ed.: Clarendon Press, 1999.
- [52] P. P. Debye and E. M. Conwell, "Electrical Properties of N-Type Germanium," *Physical Review*, vol. 93, p. 693, 1954.
- [53] J. M. Ziman, *Principles of the Theory of Solids*, Second ed.: Cambridge University Press, 1972.
- [54] J. S. Blakemore, "Approximations for Fermi-Dirac integrals, especially the function used to describe electron density in a semiconductor," *Solid-State Electronics*, vol. 25, pp. 1067-1076, 1982.
- [55] B. K. Ridley, "Reconciliation of the Conwell-Weisskopf and Brooks-Herring formulae for charged-impurity scattering in semiconductors: Third-body interference," *Journal of Physics C: Solid State Physics*, vol. 10, p. 1589, 1977.
- [56] H. Kosina, G. Kaiblinger-Grujin, and S. Selberherr, "A new approach to ionized-impurity scattering," in *Simulation of Semiconductor Processes and Devices*, 1997. SISPAD '97., 1997 International Conference on, 1997, pp. 205-208.
- [57] J. B. Krieger, J. Gruenebaum, and T. Meeks, "Resistivity of n-type degenerately doped silicon at low temperature: Dielectric-screening effects," *Physical Review B*, vol. 9, p. 3627, 1974.
- [58] D. K. Ferry, *Semiconductor Transport*, First ed.: Taylor & Francis, 2000.
- [59] N. Takimoto, "On the screening of Impurity Potential by Conduction Electrons," *Journal of the Physical Society of Japan,* vol. 14, pp. 1142-1158, September 1959.
- [60] T. G. Van de Roer and F. P. Widdershoven, "Ionized impurity scattering in Monte Carlo calculations," *Journal of Applied Physics*, vol. 59, pp. 813-815, 1986.
- [61] H. Kosina, "A method to reduce small-angle scattering in Monte Carlo device analysis," *Electron Devices, IEEE Transactions on*, vol. 46, pp. 1196-1200, 1999.
- [62] C. Alexander, G. Roy, and A. Asenov, "Random-Dopant-Induced Drain Current Variation in Nano-MOSFETs: A Three-Dimensional Self-Consistent Monte Carlo Simulation Study Using *Ab Initio* Ionized Impurity Scattering," *Electron Devices, IEEE Transactions on*, vol. 55, pp. 3251-3258, 2008.

- [63] A. Asenov, "Random dopant induced threshold voltage lowering and fluctuations in sub-0.1μm MOSFET's: A 3-D atomistic simulation study," *Electron Devices*, *IEEE Transactions on*, vol. 45, pp. 2505-2513, 1998.
- [64] A. Asenov, G. Slavcheva, A. R. Brown, J. H. Davies, and S. Saini, "Increase in the random dopant induced threshold fluctuations and lowering in sub-100 nm MOSFETs due to quantum effects: a 3-D density-gradient simulation study," *Electron Devices, IEEE Transactions on*, vol. 48, pp. 722-729, 2001.
- [65] A. R. Brown, A. Asenov, and J. R. Watling, "Intrinsic fluctuations in sub 10-nm doublegate MOSFETs introduced by discreteness of charge and matter," *Nanotechnology, IEEE Transactions on*, vol. 1, pp. 195-200, 2002.
- [66] C. L. Alexander, A. R. Brown, J. R. Watling, and A. Asenov, "Impact of scattering on random dopant induced current fluctuations in decanano MOSFETs," in *SISPAD*, 2004, pp. 223-226.
- [67] C. Alexander, A. R. Brown, J. R. Watling, and A. Asenov, "Impact of scattering in `atomistic' device simulations," *Solid-State Electronics*, vol. 49, pp. 733-739, 2005.
- [68] C. L. Alexander, G. Roy, and A. Asenov, "Random Impurity Scattering Induced Variability in Conventional Nano-Scaled MOSFETs: Ab initio Impurity Scattering Monte Carlo Simulation Study," in *Electron Devices Meeting*, 2006. *IEDM '06. International*, 2006, pp. 1-4.
- [69] G. Baccarani, F. Odeh, A. Gnudi, and D. Ventura, "A Critical Review of the Fundamental Semiconductor Equations," *Institute for Mathematics and Its Applications Preprints*, p. 1019, August 1992.
- [70] D. A. McQuarrie, *Statistical Mechanics*, 1st ed.: University Science Books, 2000.
- [71] C. Jacoboni and P. Lugli, *The Monte Carlo Method for Semiconductor Device Simulation*: Springer-Werlag/Wien, 1989.
- [72] O. Madelung, *Introduction to Solid-State Theory*, 2nd ed.: Springer-Verlag, 1981.
- [73] A. Gnudi, D. Ventura, G. Baccarani, and F. Odeh, "Two-dimensional MOSFET simulation by means of a multidimensional spherical harmonics expansion of the Boltzmann transport equation," *Solid-State Electronics*, vol. 36, pp. 575-581, 1993.
- [74] W. Fawcett, A. D. Boardman, and S. Swain, "Monte Carlo determination of electron transport properties in gallium arsenide," *Journal of Physics and Chemistry of Solids*, vol. 31, pp. 1963-1990, 1970.
- [75] C. Jacoboni and L. Reggiani, "The Monte Carlo method for the solution of charge transport in semiconductors with applications to covalent materials," *Reviews of Modern Physics*, vol. 55, p. 645, 1983.
- [76] C. C. Moglestue, "A Self-Consistent Monte Carlo Particle Model to Analyze Semiconductor Microcomponents of any Geometry," *Computer-Aided Design of Integrated Circuits and Systems, IEEE Transactions on*, vol. 5, pp. 326-345, 1986.
- [77] M. V. Fischetti, S. E. Laux, and E. Crabbe, "Understanding hot-electron transport in silicon devices: Is there a shortcut?," *Journal of Applied Physics*, vol. 78, pp. 1058-1087, 1995.
- [78] C. J. Wordelman and U. Ravaioli, "Integration of a particle-particle-mesh algorithm with the ensemble Monte Carlo method for the simulation of ultra-small

semiconductor devices," *Electron Devices, IEEE Transactions on*, vol. 47, pp. 410-416, 2000.

- [79] C. Jacoboni, R. Minder, and G. Majni, "Effects of band non-parabolicity on electron drift velocity in silicon above room temperature," *Journal of Physics and Chemistry of Solids*, vol. 36, pp. 1129-1133, 1975.
- [80] T. Kunikiyo, M. Takenaka, Y. Kamakura, M. Yamaji, H. Mizuno, M. Morifuji, K. Taniguchi, and C. Hamaguchi, "A Monte Carlo simulation of anisotropic electron transport in silicon including full band structure and anisotropic impact-ionization model," *Journal of Applied Physics*, vol. 75, pp. 297-312, 1994.
- [81] L. Reggiani and C. Calandra, "Bloch states mixing in Si conduction band," *Physics Letters A*, vol. 43, pp. 339-340, 1973.
- [82] H. D. Rees, "Calculation of steady state distribution functions by exploiting stability," *Physics Letters A*, vol. 26, pp. 416-417, 1968.
- [83] H. D. Rees, "Calculation of distribution functions by exploiting the stability of the steady state," *Journal of Physics and Chemistry of Solids*, vol. 30, pp. 643-655, 1969.
- [84] E. Pop, R. W. Dutton, and K. E. Goodson, "Analytic band Monte Carlo model for electron transport in Si including acoustic and optical phonon dispersion," *Journal of Applied Physics*, vol. 96, pp. 4998-5005, 2004.
- [85] C. Alexander, "*private communication*," Discussion on the application and calibration of empirical ionized impurity correction factor. ed: Unpublished work, 2008.
- [86] S. M. Goodnick, D. K. Ferry, C. W. Wilmsen, Z. Liliental, D. Fathy, and O. L. Krivanek, "Surface roughness at the Si(100)-SiO2 interface," *Physical Review B*, vol. 32, p. 8171, 1985.
- [87] T. Ando, "Screening Effect and Quantum Transport in a Silicon Inversion Layer in Strong Magnetic Fields," *Journal of the Physical Society of Japan*, vol. 43, pp. 1616-1626, 1977.
- [88] G. F. Formicone, M. Saraniti, D. Z. Vasileska, and D. K. Ferry, "Study of a 50 nm nMOSFET by ensemble Monte Carlo simulation including a new approach to surface roughness and impurity scattering in the Si inversion layer," *Electron Devices, IEEE Transactions on*, vol. 49, pp. 125-132, 2002.
- [89] R. W. Hockney and J. W. Eastwood, *Computer Simulation Using Particles*: Taylor & Francis, 1988.
- [90] C. K. Birdsall and D. Fuss, "Clouds-in-clouds, clouds-in-cells physics for many-body plasma simulation," *Journal of Computational Physics*, vol. 3, pp. 494-511, 1969.
- [91] P. Palestri, N. Barin, D. Esseni, and C. Fiegna, "Revised stability analysis of the nonlinear Poisson scheme in self-consistent Monte Carlo device simulations," *Electron Devices, IEEE Transactions on*, vol. 53, pp. 1443-1451, 2006.
- [92] P. W. Rambo and J. Denavit, "Time stability of Monte Carlo device simulation," *Computer-Aided Design of Integrated Circuits and Systems, IEEE Transactions on*, vol. 12, pp. 1734-1741, 1993.
- [93] P. Palestri, N. Barin, D. Esseni, and C. Fiegna, "Stability of self-consistent Monte Carlo Simulations: effects of the grid size and of the coupling scheme," *Electron Devices, IEEE Transactions on*, vol. 53, pp. 1433-1442, 2006.

- [94] C. Canali, C. Jacoboni, F. Nava, G. Ottaviani, and A. Alberigi-Quaranta, "Electron drift velocity in silicon," *Physical Review B*, vol. 12, p. 2265, 1975.
- [95] N. W. Ashcroft and N. D. Mermin, *Solid State Physics*: Thomson Learning, 1976.
- [96] G. B. Arfken and H. J. Weber, *Mathematical Methods for Physicists*, Sixth ed.: Elsevier Academic, 2005.
- [97] G. Stephenson and P. M. Radmore, *Advanced mathematical methods for engineering and science students*: Cambridge University Press, 1990.
- [98] I. S. Gradshteyn and I. M. Ryzhik, *Table of Integrals, Series and Products*, Sixth ed.: Academic Press, 2000.
- [99] G. Roy, A. R. Brown, A. Asenov, and S. Roy, "Bipolar quantum corrections in resolving individual dopants in `atomistic' device simulation," *Superlattices and Microstructures*, vol. 34, pp. 327-334, 2003.
- [100] J. D. Jackson, Classical Electrodynamics, 3rd ed.: Wiley, 1999.
- [101] A. G. Samoilovich, I. Y. Korenblit, and I. V. Dakhovskii, "Anisotropic Electron Scattering By Ionized Impurities," *Soviet Physics - Doklady*, vol. 6, p. 606, 1962.
- [102] D. Long, "Ionized-Impurity Scattering Mobility of Electrons in Silicon," *Physical Review*, vol. 129, p. 2464, 1963.
- [103] A. D. Boardman, "Many-Valley Dipole Scattering of Electrons in Germanium and Silicon," *Physical Review*, vol. 147, p. 532, 1966.
- [104] V. Khrapai and A. Gold, "Calculation of the anisotropic mobility in (110) AlAs quantum wells at zero temperature," *JETP Letters*, vol. 90, pp. 346-349, 2009.
- [105] A. D. Boardman, A. D. Hill, and S. Sampanthar, "Partial-Wave Scattering by Non-Spherically-Symmetric Potentials. I. General Theory of Elastic Scattering," *Physical Review*, vol. 160, p. 472, 1967.
- [106] E. H. Sondheimer, "The Boltzman Equation for Anisotropic Metals," *Proceedings of the Royal Society of London. Series A. Mathematical and Physical Sciences*, vol. 268, pp. 100-108, June 19, 1962 1962.
- [107] G. Harań and A. D. S. Nagi, "Role of anisotropic impurity scattering in anisotropic superconductors," *Physical Review B*, vol. 54, p. 15463, 1996.
- [108] Y. Tokura, "Two-dimensional electron transport with anisotropic scattering potentials," *Physical Review B*, vol. 58, p. 7151, 1998.
- [109] V. S. Khrapai and A. Gold, "Transport properties of the two-dimensional interacting anisotropic electron gas at zero temperature," *Physical Review B*, vol. 81, p. 085309, 2010.
- [110] C. Jacoboni and L. Reggiani, "Bulk hot-electron properties of cubic semiconductors," *Advances in Physics*, vol. 28, pp. 493 553, 1979.
- [111] J. R. Watling, "*private communication*," Discussion on the numerically capped ionized impurity scattering model. ed: unpublished work, 2008.
- [112] W. H. Press, B. P. Flannery, S. A. Teukolsky, and W. T. Vetterling, *Numerical Recipes in FORTRAN* 77, Second ed.: Cambridge University Press, 1992.

References

- [113] S. Inaba, K. Okano, S. Matsuda, M. Fujiwara, A. Hokazono, K. Adachi, K. Ohuchi, H. Suto, H. Fukui, T. Shimizu, S. Mori, H. Oguma, A. Murakoshi, T. Itani, T. Iinuma, T. Kudo, H. Shibata, S. Taniguchi, T. Matsushita, S. Magoshi, Y. Watanabe, M. Takayanagi, A. Azuma, H. Oyamatsu, K. Suguro, Y. Katsumata, Y. Toyoshima, and H. Ishiuchi, "High performance 35 nm gate length CMOS with NO oxynitride gate dielectric and ni SALICIDE," in *Electron Devices Meeting, 2001. IEDM Technical Digest. International*, 2001, pp. 29.6.1-29.6.4.
- [114] S. Inaba, K. Okano, S. Matsuda, M. Fujiwara, A. Hokazono, K. Adachi, K. Ohuchi, H. Suto, H. Fukui, T. Shimizu, S. Mori, H. Oguma, A. Murakoshi, T. Itani, T. Iinuma, T. Kudo, H. Shibata, S. Taniguchi, M. Takayanagi, A. Azuma, H. Oyamatsu, K. Suguro, Y. Katsumata, Y. Toyoshima, and H. Ishiuchi, "High performance 35 nm gate length CMOS with NO oxynitride gate dielectric and Ni salicide," *Electron Devices, IEEE Transactions on*, vol. 49, pp. 2263-2270, 2002.
- [115] W. Xingsheng, S. Roy, and A. Asenov, "High performance MOSFET scaling study from bulk 45 nm technology generation," in *Solid-State and Integrated-Circuit Technology*, 2008. ICSICT 2008. 9th International Conference on, 2008, pp. 484-487.
- [116] P. Palestri, C. Alexander, A. Asenov, G. Baccarani, A. Bournel, M. Braccioli, B. Cheng, P. Dollfus, A. Esposito, D. Esseni, A. Ghetti, C. Fiegna, G. Fiori, V. Aubry-Fortuna, G. Iannaccone, A. Martinez, B. Majkusiak, S. Monfray, S. Reggiani, C. Riddet, J. Saint-Martin, E. Sangiorgi, A. Schenk, L. Selmi, L. Silvestri, and J. Walczak, "Comparison of advanced transport models for nanoscale nMOSFETs," in *Ultimate Integration of Silicon*, 2009. ULIS 2009. 10th International Conference on, 2009, pp. 125-128.
- [117] Synopsys, "Sentaurus device user manual," <u>http://www.synopsys.com/</u>.
- [118] F. Aduma-Lema, "Scaling and Intrinsic Parameter Fluctuations in Nano-CMOS Devices," PhD thesis, Electronics and Electrical Engineering, University of Glasgow, 2005.
- [119] G. D. Roy, "Simulation of Intrinsic Parameter Fluctuations in Nano-CMOS Devices," PhD thesis, Electronics and Electrical Engineering, University of Glasgow, 2005.
- [120] M. Fischetti, S. Jin, T. W. Tang, P. Asbeck, Y. Taur, S. Laux, M. Rodwell, and N. Sano, "Scaling MOSFETs to 10 nm: Coulomb effects, source starvation, and virtual source model," *Journal of Computational Electronics*, vol. 8, pp. 60-77, 2009.
- [121] M. Ishizaka, T. Iizuka, S. Ohi, M. Fukuma, and H. Mikoshiba, "Advanced electron mobility model of MOS inversion layer considering 2D-degenerate electron gas physics," in *Electron Devices Meeting*, 1990. IEDM '90. Technical Digest., International, 1990, pp. 763-766.
- [122] G. Ghibaudo, "A Simple Derivation of the Kubo-Greenwood Formula," *physica status solidi* (*b*), vol. 153, pp. K155-K158, 1989.
- [123] A. R. Leach, *Molecular Modelling: Principles and Applications*, 2nd ed.: Pearson Education, 2001.
- [124] G. E. P. Box, G. M. Jenkins, and G. C. Reinsel, *Time Series Analysis: Forecasting and Control*, 3rd ed. New Jersey: Prentice-Hall, 1994.
- [125] F. Balestra, S. Cristoloveanu, M. Benachir, J. Brini, and T. Elewa, "Double-gate siliconon-insulator transistor with volume inversion: A new device with greatly enhanced performance," *Electron Device Letters, IEEE*, vol. 8, pp. 410-412, 1987.

- [126] D. Esseni, A. Abramo, L. Selmi, and E. Sangiorgi, "Physically based modeling of low field electron mobility in ultrathin single- and double-gate SOI n-MOSFETs," *Electron Devices, IEEE Transactions on*, vol. 50, pp. 2445-2455, 2003.
- [127] C. Riddet, "Monte Carlo Study of Current Variability in UTB SOI DG MOSFETs," PhD thesis, Electronics and Electrical Engineering, University of Glasgow, 2008.
- [128] K. Banoo, "Direct solution of the Boltzmann transport equation in nanoscale silicon devices," in *Electrical and Computer Engineering*: Purdue University, 2000.
- [129] S. M. Hong and C. Jungemann, "A fully coupled scheme for a Boltzmann-Poisson equation solver based on a spherical harmonics expansion," *Journal of Computational Electronics*, vol. 8, pp. 225-241, 2009.
- [130] I. S. Gradshteyn and I. M. Ryzhik, *Table of Integrals, Series, and Products*, 7th ed.: Academic Press, 2007.
- [131] "Mathematica," Version 7.0 ed Champaign, Illinois: Wolfram Research, Inc., 2008.